# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q  Mark One)  QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the quarterly period ended April 30, 2024 OR  TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the transition period from to Commission File Number: 001-40601  Couchbase, Inc. (Exact name of registrant as specified in its charter)  Delaware (State or other jurisdiction of incorporation or organization)  3250 Olcott Street  Santa Clara, California 95054 (Address of principal securities effices and zije Code) (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act or 1934 during the precede 12 months (or for such shorter period that the registrant was required to Tile such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Evolution 15 or 15(d) of the Securities Exchange Act or 1934 during the preceding 12 months (or for such shorter period that the registrant was required to be abonited pursuant to Rule 405 of Regulation S-T (§232-405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes Evolutional by check mark whether the registrant is a large accelerated filer, an ancelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchanded.  Large accelerated filer  Non-accelerated filer  Smaller reporting company  Emerging growth company, indicate by check mark if the registrant has elected not to use the extend				
For the quarterly period ended April 30, 2024 OR  TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the quarterly period ended April 30, 2024 OR  TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the transition period from to Commission File Number: 001-40601  COUChbase, Inc. (Exact name of registrant as specified in its charter)  Delaware (State or other paradiction of incorporation or organization)  3250 Oleott Street Santa Clara, California 95054 (Address of principal excentive office and Zip Code) (650) 417-7500 (Registrant's tolephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act: Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Indicate by check mark whether the registrant has submitted electronically every Interestive Data File required to such filing requirements for the past 90 days. Yes Endown of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to this chapter) during the preceding 12 months (or for such shorter period that the registrant was precided to the registrant was required to be submitted pursuant to Rule 405 of Regulation S-T [8323-405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes END O Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. In Rule 12b-2 of the Exchandet.  Large accelerated filer		FORM 10-Q		
For the quarterly period ended April 30, 2024 OR  TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the transition period from to Commission File Number: 001-40601  Couchbase, Inc. (Exact name of registrant as specified in its charter)  Delaware (State or other jurisdiction of incorporation or organization)  3250 Olcott Street Santa Clara, California 95054 (Address of principal executive offices and Zip Code) (G50) 417-7500 (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class Trading Symbol(s) Name of each exchange on which registered  Indicate by check mark whether the registrant was required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding Immediate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 22 (1932-406)  Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T [8323-405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to be submitted pursuant to Rule 405 of Regulation S-T [8323-405 of this chapter) during the preceding 12 months for for such shorter period that the registrant was required to obsubnit such files). Yes 28 No    Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, in Rule 12b-2 of the Exchan Act.  Large accelerated filer   Accelerated filer Smaller reporting company in Rule 12b-2 of the Exchan Act.  Large accelerated filer   Accelerated filer Smaller reporting company in Rule 12b-2 of the Exchan Act.	(Mark One)			
TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934  For the transition period from to Commission File Number: 001-40601  Couchbase, Inc. (Exact name of registrant as specified in its charter)  Delaware (State or other jurisdiction of incorporation of or organization)  3250 Olcott Street  Santa Clara, California 95054 (Address of principal excuite office and Zip Code) (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the precede 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Excluding the preceding 12 months (or for such shorter period that the registrant was required to the such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Excluding the preceding 12 months (or for such shorter period that the registrant was required to submit such files, Yes Excluding the preceding 12 months (or for such shorter period that the registrant was required to submit such files, Yes Excluding the preceding 12 months (or for such shorter period that the registrant was required to submit such files, Yes Excluding the preceding 12 months (or for such shorter period that the registrant was required to submit such files, Yes Excluding the preceding th	☑ QUARTERLY REPORT PURSUANT TO SEC	TION 13 OR 15(d) OF THE SECURITI	ES EXCHANGE ACT OF 1934	
For the transition period from to Commission File Number: 001-40601  Couchbase, Inc.  (Exact name of registrant as specified in its charter)  Delaware (State or other jurisdiction of incorporation or organization)  3250 Olcott Street Santa Clara, California 95054 (Address of principal executive offices and Zip Code) ((650) 417-7500 (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Tes Exchange of the past 90 days. Tes Exchange of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submitted pursuant to Rule 405 of Regulation S-T (8232-405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \( \to	For		)24	
Couchbase, Inc.  (Exact name of registrant as specified in its charter)  Delaware (State or other jurisdiction of incorporation of organization)  Address of principal executive offices and Zip Code) (G50) 417-7500 (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class Trading Symbol(s) Name of each exchange on which registered  Common stock, par value \$0.00001 per share BASE Nasdag Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ENO   Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ENO   Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (8232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ENO   Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer, an accelerated filer, and "emerging growth company" in Rule 12b-2 of the Exchandat.  Large accelerated filer  Non-accelerated filer  Accelerated filer  Smaller reporting company  Emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	☐ TRANSITION REPORT PURSUANT TO SEC	TION 13 OR 15(d) OF THE SECURITI	ES EXCHANGE ACT OF 1934	
(State or other jurisdiction of incorporation or organization)    Common stock, par value \$0.00001 per share   BASE   Nasdaq Global Select Market				
Delaware (State or other jurisdiction of incorporation or organization)  3250 Olcott Street Santa Clara, California 95054 (Address of principal executive offices and Zip Code) (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ☑ No ☐  Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232,405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required that the registrant was required that the registrant was required to submit such files). Yes ☑ No ☐  Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchance.  Large accelerated filer  Accelerated filer  Smaller reporting company, Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised		Couchbase, Inc.		
(State or other jurisdiction of incorporation or organization)  3250 Olcott Street Santa Clara, California 95054 (Address of principal executive offices and Zip Code) (Address of principal executive offices and Zip Code) (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class Trading Symbol(s) Name of each exchange on which registered  Common stock, par value \$0.00001 per share BASE Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ☑ No ☐  Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ☑ No ☐  Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	(Exac	,	arter)	
Santa Clara, California 95054 (Address of principal executive offices and Zip Code)  (650) 417-7500 (Registrant's telephone number, including area code)  Securities registered pursuant to Section 12(b) of the Act:  Title of each class Trading Symbol(s) Name of each exchange on which registered  Common stock, par value \$0.00001 per share BASE Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 2 not be submitted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to learn the registrant was required to submit such files). Yes 10 not prompany. See the definitions of "large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchangate.  Large accelerated filer  Non-accelerated filer  Smaller reporting company Emerging growth company Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	(State or other jurisdiction of	3250 Olcott Street	(I.R.S. Employer	
Securities registered pursuant to Section 12(b) of the Act:  Title of each class Title of each class Tommon stock, par value \$0.00001 per share  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 20 no 10 notice by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes 20 no 10 notice by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Non-accelerated filer  Non-accelerated filer  Smaller reporting company Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised		Santa Clara, California 95054		
Title of each class  Trading Symbol(s)  Name of each exchange on which registered  Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preced 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ☑ No ☐  Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ☑ No ☐  Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Non-accelerated filer  Accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised		` '		
Common stock, par value \$0.00001 per share  BASE  Nasdaq Global Select Market  Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	Securities registered pursuant to Section 12(b) of the Act:			
Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes No Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes No Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes ⊠ No □  Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ☒ No □  Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchangact.  Large accelerated filer □ □ Accelerated filer  Non-accelerated filer □ Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	Common stock, par value \$0.00001 per share	BASE	Nasdaq Global Select Market	
(§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ☒ No ☐  Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchanged accelerated filer  Large accelerated filer  Non-accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	12 months (or for such shorter period that the registrant was recond $\square$	quired to file such reports), and (2) has been sub	eject to such filing requirements for the past 90 days.	Yes ⊠
company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.  Large accelerated filer  Non-accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised				3-1
Non-accelerated filer  Smaller reporting company  Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	company. See the definitions of "large accelerated filer," "accel			
Emerging growth company  If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	Large accelerated filer □		Accelerated filer	X
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised	Non-accelerated filer		Smaller reporting company	
			Emerging growth company	X
financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. □			ansition period for complying with any new or revise	d
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes □ No 区	Indicate by check mark whether the registrant is a shell compar	ny (as defined in Rule 12b-2 of the Exchange A	ct). Yes $\square$ No $\boxtimes$	
As of May 31, 2024, the registrant had 50,309,880 shares of common stock outstanding.	As of May 31, 2024, the registrant had 50,309,880 shares of co	ommon stock outstanding.		

# **Table of Contents**

		Page
Part I.	Financial Information	
Item 1.	<u>Financial Statements (unaudited)</u>	4
	Condensed Consolidated Balance Sheets	4
	Condensed Consolidated Statements of Operations	5
	Condensed Consolidated Statements of Comprehensive Loss	6
	Condensed Consolidated Statements of Stockholders' Equity	7
	Condensed Consolidated Statements of Cash Flows	8
	Notes to Condensed Consolidated Financial Statements	9
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	21
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	36
Item 4.	Controls and Procedures	36
Part II.	Other Information	38
Item 1.	<u>Legal Proceedings</u>	38
Item 1A.	Risk Factors	38
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	68
Item 3.	<u>Defaults Upon Senior Securities</u>	68
Item 4.	Mine Safety Disclosures	68
Item 5.	Other Information	68
Item 6.	<u>Exhibits</u>	69
<b>Signatures</b>		70

# Special Note Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of the federal securities laws, which statements involve substantial risks and uncertainties. Forward-looking statements generally relate to future plans or events, management's expectations and opinions or our future financial or operating performance, intentions, designs, expectations or objectives. In some cases, you can identify forward-looking statements because they contain words such as "may," "will," "should," "expect," "plan," "anticipate," "could," "would," "intend," "target," "project," "forecast," "contemplate," "believe," "estimate," "predict," "seek," "pursue," "potential" or "continue" or the negative of these words or other similar terms or expressions that concern our expectations, strategy, plans or intentions. Forward-looking statements contained in this Quarterly Report on Form 10-Q include, but are not limited to, statements about our expectations regarding:

- our future financial performance, including our expectations regarding our revenue, cost of revenue, operating expenses, period of benefit for deferred commissions, our ability to determine reserves and our ability to achieve and maintain future profitability;
- the sufficiency of our cash, cash equivalents and short-term investments to meet our liquidity needs;
- the demand for our products and services or for data management solutions in general;
- our ability to attract and retain customers and partners;
- our ability to develop new products and features and bring them to market in a timely manner and make enhancements to our offerings, as well as market acceptance of new products and features;
- our expectations regarding future developments with respect to Couchbase Capella, our fully-managed database-as-a-service ("DBaaS")
  offering;
- our ability to compete with existing and new competitors in existing and new markets and offerings;
- macroeconomic conditions, foreign exchange fluctuation, inflation concerns, increased interest rates and recessionary fears, on our business and results of operations;
- our expectations regarding the effects of existing and developing laws, rules, regulations and other legal obligations, including with respect to taxation and data privacy and security;
- our ability to manage risk associated with our business;
- our expectations regarding new and evolving markets, as well as the impact of artificial intelligence ("AI") in business, including development of Capella iQ, our generative AI-powered developer's coding assistant built into Couchbase Capella;
- our ability to maintain, develop and protect our brand;
- our ability, and our customers' and our third-party service providers' ability, to maintain the security and availability to each of our technological and physical infrastructures;
- our expectations and management of future growth;
- our expectations concerning relationships with third parties;
- our ability to obtain, maintain, defend and enforce our intellectual property;
- our use of third-party open source software in our solutions and the availability of portions of our source code on an open source basis;
- our ability to successfully acquire and integrate companies and assets; and
- the increased expenses associated with being a public company.

We caution you that the foregoing list may not contain all of the forward-looking statements made in this Quarterly Report on Form 10-Q. You should not rely upon forward-looking statements as predictions of future events.

#### Table of Contents

We have based the forward-looking statements contained in this Quarterly Report on Form 10-Q primarily on our current expectations and projections about future events and trends that we believe may affect our business, financial condition, results of operations and prospects. The outcome of the events described in these forward-looking statements is subject to risks, assumptions, uncertainties and other factors, including those described in the section titled "Risk Factors" and elsewhere in this Quarterly Report on Form 10-Q. Moreover, we operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time and it is not possible for us to predict all risks and uncertainties that could have an impact on the forward-looking statements contained in this Quarterly Report on Form 10-Q. We cannot assure you that the results, events and circumstances reflected in the forward-looking statements will be achieved or occur, and actual results, events or circumstances could differ materially from those described in the forward-looking statements.

The forward-looking statements made in this Quarterly Report on Form 10-Q relate only to events as of the date on which the statements are made. We undertake no obligation to update any forward-looking statements made in this Quarterly Report on Form 10-Q to reflect events or circumstances after the date of this Quarterly Report on Form 10-Q or to reflect new information or the occurrence of unanticipated events, except as required by law. We may not actually achieve the plans, intentions or expectations disclosed in our forward-looking statements and you should not place undue reliance on our forward-looking statements. Our forward-looking statements do not reflect the potential impact of any future acquisitions, mergers, dispositions, joint ventures or investments we may make.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based upon information available to us as of the date of this Quarterly Report on Form 10-Q, and while we believe such information forms a reasonable basis for such statements, such information may be limited or incomplete, and our statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain, and you are cautioned not to unduly rely upon these statements.

You should read this Quarterly Report on Form 10-Q and the documents that we reference in this Quarterly Report on Form 10-Q and have filed as exhibits to this Quarterly Report on Form 10-Q with the understanding that our actual future results, levels of activity, performance and achievements may be different from what we expect. We qualify all of our forward-looking statements by these cautionary statements.

#### **Risk Factor Summary**

Our business is subject to significant risks and uncertainties that make an investment in us speculative and risky. Below we summarize what we believe are the principal risk factors but these risks are not the only ones we face. You should carefully review and consider the full discussion of our risk factors below this summary, together with the other information in this Quarterly Report on Form 10-Q. If any of the following risks or if any of those listed elsewhere in this Quarterly Report on Form 10-Q actually occur, our business, reputation, financial condition, results of operations, revenue and future prospects could be seriously harmed. Additional risks and uncertainties that we are unaware of, or that we currently believe are not material, may also become important factors that adversely affect our business.

- We have a history of net losses, may not achieve or maintain profitability in the future and may not continue to grow on pace with historical rates.
- We face intense competition and if we are unable to compete effectively, our business, financial condition and results of operations would be adversely affected.
- We may fail to cost-effectively acquire new customers or obtain renewals, upgrades or expansions from our existing customers, which would adversely affect our business, financial condition and results of operations.
- The market for our products and services is relatively new and evolving, and our future success depends on the growth and expansion of this
  market.
- If we fail to innovate in response to changing customer needs, new technologies or other market requirements, our business, financial condition and results of operations could be harmed.
- We have a limited operating history, which makes it difficult to predict our future results of operations.
- Our future results of operations and key business metrics may fluctuate significantly, and if we fail to meet the expectations of analysts or
  investors, the market price of our common stock and the value of your investment could decline substantially.
- We recognize a significant portion of revenue from subscriptions over the term of the relevant subscription period, and as a result, downturns or upturns in sales are not immediately reflected in full in our results of operations.
- We depend on our sales force, and we may fail to attract, retain, motivate or train our sales force, which could adversely affect our business, financial condition and results of operations.
- Our sales strategy to target larger enterprises involves risks that may not be present or that are present to a lesser extent with respect to smaller
  enterprises, such as long and unpredictable sales cycles and sales efforts that require considerable time and expense, particularly in the current
  macroeconomic environment.
- If we are not able to maintain and enhance our brand, especially among enterprise architects, application developers and other key functions that support them, our business and results of operations may be adversely affected.
- Real or perceived errors, failures or bugs in our products or interruptions or performance problems associated with our technology and infrastructure could adversely affect our growth prospects, business, financial condition and results of operations.
- Our ability to maintain and increase sales with our existing customers depends, in part, on the quality of our customer support, and our failure to offer high-quality support would harm our reputation and adversely affect our business and results of operations.
- We track certain key business metrics with internal systems and tools and do not independently verify such metrics. Certain of these metrics are subject to inherent challenges in measurement, and any real or perceived inaccuracies in such metrics may adversely affect our business and reputation.
- Our company culture has contributed to our success and if we cannot maintain this culture as we grow, our business could be harmed.

- We may be unable to make acquisitions and investments or successfully integrate acquired companies and assets into our business, and our
  acquisitions and investments may not meet our expectations, any of which could adversely affect our business, financial condition and results
  of operations.
- Use of AI, including in our products and services, combined with an uncertain regulatory environment, may result in reputational harm, liability, or other adverse consequences to our business.
- Our business could be adversely affected by economic downturns.
- If we are unable to maintain successful relationships with our partners, our business, financial condition and results of operations could be harmed
- Certain estimates and information we refer to publicly are based on information from third-party sources and we do not independently verify the accuracy or completeness of the data contained in such sources or the methodologies for collecting such data, and any real or perceived inaccuracies in such estimates and information may harm our reputation and adversely affect our business.
- Our use of third-party open source software in our solutions, the availability of core portions of our source code on an open source or source
  available basis and contributions to our open source projects could negatively affect our ability to sell our products and provide our services,
  subject us to possible litigation and allow third parties to access and use software and technology that we use in our business, all of which
  could adversely affect our business and results of operations.
- Our distribution and licensing model could negatively affect our ability to monetize and protect our intellectual property rights.
- Because of the rights accorded to third parties under open source licenses, there may be fewer technology barriers to entry in the markets in which we compete and it may be relatively easy for new and existing competitors, some of whom may have greater resources than we have, to compete with us.
- We could incur substantial costs in obtaining, maintaining, protecting, defending and enforcing our intellectual property rights and any failure to obtain, maintain, protect, defend or enforce our intellectual property rights could reduce the value of our software and brand.
- We have been and may in the future become subject to intellectual property disputes which may be costly to defend, subject us to significant liability, require us to pay significant damages and limit our ability to use certain technologies.
- If our security measures, or those of our service providers or customers, are breached or unauthorized parties otherwise obtain access to our or our customers' data or software, our products and services may be perceived as not being secure, customers may reduce or terminate their use of our products and services and we may face claims, litigation, regulatory investigations, significant liability and reputational damage.
- A portion of our revenue is generated by sales to heavily regulated organizations, which are subject to a number of challenges and risks.
- Our executive officers, directors and holders of 5% or more of our common stock continue to have substantial control over us, which will limit your ability to influence the outcome of important transactions, including a change in control.

# Part I – Financial Information

# Item 1. Financial Statements (unaudited)

# COUCHBASE, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

(unaudited)

(in thousands, except share and per share amounts)

		As of April 30, 2024	As of January 31, 2024
Assets			
Current assets			
Cash and cash equivalents	\$	51,975	\$ 41,351
Short-term investments		108,220	112,281
Accounts receivable, net		34,580	44,848
Deferred commissions		13,233	15,421
Prepaid expenses and other current assets		10,307	10,385
Total current assets		218,315	224,286
Property and equipment, net		6,113	5,327
Operating lease right-of-use assets		4,135	4,848
Deferred commissions, noncurrent		12,562	11,400
Other assets		1,446	1,891
Total assets	\$	242,571	\$ 247,752
Liabilities and Stockholders' Equity			
Current liabilities			
Accounts payable	\$	4,071	\$ 4,865
Accrued compensation and benefits		8,939	18,116
Other accrued expenses		3,814	4,581
Operating lease liabilities		3,106	3,208
Deferred revenue		89,619	81,736
Total current liabilities		109,549	112,506
Operating lease liabilities, noncurrent		1,388	2,078
Deferred revenue, noncurrent		2,572	2,747
Total liabilities		113,509	117,331
Commitments and contingencies (Note 9)			
Stockholders' equity			
Preferred stock, \$0.00001 par value; 200,000,000 shares authorized as of April 30, 2024 and January 31, 2024; zero shares issued outstanding as of April 30, 2024 and January 31, 2024		_	_
Common stock, \$0.00001 par value; 1,000,000,000 shares authorized as of April 30, 2024 and January 31, 2024; 50,220,260 and 49,079,876 shares issued and outstanding as of April 30, 2024 and January 31, 2024, respectively	y	_	_
Additional paid-in capital		640,931	621,024
Accumulated other comprehensive (loss) income		(215)	56
Accumulated deficit		(511,654)	(490,659)
Total stockholders' equity		129,062	130,421
Total liabilities and stockholders' equity	\$	242,571	\$ 247,752

# COUCHBASE, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(unaudited)

(in thousands, except per share amounts)

	Three Months l	Ended A <sub>l</sub>	pril 30,
	 2024		2023
Revenue:			
License	\$ 6,859	\$	4,943
Support and other	 42,179		33,599
Total subscription revenue	49,038		38,542
Services	2,289		2,454
Total revenue	51,327		40,996
Cost of revenue:			
Subscription	3,957		3,673
Services	1,725		2,249
Total cost of revenue	5,682		5,922
Gross profit	 45,645		35,074
Operating expenses:			
Research and development	17,847		15,383
Sales and marketing	37,755		32,553
General and administrative	12,583		9,625
Restructuring	 		46
Total operating expenses	68,185		57,607
Loss from operations	(22,540)		(22,533)
Interest expense	_		(25)
Other income, net	1,531		1,433
Loss before income taxes	(21,009)		(21,125)
Provision (benefit) for income taxes	(14)		750
Net loss	\$ (20,995)	\$	(21,875)
Net loss per share, basic and diluted	\$ (0.42)	\$	(0.48)
Weighted-average shares used in computing net loss per share, basic and diluted	 49,788		45,843

# COUCHBASE, INC. CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS

(unaudited) (in thousands)

	Three Months Ended April 30,							
	 2024		2023					
Net loss	\$ (20,995)	\$	(21,875)					
Other comprehensive loss:								
Net unrealized (losses) gains on investments, net of tax	(271)		317					
Total comprehensive loss	\$ (21,266)	\$	(21,558)					

# COUCHBASE, INC. CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

(unaudited)

(in thousands, except shares)

	Commo	on S	tock	A	Additional	(	Accumulated Other Comprehensive		G.	Total
	Shares		Amount		Paid-In Capital		Income (Loss)	ımulated Deficit		ckholders' Equity
Balance as of January 31, 2024	49,079,876	\$	_	\$	621,024	\$	56	\$ (490,659)	\$	130,421
Issuance of common stock upon exercise of stock options	406,055		_		3,294		_	_		3,294
Issuance of common stock in connection with employee stock purchase plan	123,778		_		1,795		_	_		1,795
Vesting of restricted stock units	610,551		_		_		_	_		_
Stock-based compensation	_		_		14,818		_	_		14,818
Net unrealized losses on investments	_		_		_		(271)	_		(271)
Net loss	_		_		_		_	(20,995)		(20,995)
Balance as of April 30, 2024	50,220,260	\$		\$	640,931	\$	(215)	\$ (511,654)	\$	129,062

						Accumulated Other				
	Comme	on S	Stock	Additional Paid-In	(	Comprehensive Income	Accumula	ited	Sto	Total ckholders'
	Shares		Amount	Capital		(Loss)	Defici			Equity
Balance as of January 31, 2023	45,432,029	\$		\$ 561,547	\$	(807)	\$ (410.	476)	\$	150,264
Issuance of common stock upon exercise of stock options	389,377		_	1,917		_		_		1,917
Issuance of common stock in connection with employee stock purchase plan	74,113		_	847		_		_		847
Vesting of restricted stock units	259,980		_	_		_		_		_
Stock-based compensation	_		_	9,480		_		—		9,480
Net unrealized gains on investments	_		_	_		317		_		317
Net loss	_		_	_		_	(21,	875)		(21,875)
Balance as of April 30, 2023	46,155,499	\$		\$ 573,791	\$	(490)	\$ (432	351)	\$	140,950

# COUCHBASE, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(unaudited) (in thousands)

	Three Months End	ded April 30,
	2024	2023
Cash flows from operating activities		
Net loss	\$ (20,995) \$	(21,875)
Adjustments to reconcile net loss to net cash used in operating activities		
Depreciation and amortization	400	890
Stock-based compensation, net of amounts capitalized	14,627	9,276
Amortization of deferred commissions	4,096	4,540
Non-cash lease expense	765	772
Foreign currency transaction losses (gains)	283	(84)
Other	(824)	(746)
Changes in operating assets and liabilities		
Accounts receivable	10,165	(2,274)
Deferred commissions	(3,070)	(4,824)
Prepaid expenses and other assets	31	1,405
Accounts payable	(792)	5,458
Accrued compensation and benefits	(9,179)	(4,060)
Other accrued expenses	(813)	(1,256)
Operating lease liabilities	(843)	(826)
Deferred revenue	 7,708	6,423
Net cash provided by (used in) operating activities	1,559	(7,181)
Cash flows from investing activities	 	
Purchases of short-term investments	(19,454)	(7,821)
Maturities and sales of short-term investments	24,144	19,423
Additions to property and equipment	(995)	(1,288)
Net cash provided by investing activities	 3,695	10,314
Cash flows from financing activities		
Proceeds from exercise of stock options	3,294	1,917
Proceeds from issuance of common stock under ESPP	1,795	847
Net cash provided by financing activities	 5,089	2,764
Effect of exchange rate changes on cash, cash equivalents and restricted cash	(262)	(103)
Net increase in cash, cash equivalents and restricted cash	 10,081	5,794
Cash, cash equivalents and restricted cash		
Beginning of period	41,894	40,989
End of period	\$ 51,975 \$	46,783
Cash and cash equivalents	\$ 51,975 \$	46,240
Restricted cash included in other assets	_	543
Total cash, cash equivalents and restricted cash	\$ 51,975 \$	46,783
Supplemental disclosures of cash activities		
Cash paid for income taxes	\$ 591 \$	294
Cash paid for interest	\$ — \$	25
Non-cash investing and financing activities:		
Stock-based compensation capitalized as internal-use software costs	\$ 191 \$	204
Net change in unrealized gains or losses on available-for-sale debt securities	\$ (271) \$	317

#### COUCHBASE, INC.

# NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

#### 1. Description of Business

Couchbase, Inc. provides a leading cloud database platform for modern applications. Couchbase was incorporated in the State of Delaware in 2008 and is headquartered in Santa Clara, California. In these notes to the unaudited condensed consolidated financial statements, the "Company," "Couchbase," "we," "us" and "our" refers to Couchbase, Inc. and its subsidiaries on a consolidated basis.

#### 2. Basis of Presentation and Summary of Significant Accounting Policies

#### **Basis of Presentation**

The Company's unaudited condensed consolidated financial statements and accompanying notes have been prepared in conformity with accounting principles generally accepted in the United States of America ("GAAP") and applicable rules and regulations of the Securities and Exchange Commission ("SEC"), regarding interim financial reporting. As permitted under those rules, certain footnotes or other financial information that are normally required by GAAP have been condensed or omitted, and accordingly the balance sheet as of January 31, 2024, and related disclosures, have been derived from the audited consolidated financial statements at that date but do not include all the information required by GAAP for complete consolidated financial statements. These unaudited condensed consolidated financial statements have been prepared on the same basis as the Company's annual consolidated financial statements and, in the opinion of management, reflect all adjustments (consisting of only normal recurring adjustments) that are necessary for the fair statement of the Company's condensed consolidated financial information. The results of operations for the three months ended April 30, 2024, are not necessarily indicative of the results to be expected for the year ending January 31, 2025, or for any other interim period or for any other future year.

The accompanying unaudited condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements and the related notes contained in the Company's Annual Report on Form 10-K for the year ended January 31, 2024, as filed with the SEC on March 26, 2024.

#### Fiscal Year

The Company's fiscal year ends on January 31. Unless otherwise stated, references to year in these condensed consolidated financial statements relate to fiscal year rather than calendar year.

### **Principles of Consolidation**

The accompanying condensed consolidated financial statements include the accounts of Couchbase, Inc. and its wholly owned subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

# **Use of Estimates**

The preparation of the condensed consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts stated in the financial statements and accompanying notes. Such estimates include, but are not limited to, standalone selling prices for each distinct performance obligation, capitalized internal-use software costs, expected period of benefit for deferred commissions, valuation of stock-based awards, the determination of allowance for credit losses, the incremental borrowing rate used to measure operating lease liabilities, and accounting for income taxes. The Company bases its estimates on historical experience and assumptions that management considers reasonable.

The Company assesses these estimates on a regular basis; however, actual results could differ from these estimates. Estimates and assumptions about future events and their effects, including the impact of macroeconomic conditions such as inflation and foreign exchange fluctuations, cannot be determined with certainty and therefore require increased judgment. These estimates and assumptions may change in future periods and will be recognized in the condensed consolidated financial statements as new events occur and additional information becomes known. To the extent the Company's actual results differ materially from those estimates and assumptions, the Company's future financial statements could be affected.

#### **Significant Accounting Policies**

There have been no material changes to the Company's significant accounting policies described in the Company's Annual Report on Form 10-K for the fiscal year ended January 31, 2024, that have had a material impact on its condensed consolidated financial statements and related notes.

#### **Deferred Commissions**

The Company capitalizes certain sales commissions, including related payroll taxes, earned by the Company's sales force, which are considered to be incremental costs that would not be incurred absent of the contract. On an annual basis, the Company assesses the expected period of benefit by taking into consideration its customer contracts, its technology and duration of customer relationships. Based on our most recent assessment, the Company determined the expected period of benefit for incremental costs of customer contracts should be increased from three to four years. This change in accounting estimate was effective February 1, 2024 and is accounted for prospectively in the Company's unaudited condensed consolidated financial statements. Commissions for renewal contracts are not commensurate with the commission paid for initial acquisition of a contract and are amortized based over the related contractual renewal period. This change in the period of benefit did not have a material impact to our unaudited condensed consolidated financial statements.

### **Concentration of Credit Risk**

Financial instruments that potentially subject the Company to credit risk primarily consist of cash, cash equivalents, restricted cash, short-term investments and accounts receivable. The Company maintains its cash and cash equivalents, restricted cash and short-term investments with high-quality financial institutions. Cash equivalents consist of money market funds which are invested through financial institutions in the United States ("U.S."). Deposits, including those held in foreign branches of global banks, may exceed the amount of insurance provided on such deposits. The Company has not experienced any losses on these deposits.

For its accounts receivable, the Company is exposed to credit risk in the event of nonpayment by customers to the extent of the amounts recorded on the consolidated balance sheet. Generally, credit risk with respect to accounts receivable is diversified due to the number of entities comprising the Company's customer base and their dispersion across different geographies and industries. The Company performs ongoing credit evaluations on certain customers and generally does not require collateral on accounts receivable. The Company maintains an allowance for credit losses and historically bad debts have not been material.

No customer accounted for 10% or more of total revenue for the three months ended April 30, 2024 and 2023. Two customers accounted for approximately 16% and 13% of gross accounts receivable as of April 30, 2024, respectively. No customer accounted for 10% or more of gross accounts receivable as of January 31, 2024.

# **Recent Accounting Pronouncements**

#### Recently Adopted Accounting Pronouncements

There were no significant changes to the Company's significant accounting policies disclosed in "Note 2 – Basis of Presentation and Summary of Significant Accounting Policies" of the Company's Annual Report.

#### Recently Issued Accounting Pronouncements

In November 2023, the FASB issued ASU No. 2023-07, Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures ("ASU 2023-07"), which is intended to improve reportable segment disclosure requirements. ASU 2023-07 expands segment disclosures by requiring disclosure of significant segment expenses that are regularly provided to the chief operating decision maker and included within each reported measure of segment profit or loss, an amount and description of its composition for other segment items, and interim disclosures of a reportable segment's profit or loss and assets. Additionally, the amendments require disclosure of the title and position of the CODM and an explanation of how the CODM uses the reported measure(s) of segment profit or loss in assessing segment performance and deciding how to allocate resources. All disclosure requirements of ASU 2023-07 are required for all entities including entities with a single reportable segment. This ASU is effective for fiscal years beginning after December 15, 2023, and interim periods within fiscal years beginning after December 15, 2024. Early adoption is permitted. The Company is in the process of evaluating the impact of the adoption on its consolidated financial statements and related disclosures.

In December 2023, the FASB issued ASU No. 2023-09, *Income Taxes (Topic 740): Improvements to Income Tax Disclosures* ("ASU 2023-09"), which requires companies to disclose additional information about income taxes, primarily their rate reconciliation information and income taxes paid. The new guidance requires companies to disclose in their rate

reconciliation table additional categories of information about federal, state and foreign income taxes and to provide more details about the reconciling items in some categories if the items meet a quantitative threshold. Additionally companies will be required to disclose annually income taxes paid (net of refunds received) disaggregated by federal (national), state and foreign taxes and to disaggregate the information by jurisdiction based on a quantitative threshold. The guidance is effective for the Company for the fiscal year ending January 31, 2026, and early adoption is permitted. The Company is currently evaluating the impact of ASU 2023-09 on its consolidated financial statements and related disclosures.

Under the Jumpstart Our Business Startups Act (the "JOBS Act"), the Company meets the definition of an emerging growth company and can delay adopting new or revised accounting standards issued after the enactment of the JOBS Act until such time as those standards apply to private companies. The Company has elected to use this extended transition period for complying with new or revised accounting standards that have different effective dates for public and private companies until the Company is no longer an emerging growth company or until the Company affirmatively and irrevocably opts out of the extended transition period.

Ac of April 20, 2024

# 3. Cash Equivalents and Short-Term Investments

The following tables summarize the Company's cash equivalents and short-term investments (in thousands):

		As of Apr	11 30, 2	2024		
Amortized Cost		Gross Unrealized Gains		Gross Unrealized Losses		Estimated Fair Value
\$ 37,117	\$	_	\$	_	\$	37,117
37,117				_		37,117
	_					
83,723		1		(157)		83,567
13,804		2		(49)		13,757
5,999		_		(12)		5,987
4,909		_		_		4,909
108,435		3		(218)		108,220
\$ 145,552	\$	3	\$	(218)	\$	145,337
\$	\$ 37,117 37,117 83,723 13,804 5,999 4,909 108,435	\$ 37,117 \$ 3	Amortized Cost   Gross Unrealized Gains	Amortized Cost   Gross Unrealized Gains	Amortized Cost   Unrealized Gains   Unrealized Losses	Amortized Cost   Unrealized Gains   Unrealized Losses

		As of Janua	ary 31, 2024	
	Amortized Cost	Gross Unrealized Gains	Gross Unrealized Losses	Estimated Fair Value
Cash Equivalents				
Money market funds	\$ 32,895	\$ _	\$ —	\$ 32,895
Total cash equivalents	32,895	_		32,895
Short-Term Investments				
U.S. government treasury securities	85,525	58	(7)	85,576
Corporate debt securities	13,712	14	(1)	13,725
U.S. government agency securities	7,999	_	(8)	7,991
Commercial paper	4,845	_	_	4,845
Asset-backed securities	144	_	_	144
Total short-term investments	112,225	72	(16)	112,281
Total	\$ 145,120	\$ 72	\$ (16)	\$ 145,176

During the three months ended April 30, 2024 and 2023, the Company did not reclassify any amounts to earnings from accumulated other comprehensive income (loss) related to unrealized gains or losses in other income (expense), net in the condensed consolidated statements of operations.

As of April 30, 2024, the Company's short-term investments consisted of \$88.8 million and \$19.4 million with a contractual maturity date of less than one year and greater than one year, respectively. As of January 31, 2024, the Company's short-term investments consisted of \$97.6 million and \$14.7 million with a contractual maturity of less than one year and greater than one year, respectively.

The Company's gross unrealized losses and fair values for short-term investments that were in an unrealized loss position as of April 30, 2024 and January 31, 2024 aggregated by investment category and the length of time that individual securities have been in a continuous loss position were as follows (in thousands):

					As of Apr	il 30	, 2024				
		Less Than	12 N	Ionths	12 Months	or (	Greater	Total			
	Un	realized Losses		Fair Value	Unrealized Losses		Fair Value	Unrealized Losses		Fair Value	
U.S. government treasury securities	\$	(157)	\$	77,620	\$ 	\$	_	\$ (157)	\$	77,620	
U.S. government agency securities		(12)		5,987	_		_	(12)		5,987	
Corporate debt securities		(49)		9,826	_		_	(49)		9,826	
Total	\$	(218)	\$	93,433	\$ 	\$	_	\$ (218)	\$	93,433	

						As of Janua	ary 3	31, 2024			
	Less Than 12 Months					12 Months	or C	Greater	To	otal	
	Unrea	lized Losses		Fair Value		Unrealized Losses		Fair Value	Unrealized Losses		Fair Value
U.S. government treasury securities	\$	(7)	\$	22,746	\$	_	\$	_	\$ (7)	\$	22,746
U.S. government agency securities		(8)		7,991		_		_	(8)		7,991
Corporate debt securities		(1)		5,008		_		_	(1)		5,008
Asset-backed securities		_		_		_		144	_		144
Total	\$	(16)	\$	35,745	\$		\$	144	\$ (16)	\$	35,889

As of April 30, 2024, the Company had 19 short-term investments in an unrealized loss position. As of January 31, 2024, the Company had 13 short-term investments in an unrealized loss position. As of April 30, 2024, the Company determined that the declines in the market value of its investment portfolio were not driven by credit related factors. During the three months ended April 30, 2024, there were no credit or non-credit related impairment charges recorded by the Company.

# 4. Fair Value Measurements

The Company accounts for certain of its financial assets and liabilities at fair value. Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability (an exit price) in an orderly transaction between market participants at the reporting date. The accounting guidance establishes a three-tiered hierarchy, which prioritizes the inputs used in the valuation methodologies in measuring fair value as follows:

- Level 1: Observable inputs, such as quoted prices in active markets for identical assets or liabilities.
- Level 2: Observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities, quoted prices in markets that are not active, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.
  - Level 3: Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The carrying amounts reflected on the condensed consolidated balance sheets for cash equivalents, accounts receivable, accounts payable and accrued liabilities approximate fair value due to the short maturities of those instruments.

The following tables present the fair value hierarchy for the Company's assets measured at fair value on a recurring basis (in thousands):

		As of April 30, 2024			
		Level 1	Level 2		Total
Cash Equivalents					
Money market funds	\$	37,117	\$	\$	37,117
Total cash equivalents		37,117	_		37,117
Short-Term Investments		_			
U.S. government treasury securities		_	83,567		83,567
Corporate debt securities		_	13,757		13,757
U.S. government agency securities		_	5,987		5,987
Commercial paper			4,909		4,909
Total short-term investments		_	108,220		108,220
Total	\$	37,117	\$ 108,220	\$	145,337
			As of January 31, 2024		
		Level 1	Level 2		
					Total
Cash Equivalents		Level 1	Level 2		Total
Cash Equivalents Money market funds	<u> </u>			\$	
Money market funds	<u>\$</u>	32,895	\$ —	\$	32,895
•	<u>\$</u>			\$	
Money market funds Total cash equivalents Short-Term Investments	<u>\$</u>	32,895	\$ <u> </u>	\$	32,895 32,895
Money market funds Total cash equivalents	<u>\$</u>	32,895		\$	32,895
Money market funds Total cash equivalents  Short-Term Investments U.S. government treasury securities	<u>\$</u>	32,895	\$ — — 85,576	\$	32,895 32,895 85,576
Money market funds Total cash equivalents Short-Term Investments U.S. government treasury securities Corporate debt securities	\$	32,895	\$ — 85,576 13,725	\$	32,895 32,895 85,576 13,725
Money market funds Total cash equivalents  Short-Term Investments U.S. government treasury securities Corporate debt securities U.S. government agency securities	\$	32,895	\$ — 85,576 13,725 7,991	\$	32,895 32,895 85,576 13,725 7,991
Money market funds Total cash equivalents  Short-Term Investments U.S. government treasury securities Corporate debt securities U.S. government agency securities Commercial paper	\$	32,895	\$ —  85,576 13,725 7,991 4,845	\$	32,895 32,895 85,576 13,725 7,991 4,845

The Company classifies its money market funds within Level 1 of the fair value hierarchy because they are valued based on quoted market prices in active markets. The Company classifies its U.S. government agency securities, asset-backed securities, commercial paper, U.S. government treasury securities, and corporate debt securities within Level 2 because they are valued using inputs other than quoted prices which are directly or indirectly observable in the market, including readily available pricing sources for the identical underlying security which may not be actively traded.

# 5. Balance Sheet Components

# **Prepaid Expenses and Other Current Assets**

Prepaid expenses and other current assets consisted of the following (in thousands):

	As	of April 30, 2024	As o	f January 31, 2024
Prepaid expenses	\$	4,720	\$	4,793
Prepaid software		4,390		4,429
Other current assets		1,197		1,163
Total prepaid expenses and other current assets	\$	10,307	\$	10,385

# Property and Equipment, Net

Property and equipment, net consisted of the following (in thousands):

	As of April 30, 2024		As	s of January 31, 2024
Computer equipment	\$	3,744	\$	3,736
Furniture and fixtures		418		418
Capitalized internal-use software		8,743		8,743
Leasehold improvements		1,903		1,903
Construction in progress—capitalized internal-use software		3,749		2,571
Total gross property and equipment		18,557		17,371
Accumulated depreciation and amortization		(12,444)		(12,044)
Total property and equipment, net	\$	6,113	\$	5,327

Depreciation and amortization expense was \$0.4 million and \$0.9 million for three months ended April 30, 2024 and 2023, respectively. Included in these amounts were the amortization of capitalized internal-use software development costs of \$0.2 million and \$0.7 million in the three months ended April 30, 2024 and 2023, respectively.

# **Accrued Compensation and Benefits**

Accrued compensation and benefits consisted of the following (in thousands):

	As o	of April 30, 2024	As o	of January 31, 2024
Accrued bonus	\$	2,784	\$	7,056
Accrued commissions		2,150		4,852
Accrued payroll and benefits		3,304		4,690
Employee contributions under the ESPP		701		1,518
Total accrued compensation and benefits	\$	8,939	\$	18,116

# Other Accrued Liabilities

Other accrued liabilities consisted of the following (in thousands):

	As	of April 30, 2024	As	s of January 31, 2024
Accrued professional fees	\$	937	\$	1,190
Sales and value added tax payable		287		517
Income taxes payable		137		173
Other		2,453		2,701
Total other accrued liabilities	\$	3,814	\$	4,581

# 6. Deferred Revenue and Remaining Performance Obligations

The following table presents the deferred revenue balances (in thousands):

	As of April 30,	As of January 31,
	2024	2024
Deferred revenue, current	\$ 89,619	\$ 81,736
Deferred revenue, noncurrent	2,572	2,747
Total deferred revenue	\$ 92,191	\$ 84,483

Changes in the deferred revenue balances during the three months ended April 30, 2024 and 2023 were as follows (in thousands):

	Three Months Ended April 30,			
		2024		2023
Beginning balance	\$	84,483	\$	74,991
Performance obligations satisfied during the period that were included in the deferred revenue balance at the beginning of the year		(34,359)		(29,061)
Increases due to invoicing prior to satisfaction of performance obligations		42,067		35,484
Ending balance	\$	92,191	\$	81,414

Remaining performance obligations ("RPOs") represent the amount of contracted future revenue that has not yet been recognized, including both deferred revenue and non-cancelable contracted amounts that will be invoiced and recognized as revenue in future periods.

As of April 30, 2024, the Company's RPOs were \$220.0 million. The Company expects to recognize revenue of \$137.0 million of these remaining performance obligations over the next twelve months with the remaining balances recognized thereafter.

#### 7. Debt

# Loan and Security Agreement

On February 7, 2024, the Company entered into a loan and security agreement with MUFG Bank, Ltd., as lender, for a three-year senior secured revolving loan facility of up to \$25.0 million, including a letter of credit sublimit of up to \$5.0 million, with an original maturity date on February 7, 2027 and a uncommitted accordion feature that provides up to \$25.0 million of additional borrowing capacity (the "Credit Facility"). Borrowings under the Credit Facility accrue interest at a floating per annum rate based on secured overnight financing rate ("SOFR"), plus 3.0% for the applicable interest rate period. Accrued interest on the Credit Facility will be paid at the end of the applicable interest rate period, but at least every three months. The Company is also obligated to pay other customary fees and expenses, including an unused revolving line facility fee of 0.25% per annum of the average daily unused portion of the Credit Facility.

Under the Credit Facility, the Company is subject to a minimum consolidated adjusted EBITDA covenant, tested quarterly. The Credit Facility also contains certain customary affirmative and negative covenants as well as customary events of default, subject to certain exceptions, including restrictions on the ability of the Company and its subsidiaries to, among other things, incur debt, grant liens, make acquisitions, suffer changes in control, make investments, make certain dividends or distributions, repurchase or redeem stock, dispose of or transfer assets, and enter into transactions with affiliates, in each case, subject to customary and other agreed limitations and exceptions. To date, the Company has an outstanding letter of credit of \$0.5 million against the \$5.0 million sublimit in connection with a lease arrangement.

The Company did not have any debt outstanding under the Credit Facility as of April 30, 2024 and was in compliance with the financial covenants associated with the Credit Facility as of April 30, 2024.

# 8. Leases

The Company leases facilities under non-cancelable operating leases, primarily for rent of office space. The Company's leases have various expiration dates through November 2027, some of which include options to extend the leases for up to three years. The Company does not have any finance leases.

The components of lease costs were as follows (in thousands):

		Three Months Ended April 30,			
	_	202	24		2023
Operating lease costs	5	\$	765	\$	772
Variable lease costs		\$	157	\$	146

Short-term lease costs were immaterial during the three months ended April 30, 2024 and 2023.

The following table presents supplemental cash flow information related to leases (in thousands):

	Ti	Three Months Ended April 30,		
	20	124	2023	
Cash paid for amounts included in the measurement of lease liabilities:				
Operating cash outflows from operating leases	\$	843 \$	823	
Right-of-use assets obtained in exchange for lease obligations:				
Operating leases	\$	— \$	_	

The following table presents supplemental balance sheet information related to operating leases (in thousands, except for lease term and discount rate):

	Ap	oril 30, 2024	Jar	nuary 31, 2024
Operating lease right-of-use assets	\$	4,135	\$	4,848
Operating lease liabilities	\$	3,106	\$	3,208
Operating lease liabilities, noncurrent		1,388		2,078
Total operating lease liabilities	\$	4,494	\$	5,286
Weighted-average remaining lease term		1.8 years		2.0 years
Weighted-average discount rate		4.4 %		4.4 %

As of April 30, 2024, remaining maturities of operating lease liabilities were as follows (in thousands):

Period	Operating Leases
Remaining for Fiscal 2025	\$ 2,514
Fiscal 2026	1,481
Fiscal 2027	437
Fiscal 2028	245
Fiscal 2029 and thereafter	_
Total lease payments	4,677
Less: imputed interest	(183)
Total	\$ 4,494

### 9. Commitments and Contingencies

### **Other Contractual Commitments**

Other contractual commitments relate to third-party cloud infrastructure agreements and subscription arrangements.

There were no material contractual commitments that were entered into during the three months ended April 30, 2024 that were outside the ordinary course of business.

# Legal Matters

From time to time, the Company may be a party to various legal matters, threatened claims, or proceedings in the normal course of business. Legal fees and other costs associated with such actions are expensed as incurred. The Company assesses, in conjunction with its legal counsel, the need to record a liability for litigation and contingencies. Legal accruals are recorded when and if it is determined that a loss related to a certain matter is both probable and reasonably estimable. The Company is not currently a party to any legal proceedings that, if determined adversely to it, would, in management's opinion, have a material and adverse effect on the Company's financial condition, results of operations, or cash flows.

# **Indemnification Agreements**

In the ordinary course of business, the Company enters into agreements of varying scope and terms pursuant to which the Company agrees to indemnify customers, vendors, lessors and other business partners with respect to certain matters, including, but not limited to, losses arising out of the breach of such agreements, services to be provided by the Company or from intellectual property infringement claims made by third parties. Additionally, the Company entered into

indemnification agreements with the Company's directors and officers that require the Company, among other things, to indemnify them against certain liabilities that may arise from their services as a director or executive officer in any capacity as the Company's director, trustee, general partner, managing member, officer, employee, agent or fiduciary or with respect to any employee benefit plans. The Company has not incurred material costs to defend lawsuits or settle claims related to these indemnification agreements nor are we aware of any such claims that could reasonably be expected to incur material costs.

#### 10. Stockholders' Equity and Employee Incentive Plans

# Redeemable Convertible Preferred Stock

As of April 30, 2024, there were no shares of redeemable convertible preferred stock issued and outstanding.

The Company's Amended and Restated Certificate of Incorporation authorizes the issuance of 200,000,000 shares of undesignated preferred stock with a par value of \$0.00001 per share with rights and preferences, including voting rights, designated from time to time by the board of directors.

#### Common Stock

The Company's Amended and Restated Certificate of Incorporation authorized the Company to issue 1,000,000,000 shares of common stock at a par value of \$0.00001 as of April 30, 2024 and January 31, 2024.

Each share of common stock is entitled to one vote. The holders of common stock are also entitled to receive dividends whenever funds are legally available and when and if declared by the board of directors, subject to the prior rights of holders of all classes of stock outstanding. As of April 30, 2024 and January 31, 2024, no dividends had been declared.

As of April 30, 2024, the Company has reserved common stock for future issuance as follows:

	Number of Shares
Stock options outstanding	5,465,173
Restricted stock units issued and outstanding	5,930,054
Remaining shares available for issuance under the 2021 Plan	3,565,156
Shares available for issuance under the 2023 Inducement Plan	955,779
ESPP	1,606,656
Common stock warrants	105,350
Total	17,628,168

#### Common Stock Warrants

In April, 2019, the Company issued warrants in connection with a term loan agreement with a certain lender to purchase 105,350 shares of the Company's common stock at \$7.48 per share, exercisable over 10 years. As of April 30, 2024, all warrants were outstanding and exercisable.

# Stock Options

The following table summarizes stock option activity under the Stock Plans for the three months ended April 30, 2024 (aggregate intrinsic value in thousands):

	Options O	utsta	nding		
	Number of Options		Weighted- Average Exercise Price	Weighted- Average Contractual Term	Aggregate Intrinsic Value
Balances as of January 31, 2024	5,889,938	\$	10.38	4.84	\$ 86,742
Options exercised	(406,055)	\$	8.11		
Options cancelled	(18,710)	\$	24.78		
Balances as of April 30, 2024	5,465,173	\$	10.50	4.55	\$ 75,355
Options vested and expected to vest as of April 30, 2024	5,465,173	\$	10.50	4.55	\$ 75,355
Options vested and exercisable as of April 30, 2024	5,152,472	\$	9.80	4.41	\$ 74,479

No stock options were granted during the three months ended April 30, 2024 and 2023.

The aggregate intrinsic value of options exercised during the three months ended April 30, 2024 and 2023 was \$8.1 million and \$4.2 million, respectively. Aggregate intrinsic value represents the difference between the exercise price of the options and the estimated fair value of the Company's common stock.

The Company recognized stock-based compensation expense related to stock options of \$0.9 million and \$1.2 million during the three months ended April 30, 2024 and 2023, respectively.

As of April 30, 2024, there was \$2.6 million of unrecognized stock-based compensation expense related to unvested stock options, which is expected to be recognized over a weighted-average period of 0.8 years.

#### Service-Based RSUs

During the year ended January 31, 2022, the Company began granting RSUs to its employees which have service-based vesting conditions. The service-based vesting condition for these awards is generally satisfied by rendering continuous service over two to four years, depending on the award, during which time the grants will vest either quarterly or after a one-year cliff with quarterly vesting thereafter.

The following table is a summary of service-based RSU activity for the three months ended April 30, 2024:

	RSUs Ou	3,909,920 \$ 17.56 1,743,216 \$ 27.48			
	Number of RSUs	We Gran	ıt Date Fair Value		
Balances as of January 31, 2024	3,909,920	\$	17.56		
RSUs granted	1,743,216	\$	27.48		
RSUs vested	(531,351)	\$	17.32		
RSUs forfeited	(137,331)	\$	18.93		
Balances as of April 30, 2024	4,984,454	\$	21.02		

The aggregate fair value of the RSU awards granted was \$47.9 million and \$39.3 million during the three months ended April 30, 2024 and 2023, respectively. This represents the fair value of the common stock on the date the service-based vesting awards were granted.

We recognized \$11.6 million and \$6.8 million in stock-based compensation expense related to service vesting-based RSUs during the three months ended April 30, 2024 and 2023, respectively. As of April 30, 2024, there was \$93.6 million of unrecognized compensation expense related to service-based RSUs expected to be recognized over a weighted-average vesting period of 1.8 years.

#### Performance-based and Market-based Awards

# Performance-based Awards

We recognized a total of \$1.3 million and \$0.7 million in stock-based compensation expense related to PSUs during the three months ended April 30, 2024 and 2023, respectively. As of April 30, 2024, there were 840,000 awards outstanding and a total of \$4.7 million of unrecognized compensation expense related to PSUs expected to be recognized over an average vesting period of 1.4 years.

#### Market-based Awards

The Company recognized a total of \$0.3 million and \$0.2 million in stock-based compensation expense related to market-based awards during the three months ended April 30, 2024 and 2023, respectively. There were 79,200 awards vested during the three months ended April 30, 2024. As of April 30, 2024, there were 105,600 awards outstanding and a total of \$0.5 million of unrecognized compensation expense related to market-based RSUs expected to be recognized over an average vesting period of 0.9 years.

# Determination of Fair Value

The Company estimates the fair value of purchase rights issued to employees under the ESPP using the Black-Scholes option-pricing model, which is dependent upon several variables, such as the fair value of the Company's common stock, the expected option term, expected volatility of the Company's stock price over the expected term, expected risk-free interest rate over the expected option term, and expected dividend yield.

The fair value of employee stock purchase rights for the offering period under the 2021 ESPP was determined on the date of grant using the Black-Scholes option-pricing model with the following weighted-average assumptions:

	Three Months Ende	ed April 30,
	2024	2023
Employee Stock Purchase Plan:		
Expected term (in years)	1.2	0.8
Expected volatility	60.1 %	66.9 %
Risk-free interest rate	5.0 %	4.8 %
Dividend yield	_	_

The Company recognized stock-based compensation expense related to the ESPP of \$0.6 million and \$0.3 million during the three months ended April 30, 2024 and 2023, respectively. As of April 30, 2024, \$2.4 million of unrecognized stock-based compensation expense related to the ESPP is expected to be recognized over a weighted-average vesting period of 1.0 year.

During the three months ended April 30, 2024, 123,778 shares of common stock were issued under the ESPP.

# Stock-Based Compensation

Stock-based compensation expense, net of amounts capitalized was as follows (in thousands):

	Three Months Ended April 30,			
	2024		2023	
Cost of revenue—subscription	\$ 266	\$	193	
Cost of revenue—services	141		145	
Research and development	3,993		2,768	
Sales and marketing	5,223		3,241	
General and administrative	5,004		2,928	
Restructuring	_		1	
Total stock-based compensation expense	\$ 14,627	\$	9,276	

#### 11. Income Taxes

Accounting for income taxes for interim periods generally requires the provision for income taxes to be determined by applying an estimate of the annual effective tax rate for the full fiscal year to income or loss before income

taxes, adjusted for discrete items, if any, for the reporting period. The Company updates its estimate of the annual effective tax rate each quarter and records a cumulative adjustment in such period.

The Company recorded an immaterial amount of income tax expense for the three months ended April 30, 2024 and \$0.8 million for the three months ended April 30, 2023, respectively. Income tax expense consists primarily of income taxes in foreign jurisdictions in which the Company conducts business. Due to the Company's history of losses in the U.S., a full valuation allowance on the Company's domestic deferred tax assets, including net operating loss carryforwards, research and development tax credits, capitalized research and development, and other book versus tax differences was maintained. The Company has deferred tax attributes for stock-based compensation and fixed assets in the United Kingdom, and has not recorded a valuation allowance on the deferred tax attributes as of April 30, 2024. The Company will continue to evaluate for any future developments.

### 12. Geographic Information

The following table depicts the disaggregation of revenue by geographic area based on the billing address of the customers (in thousands):

	Three Months	Three Months Ended April 30,  2024  2023  34,682 \$ 25,991  16,645 15,005		
	2024		2023	
United States	\$ 34,682	\$	25,991	
International	16,645		15,005	
Total	\$ 51,327	\$	40,996	

No individual foreign country contributed 10% or more of total revenue for the three months ended April 30, 2024 and 2023.

As of April 30, 2024 and January 31, 2024, the majority of the Company's long-lived assets, including operating lease ROU assets, were located in the United States.

#### 13. Net Loss per Share

Basic net loss per share attributable to the Company's common stockholders is computed by dividing the net loss attributable to the Company's common stockholders by the weighted-average number of shares of common stock outstanding during the period. Diluted net loss per share is the same as basic net loss per share for all years presented because the effects of potentially dilutive items were anti-dilutive given the Company's net loss position in each period presented.

The following table presents the calculation of basic and diluted net loss per share attributable to common stockholders (in thousands, except per share data):

		Three Months	Ended A	pril 30,
		2024		2023
Numerator				
Net loss	\$	(20,995)	\$	(21,875)
Denominator	_			
Weighted-average shares used in computing net loss per share, basic and diluted		49,788		45,843
Net loss per share, basic and diluted	\$	(0.42)	\$	(0.48)

The following potentially dilutive securities were excluded from the computation of diluted net loss per share for the periods presented because the impact of including them would have been anti-dilutive (in thousands):

	As of A	pril 30,
	2024	2023
Stock options	5,465	7,388
RSUs	5,886	6,983
Employee stock purchase rights under the ESPP	141	118
Common stock warrants	105	105
Total	11,597	14,594

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis of our financial condition and results of operations together with our condensed consolidated financial statements and the related notes and other financial information included elsewhere in this Quarterly Report on Form 10-Q. Some of the information contained in this discussion and analysis or set forth elsewhere in this Quarterly Report on Form 10-Q, including information with respect to our plans and strategy for our business, includes forward-looking statements that involve risks and uncertainties. You should review the sections titled "Special Note Regarding Forward-Looking Statements" and "Risk Factors" for a discussion of forward-looking statements and important factors that could cause actual results to differ materially from the results described in or implied by the forward-looking statements contained in the following discussion and analysis. The last day of our fiscal year is January 31. Our fiscal quarters end on April 30, July 31, October 31 and January 31. Our fiscal years ended January 31, 2023, 2024 and 2025 are referred to herein as fiscal 2023, fiscal 2024 and fiscal 2025, respectively.

#### Overview

Couchbase provides a leading cloud database platform for modern applications. Our mission is to simplify how developers and architects develop, deploy and consume modern applications that span cloud, edge and everything in between. Enterprises rely on Couchbase to cost-effectively power the core applications their businesses depend on with the highest performance, reliability, scalability and versatility requirements for which there is no tolerance for disruption or downtime. Any compromise of these requirements could cause these applications to fail—stopping or delaying package delivery for shipping companies, interrupting reservations for travel companies or causing product shortages in stores for retailers.

Our database is versatile and works in multiple configurations, from fully-managed cloud to multi- or hybrid-cloud, to on-premises environments, and beyond the edge. We have architected our database to fuse the trusted strengths of relational databases with the flexibility, performance and scale of many NoSQL systems, across the cloud. Our database platform serves the needs of both enterprise architects and application developers. Combined with our performance at scale, we believe this power enables customers to run their most important applications with the effectiveness they require, with the efficiency they desire and in the modern infrastructure environments they demand.

With nearly every aspect of our lives being transformed by digital innovation, enterprises are charged with building applications that enable delightful and meaningful customer experiences. Enterprises are increasingly reliant on applications, which in turn rely on databases to store, retrieve and operationalize data into action. Today, applications are operating at a scale, speed and dynamism unheard of just a decade ago. There is an increasing diversity of application types, modalities and delivery and consumption models, and the volume, velocity and variety of data on which they rely is growing at an exponential rate. Looking forward, there is potential for AI to drastically transform business and the nature of modern applications as a whole. Consequently, the demand on enterprises and their databases is growing exponentially.

While legacy database technologies were built to the highest performance and reliability requirements of their generation, they are approaching the limits for which they were designed. The underlying architecture of these technologies has not changed significantly, while the requirements of the applications they need to support are changing dramatically, especially with the emergence of generative AI which relies heavily on real-time data. Legacy database technologies are buckling under the pressure of digital transformation, as they were not built to update and respond in microseconds, enable rich, customized user experiences and perform without latency.

We designed Couchbase to give enterprises a database for the modern cloud world. Our platform combines the best capabilities of a relational database, like SQL transactions and ACID guarantees, with the flexibility and scalability of a NoSQL database. This allows enterprises to confidently accelerate strategic initiatives such as more quickly moving business-critical applications into the cloud, improving application flexibility and increasing developer agility. For our customers, we facilitate a seamless transition from legacy relational databases to our modern cloud database platform resulting in better application scalability, user experience and security at the pace that works for them. We believe our unique architecture is also well-suited to power AI applications which require exceptionally high performance and scalability. We deliver this cloud database platform both as a customer-managed product and as a fully-managed DBaaS that is managed by Couchbase. Our DBaaS, called Couchbase Capella, supports a broad set of use cases, reducing a customer's need to buy, deploy and manage additional databases or supporting technologies.

We sell our platform through our direct sales force and our ecosystem of partners. Our platform is broadly accessible to a wide range of enterprises, as well as governments and organizations. We have customers in a range of industries, including retail and e-commerce, travel and hospitality, financial services and insurance, software and technology, gaming, media and entertainment and industrials. We focus our selling efforts on the largest global enterprises with the most complex data requirements, and we have cloud-based, fully managed offerings for enterprises looking for a turnkey version of our platform.

We have achieved significant growth over our operating history. For the three months ended April 30, 2024 and 2023, our revenue was \$51.3 million and \$41.0 million, respectively, representing period-over-period growth rate of 25%. As of April 30, 2024 and 2023, our annual recurring revenue ("ARR") was \$207.7 million and \$172.2 million, respectively, representing period-over-period growth of 21%. For the three months ended April 30, 2024 and 2023, our net loss was \$21.0 million and \$21.9 million, respectively, as we continued to invest in the growth of our business to capture the massive opportunity that we believe is available to us.

# **Our Business Model**

We generate the substantial majority of our revenue from sales of subscriptions, which accounted for 96% and 94% of our total revenue for the three months ended April 30, 2024 and 2023, respectively. We derive a substantial majority of our subscription revenue from the Enterprise Edition of Couchbase Server and Couchbase Mobile. Couchbase Server is generally licensed per node, which we define as an instance of Couchbase running on a server. Our subscription pricing is based on the computing power and memory per instance, as well as the chosen service level. We offer three different support levels: the Platinum level offers 24/7 support and the shortest response time of 30 minutes; the Gold level offers 24/7 support with a response time of 2 hours; and the Silver level offers 7am-5pm local time support, 5 days a week. These response times are for incidents of the highest severity level, which we identify as level P1. The initial response time for levels P2 and P3 incidents, which are less severe, are longer.

We also derive subscription revenue from our DBaaS offering. Our DBaaS offering, called Couchbase Capella, is sold on a consumption basis, which removes the need to license different node types separately. Couchbase Capella pricing delivers superior customer flexibility relative to other Cloud Service Providers ("CSPs") as on-demand pricing allows customers to pay only for what they use based on hourly pricing and the credits purchased through our annual credit model expire only at the end of a 12-month period, rather than ratably throughout the year. We also provide automatic conversion to on-demand consumption when annual credits expire or are exhausted. Couchbase Capella credits can be purchased upfront to provide cost savings with volume discounts available based on credit quantity. We offer three pricing levels for Couchbase Capella, based on the support response time.

The non-cancelable term of our subscription arrangements typically ranges from one to three years but may be longer or shorter in limited circumstances and is typically billed annually in advance. The timing and billing of large, multi-year contracts can create variability in revenue and deferred revenue between periods.

We also generate revenue from services, which represented 4% and 6% of our total revenue for the three months ended April 30, 2024 and 2023, respectively. Our services revenue is derived from our professional services related to the implementation or configuration of our platform and training. We have invested in building our services organization because we believe it plays an important role in customer success, ensuring that our customers fulfill their digital transformation agendas while leveraging our platform, accelerating our customers' realization of the full benefits of our platform and driving increased adoption of our platform.

Our go-to-market strategy is focused on organizations that are modernizing existing applications or building net new applications. As an example, for large enterprises recognized as leaders in their respective industries, this could mean attempting to solve complicated business problems by digitally transforming their operations. For mid-size companies to start ups, this could be building a new product or service that seeks to disrupt an established market. As a result, Couchbase powers a wide variety of applications across a broad array of industries, from some of the largest and most complex enterprise applications worldwide to the next generation of personalized dynamic apps. Through our highly instrumented "sell-to" go-to-market motion, we have built a direct sales organization that understands the strategic needs of enterprises as well as a marketing organization that emphasizes our enablement of digital transformation through our no-compromises approach to performance, resiliency, scalability, agility and total cost of ownership ("TCO") savings.

We complement our "sell-to" go-to-market motion with a "buy-from" go-to-market motion, which is focused on targeting the application developer community to drive adoption of our platform. To accomplish this, we have and plan to continue to invest in Couchbase Capella. We also offer free Community Editions of some of our products, free trials of our Enterprise Edition of Couchbase Server and Couchbase Capella products and a web browser-based demonstration version of Couchbase Server to further accelerate application developer adoption. We believe these offerings lead to future purchases of our paid products. While our Community Edition includes the core functionality of Couchbase Server, it is not suited for mission-critical deployments, as it offers only limited functionality around the scaled performance and security that enterprises require and no direct customer support from Couchbase.

We also continuously grow and cultivate our cloud provider partner and technology provider ecosystem. A significant portion of our revenue in the three months ended April 30, 2024 and 2023 was attributable to our partner ecosystem.

We employ a land-and-expand model centered around our platform offerings, which have a rapid time to production and time to value for our customers, and our sales and customer success organizations, which proactively guide customers to realize strategic and transformative use cases and drive greater adoption of our platform and services. Our marketing organization is focused on building our brand reputation and awareness. Our marketing initiatives drive awareness and demand for Couchbase products, starting at the top of the sales funnel with trial experiences. As part of these efforts, we offer application developers robust educational resources including a robust and growing community to help them learn more about our platform, including access to on-demand instructional webinars.

# **Impact of Macroeconomic Conditions**

Current macroeconomic conditions, including recessionary fears, inflation concerns, and increased interest rates as a result of government actions to combat inflation, as well as other geopolitical developments, have impacted and may continue to impact business spending and the economy as a whole. We continue to see deal cycles that are consistent with previous quarters, along with an elevation in degree of budget scrutiny, slower than expected product migrations, lower than expected expansions, and customers electing to buy in smaller increments. Additionally, effects from foreign exchange fluctuations have impacted and may continue to impact our results of operations.

The effects of these macroeconomic conditions on our business and operations remains highly uncertain, and it is not possible for us to predict the duration and extent to which they will affect our business, future results of operations, and financial condition. See the section titled "Risk Factors" for further discussion of these challenges and risks.

#### **Factors Affecting Our Performance**

# Continuing to Acquire New Customers

We grow our subscription revenue by acquiring new customers. The size of our customer base may vary from period to period for several reasons, including the length of our sales cycle, the effectiveness of our sales and marketing efforts, enterprise application development cycles and the corresponding adoption rates of modern applications that require database solutions like ours. Additionally, our revenue has and will vary as new customers purchase our products due to the fact that we recognize a portion of such subscription revenue upfront. As digital transformation continues to accelerate, we believe that Couchbase Capella will become increasingly popular as a result of its compelling pricing model, ease of operation, lower TCO, time to market and flexibility. We will continue to offer Couchbase Capella and provide flexible, highly available and differentiated economical options to capture new customers.

# Continuing to Expand Within Existing Customers

A significant part of our growth has been, and we expect will continue to be, driven by expansion within our existing customer base. Growth of our revenue from our existing customers results from increases in the scale of their deployment for existing use cases, or when customers utilize our platform to address new use cases. In addition, our professional services organization helps customers deploy new use cases and optimize their existing implementations. Our revenue from our subscription offerings varies depending on the scale and performance requirements of our customers' deployments. We are focusing on growing our subscription revenue, particularly from enterprises, while delivering professional services and training to support this growth. We have been successful in expanding our existing customers' adoption of our platform as demonstrated by our dollar-based net retention rate of over 115% in the past eight quarters.

Our dollar-based net retention rate for any period equals the simple arithmetic average of our quarterly dollar-based net retention rate for the four quarters ending with the most recent fiscal quarter. To calculate our dollar-based net retention rate for a given quarter, we start with the ARR ("Base ARR") attributable to our customers ("Base Customers") as of the end of the same quarter of the prior fiscal year. We then determine the ARR attributable to the Base Customers as of the end of the most recent quarter and divide that amount by the Base ARR.

#### Continuing to Invest in Growth

We expect to continue to invest in our offerings, personnel, geographic presence and infrastructure in order to drive future growth, as well as to pursue adjacent opportunities. We expend research and development resources to drive innovation in our proprietary software to constantly improve the functionality and performance of our platform and to increase the deployment models available to our customers. We anticipate continuing to increase our headcount to ensure that our product development organization drives improvements in our product offerings, our sales and marketing organization can maximize opportunities for growing our business and revenue and our general and administrative organization efficiently supports the growth of our business as well as our effective operation as a public company.

#### **Key Business Metrics**

### Annual Recurring Revenue

We define ARR as of a given date as the annualized recurring revenue that we would contractually receive from our customers in the month ending 12 months following such date. Based on historical experience with customers, we assume all contracts will be renewed at the same levels unless we receive notification of non-renewal and are no longer in negotiations prior to the measurement date. For Capella products, ARR in a customer's initial year is calculated as the greater of: (i) initial year contract revenue as described above or (ii) annualized prior 90 days of actual consumption; and ARR for subsequent years is calculated with method (ii). ARR excludes services revenue.

Prior to fiscal 2025, ARR excluded on-demand revenue and, for Capella products in a customer's initial year, ARR was calculated solely on the basis of initial year contract revenue. The reason for these changes is to better reflect ARR where usage rates or timing of purchases may be uneven and to better align with how ARR is used to measure the performance of the business. ARR for prior periods has not been adjusted to reflect this change as it is not material to any period previously presented.

ARR should be viewed independently of revenue, and does not represent our revenue under GAAP on an annualized basis, as it is an operating metric that can be impacted by contract start and end dates and renewal dates. ARR is not intended to be a replacement for forecasts of revenue. Although we seek to increase ARR as part of our strategy of targeting large enterprise customers, this metric may fluctuate from period to period based on our ability to acquire new customers, expand within our existing customers and consumption dynamics. We believe that ARR is an important indicator of the growth and performance of our business.

As of April 30, 2024, ARR for Couchbase Capella products was approximately \$23.9 million.

As of April 30,	As of April 30,		
2023	2024		
(in millions)	(in mil		
207.7 \$ 172.2	\$ 207.7		

# Customers

We calculate our total number of customers, which also includes customers of Couchbase Capella, at the end of each period. Each customer account that has an active subscription contract with us or with which we are negotiating a renewal contract at the end of a given period is included in the calculation of total customer count, and the Couchbase Capella customer count includes customers who have a subscription of Couchbase Capella. Beginning in fiscal 2025, customers who used our products through an on-demand arrangement are included in the calculation of total customer count. The reason for this change is to align with our revised ARR methodology. As of April 30, 2024, we had 236 Couchbase Capella customers. Each party with which we enter into a subscription contract is considered a unique customer and, in some cases, a single organization may be counted as more than one customer. Our customer count is subject to

adjustments for acquisitions, consolidations, spin-offs and other market activity. We believe that our number of customers is an important indicator of the growth of our business and future revenue trends.

	As of A	pril 30,
	2024	2023
Customers	807	679

#### **Non-GAAP Financial Measures**

In addition to our financial information presented in accordance with GAAP, we believe certain non-GAAP financial measures are useful to investors in evaluating our operating performance. We use certain non-GAAP financial measures, collectively, to evaluate our ongoing operations and for internal planning and forecasting purposes. We believe that non-GAAP financial measures, when taken together with the corresponding GAAP financial measures, may be helpful to investors because they provide consistency and comparability with past financial performance and meaningful supplemental information regarding our performance by excluding certain items that may not be indicative of our business, results of operations or outlook. Non-GAAP financial measures are presented for supplemental informational purposes only, have limitations as analytical tools and should not be considered in isolation or as a substitute for financial information presented in accordance with GAAP and may be different from similarly-titled non-GAAP financial measures used by other companies. In addition, other companies, including companies in our industry, may calculate similarly-titled non-GAAP financial measures differently or may use other measures to evaluate their performance, all of which could reduce the usefulness of our non-GAAP financial measures as tools for comparison. Investors are encouraged to review the related GAAP financial measures and the reconciliation of these non-GAAP financial measures to their most directly comparable GAAP financial measures, and not to rely on any single financial measure to evaluate our business.

We define the non-GAAP financial measures below as their respective GAAP measures, excluding expenses related to stock-based compensation expense, employer payroll taxes on employee stock transactions, restructuring charges and impairment of capitalized internal-use software. We use these non-GAAP financial measures in conjunction with GAAP measures to assess our performance, including in the preparation of our annual operating budget and quarterly forecasts, to evaluate the effectiveness of our business strategies and to communicate with our board of directors concerning our financial performance.

Beginning with the fourth quarter of fiscal 2024, we have excluded the impairment of capitalized internal-use software, a non-cash operating expense, from our non-GAAP results as it is not reflective of ongoing operating results. This impairment charge relates to certain previously capitalized internal-use software that we determined would no longer be placed into service. Prior period non-GAAP financial measures have not been adjusted to reflect this change as we did not incur impairment of capitalized internal-use software in any prior period presented.

# Non-GAAP Gross Profit and Non-GAAP Gross Margin

We define non-GAAP gross profit and non-GAAP gross margin as gross profit and gross margin, respectively, excluding stock-based compensation expense and employer taxes on employee stock transactions. We use non-GAAP gross profit and non-GAAP gross margin in conjunction with GAAP financial measures to assess our performance, including in the preparation of our annual operating budget and quarterly forecasts, to evaluate the effectiveness of our business strategies and to communicate with our board of directors concerning our financial performance.

	Three Months Ended April 30,		
	 2024	2023	
	(dollars ir	thousand	's)
Total revenue	\$ 51,327	\$	40,996
Gross profit	\$ 45,645	\$	35,074
Add: Stock-based compensation expense	407		338
Add: Employer taxes on employee stock transactions	70		10
Non-GAAP gross profit	\$ 46,122	\$	35,422
Gross margin	 88.9%		85.6%
Non-GAAP gross margin	89.9%		86.4%

# Non-GAAP Operating Loss and Non-GAAP Operating Margin

We define non-GAAP operating loss and non-GAAP operating margin as loss from operations and operating margin, respectively, excluding stock-based compensation expense, employer payroll taxes on employee stock transactions, restructuring charges and impairment of capitalized internal-use software. We use non-GAAP operating loss and non-GAAP operating margin in conjunction with GAAP measures to assess our performance, including in the preparation of our annual operating budget and quarterly forecasts, to evaluate the effectiveness of our business strategies and to communicate with our board of directors concerning our financial performance.

	Three Months	Ended A	pril 30,
	2024		2023
	(dollars in	thousand	ds)
Total revenue	\$ 51,327	\$	40,996
Loss from operations	\$ (22,540)	\$	(22,533)
Add: Stock-based compensation expense	14,627		9,275
Add: Employer taxes on employee stock transactions	1,216		267
Add: Restructuring <sup>(1)</sup>	 _		46
Non-GAAP operating loss	\$ (6,697)	\$	(12,945)
Operating margin	 (44)%		(55)%
Non-GAAP operating margin	(13)%		(32)%

For the three months ended April 30, 2023, an immaterial amount of stock-based compensation expense related to restructuring charges was included in the restructuring expense line.

### Non-GAAP Net Loss and Non-GAAP Net Loss Per Share

We define non-GAAP net loss attributable to common stockholders as net loss attributable to common stockholders, excluding stock-based compensation expense, employer payroll taxes on employee stock transactions and restructuring charges. We use non-GAAP net loss attributable to common stockholders and non-GAAP net loss per share attributable to common stockholders in conjunction with GAAP measures to assess our performance, including in the preparation of our annual operating budget and quarterly forecasts, to evaluate the effectiveness of our business strategies and to communicate with our board of directors concerning our financial performance.

	Three Months	Ended A	April 30,	
	2024		2023	
	(in thousands, exc	ept per :	share data)	
Net loss	\$ (20,995)	\$	(21,875)	
Add: Stock-based compensation expense	14,627		9,275	
Add: Employer taxes on employee stock transactions	1,216		267	
Add: Restructuring <sup>(1)</sup>	_		46	
Non-GAAP net loss	\$ (5,152)	\$	(12,287)	
GAAP net loss per share	\$ (0.42)	\$	(0.48)	
Non-GAAP net loss per share	\$ (0.10)	\$	(0.27)	
Weighted average shares outstanding, basic and diluted	49,788		45,843	

(1) For the three months ended April 30, 2023, an immaterial amount of stock-based compensation expense related to restructuring charges was included in the restructuring expense line.

#### Free Cash Flow

We define free cash flow as cash used in operating activities less additions to property and equipment, which includes capitalized internal-use software costs. We believe free cash flow is a useful indicator of liquidity that provides our management, board of directors and investors with information about our future ability to generate or use cash to enhance the strength of our balance sheet and further invest in our business and pursue potential strategic initiatives.

	Three !	Three Months Ended April					
	2024		2023				
		( in thousand	ds)				
Net cash provided by (used in) operating activities	\$	1,559 \$	(7,181)				
Less: Additions to property and equipment		(995)	(1,288)				
Free cash flow	\$	564 \$	(8,469)				
Net cash provided by (used in) investing activities	\$	3,695 \$	10,314				
Net cash provided by financing activities	\$	5,089 \$	2,764				

#### **Components of Results of Operations**

#### Revenue

We derive revenue from sales of subscriptions and services. Our subscription revenue is primarily derived from: (1) term-based software licenses sold in conjunction with post-contract support ("PCS" or "Support") and (2) a consumption-based DBaaS offering. PCS bundled with software licenses includes internet, email and phone support, bug fixes and the right to receive unspecified software updates and upgrades released when and if available during the subscription term. The software license is presented as "License." PCS and DBaaS revenue are presented as "Support and other" in our condensed consolidated statements of operations. License revenue is recognized upon transfer when our customer has received access to our software. The PCS is recognized ratably over the term of the arrangement beginning on the date when access to the subscription is made available to the customer and represents a substantial majority of our revenue. The DBaaS revenue is recognized on a consumption basis. The non-cancelable term of our subscription arrangements typically ranges from one to three years but may be longer or shorter in limited circumstances.

Our services revenue is derived from our professional services for the implementation or configuration of our platform and training. Services revenue is recognized over time based on input measures for professional services and upon delivery for training.

We expect our revenue may vary from period to period based on, among other things, the timing and size of new subscriptions, the proportion of term license contracts that commence within the period, the rate of customer renewals and expansions, the length of sales cycles and timing, delivery of professional services and training and fluctuations in customer consumption of our DBaaS offering.

# Cost of Revenue

Cost of subscription revenue primarily consists of personnel-related costs associated with our customer support organization, including salaries, bonuses, benefits and stock-based compensation, expenses associated with software and subscription services dedicated for use by our customer support organization, third-party cloud infrastructure expenses, amortization of costs associated with capitalized internal-use software related to our DBaaS offering and allocated overhead. There is no cost of revenue associated with our license revenue. We expect our cost of subscription revenue to increase in absolute dollars as our subscription revenue increases and as we continue to amortize capitalized internal-use software costs related to our DBaaS offering.

Cost of services revenue primarily consists of personnel-related costs associated with our professional services and training organization, including salaries, bonuses, benefits and stock-based compensation, costs of contracted third-party partners for professional services, expenses associated with software and subscription services dedicated for use by our professional services and training organization, travel-related expenses and allocated overhead. We expect our cost of services revenue to fluctuate from period to period depending on the timing and delivery of professional services and training.

#### Gross Profit and Gross Margin

Our gross profit and gross margin have been and will continue to be affected by various factors, including the average sales price of our subscriptions and services, the mix of subscriptions and services we sell and the associated revenue, the mix of geographies into which we sell and transaction volume growth. We expect our gross profit and gross margin to fluctuate in the near term depending on the interplay of these factors, and to the extent the revenue from our DBaaS offering increases as a percentage of total revenue, we expect our gross margin will decline over time.

#### **Operating Expenses**

Our operating expenses consist of research and development, sales and marketing, general and administrative and restructuring expenses. Personnel-related costs are the most significant component of operating expenses and consist of salaries, bonuses, benefits, sales commissions and stock-based compensation expenses.

#### Research and Development

Research and development expenses consist primarily of personnel-related costs, expenses associated with software and subscription services dedicated for use by our research and development organization, depreciation and amortization of property and equipment and allocated overhead. We expect that our research and development expenses will increase in absolute dollars as we continue to invest in the features and functionalities of our platform. We expect research and development expenses to fluctuate as a percentage of revenue in the near term, but to decrease as a percentage of revenue over the long term as we achieve greater scale in our business.

### Sales and Marketing

Sales and marketing expenses consist primarily of personnel-related costs, expenses associated with software and subscription services dedicated for use by our sales and marketing organization, costs of general marketing and promotional activities, amortization of deferred commissions, fees for professional services related to sales and marketing, travel-related expenses and allocated overhead. We expect that our sales and marketing expenses will increase in absolute dollars as we continue to expand our sales and marketing efforts to attract new customers and deepen our engagement with existing customers. We expect sales and marketing expenses to fluctuate as a percentage of revenue in the near term as we continue to invest in growing the reach of our platform through our sales and marketing efforts, but to decrease as a percentage of revenue over the long term as we achieve greater scale in our business.

#### General and Administrative

General and administrative expenses consist primarily of personnel-related costs associated with our finance, legal, human resources and other administrative personnel. In addition, general and administrative expenses include non-personnel costs, such as fees for professional services such as external legal, accounting and other professional services, expenses associated with software and subscription services dedicated for use by our general and administrative organization, certain taxes other than income taxes and allocated overhead. We expect that our general and administrative expenses will increase in absolute dollars as we continue to invest in the growth of our business and operate as a publicly-traded company. We expect general and administrative expenses to fluctuate as a percentage of revenue in the near term, but to decrease as a percentage of revenue over the long term as we achieve greater scale in our business.

### Restructuring

Restructuring expenses primarily consist of efforts we have undertaken to improve operational efficiency. Restructuring activities include employee severance and related costs and stock-based compensation expense from modifications of vested awards granted to certain employees impacted by our restructuring plan, which was completed during fiscal year 2024.

## Interest Expense

Interest expense consists primarily of unused Credit Facility fees for the three months ended April 30, 2023.

# Other Income (Expense), Net

Other income (expense), net consists primarily of foreign currency gains and losses related to the impact of transactions denominated in a foreign currency and interest income.

# Provision (benefit) for Income Taxes

Provision for income taxes consists primarily of income taxes in certain foreign jurisdictions in which we conduct business. We recorded a full valuation allowance against our U.S. deferred tax assets as we have determined that it is not more likely than not that the deferred tax assets will be realized. The cash tax expenses are impacted by each jurisdiction's individual tax rates, laws on the timing of recognition of income and deductions and availability of NOLs and tax credits. Our effective tax rate could be adversely affected to the extent earnings are lower than anticipated in countries that have lower statutory rates and higher than anticipated in countries that have higher statutory rates.

# **Results of Operations**

The following table sets forth our condensed consolidated statements of operations for the periods indicated (in thousands):

	Three Mon	Three Months Ended Apr					
	2024		2023				
Revenue:							
License	\$ 6,85	59 \$	4,943				
Support and other	42,1	19	33,599				
Total subscription revenue	49,00	38	38,542				
Services		39	2,454				
Total revenue	51,32	27	40,996				
Cost of revenue:							
Subscription <sup>(1)</sup>	3,99	57	3,673				
Services <sup>(1)</sup>	1,72	25	2,249				
Total cost of revenue	5,68	32	5,922				
Gross profit	45,64	15	35,074				
Operating expenses:							
Research and development <sup>(1)</sup>	17,84	<b>!</b> 7	15,383				
Sales and marketing <sup>(1)</sup>	37,7:	55	32,553				
General and administrative <sup>(1)</sup>	12,58	33	9,625				
Restructuring <sup>(1)</sup>			46				
Total operating expenses	68,18	35	57,607				
Loss from operations	(22,54	0)	(22,533)				
Interest expense	-	_	(25)				
Other income, net	1,5.	31	1,433				
Loss before income taxes	(21,00	9)	(21,125)				
Provision (benefit) for income taxes	(1	.4)	750				
Net loss	\$ (20,99	95) \$	(21,875)				

Includes stock-based compensation expense as follows:

	Three Months Ended April 30,					
		2024		2023		
		(in tho	usands)			
Cost of revenue—subscription	\$	266	\$	193		
Cost of revenue—services		141		145		
Research and development		3,993		2,768		
Sales and marketing		5,223		3,241		
General and administrative		5,004		2,928		
Restructuring		_		1		
Total stock-based compensation expense	\$	14,627	\$	9,276		

The following table sets forth our condensed consolidated statements of operations data expressed as a percentage of revenue:

	• • •	
	Three Months Ende	d April 30,
	2024	2023
Revenue:		
License	13 %	12 %
Support and other	82	82
Total subscription revenue	96	94
Services	4	6
Total revenue	100	100
Cost of revenue:		
Subscription	8	9
Services	3	5
Total cost of revenue	11	14
Gross profit	89	86
Operating expenses:		
Research and development	35	38
Sales and marketing	74	79
General and administrative	25	23
Restructuring	<u> </u>	*
Total operating expenses	133	141
Loss from operations	(44)	(55)
Interest expense	_	*
Other income, net	3	3
Loss before income taxes	(41)	(52)
Provision (benefit) for income taxes	*	2
Net loss	(41)%	(53)%

 $\label{eq:continuous} \begin{tabular}{ll} * & Represents less than 1\% \\ Note: Certain figures may not sum due to rounding. \end{tabular}$ 

# Comparison of Three Months Ended April 30, 2024 and 2023

#### Revenue

	Three Months Ended April 30,						
	2024 20		2023 \$ Char		\$ Change	% Change	
				(dollars i	in thoi	usands)	
Revenue							
License	\$	6,859	\$	4,943	\$	1,916	39 %
Support and other		42,179		33,599		8,580	26 %
Total subscription revenue		49,038		38,542		10,496	27 %
Services		2,289		2,454		(165)	(7)%
Total revenue	\$	51,327	\$	40,996	\$	10,331	25 %

Subscription revenue increased by \$10.5 million, or 27%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. The increase in subscription revenue was primarily driven by growth in revenues from existing customers. Approximately 99% of the increase in revenue was attributable to growth from existing customers. The remaining increase was attributable to new customers as we increased our customer base from 679 customers as of April 30, 2023 to 807 customers as of April 30, 2024.

Services revenue decreased by \$0.2 million, or 7%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023 primarily due to a decrease in delivery of professional service hours.

# Cost of Revenue, Gross Profit and Gross Margin

	Three Months Ended April 30,						
		2024		2023	\$ Change		% Change
				(dollars in	in thousands)		
Cost of revenue:							
Subscription	\$	3,957	\$	3,673	\$	284	8 %
Services		1,725		2,249		(524)	(23)%
Total cost of revenue	\$	5,682	\$	5,922	\$	(240)	(4)%
Gross profit	\$	45,645	\$	35,074			
Gross margin		88.9 %		85.6 %			
Headcount (at period end)		63		67			

Cost of subscription revenue increased by \$0.3 million, or 8%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. This change was primarily due to an increase of \$0.6 million related to the computing infrastructure costs associated with Couchbase Capella, partially offset by a \$0.4 million decrease in the amortization of capitalized internal-use software due to certain costs that were fully amortized in the prior fiscal year.

Cost of services revenue decreased by \$0.5 million, or 23%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. This was primarily due to a decrease of \$0.4 million in personnel-related costs due to lower headcount.

Gross margin increased during the three months ended April 30, 2024 compared to the three months ended April 30, 2023 primarily due to changes in the mix of subscription and service revenue.

#### Research and Development

	Three Months Ended April 30,						
	2024			2023		S Change	% Change
	(dollars in the					ds)	
Research and development	\$	17,847	\$	15,383	\$	2,464	16 %
Percentage of revenue		35 %		38 %			
Headcount (at period end)		294		276			

Research and development increased by \$2.5 million, or 16%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. This change was primarily due to an increase of \$2.8 million in personnel-related costs driven by headcount growth and higher stock-based compensation related to our RSUs offset by other factors.

# Sales and Marketing

		Three Months					
	2024			2023		S Change	% Change
					thousan	ds)	
Sales and marketing	\$	37,755	\$	32,553	\$	5,202	16 %
Percentage of revenue		74 %		79 %			
Headcount (at period end)		358		338			

Sales and marketing increased by \$5.2 million, or 16%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. This increase was primarily due to an increase of \$4.9 million in personnel-related costs driven by headcount growth and higher stock-based compensation related to our RSUs.

# General and Administrative

		Three Months	Ended				
	2024		2023		\$	Change	% Change
	(dollars in th					ds)	
General and administrative	\$	12,583	\$	9,625	\$	2,958	31 %
Percentage of revenue		25 %		23 %			
Headcount (at period end)		81		65			

General and administrative increased by \$3.0 million, or 31%, during the three months ended April 30, 2024 compared to the three months ended April 30, 2023. This increase was primarily due to an increase of \$2.8 million in personnel-related costs driven by headcount growth and higher stock-based compensation related to our RSUs.

# Interest Expense

T	hree Months E	nded April 30,			
<u> </u>	2024	2023	\$ Change	% Change	_
		(dollars	in thousands)		
\$	_	\$ (25)	\$	25 (100)9	%

The change in interest expense during the three months ended April 30, 2024 and 2023 was not material.

#### Other Income (Expense), Net

Tl	Three Months Ended April 30,					
	2024		2023	\$ (	Change	% Change
			(dollars	in thousar	nds)	
\$	1,531	\$	1,433	\$	98	7 %

Other income (expense), net fluctuated by \$0.1 million during the three months ended April 30, 2024 compared to the three months ended April 30, 2023 primarily due to higher yields on securities, partially offset by foreign currency losses due to fluctuations in exchange rates.

# Provision (benefit) for Income Taxes

	Three Months Ended April 30,					
	 2024 2023		2023		S Change	% Change
		thousan	ds)			
Loss before income taxes	\$ (21,009)	\$	(21,125)	\$	116	(1)%
Provision (benefit) for income taxes	(14)		750	\$	(764)	(102)%
Effective tax rate	0.1 %	, )	(3.6)%	)		

The change in provision (benefit) for income taxes for three months ended April 30, 2024 compared with the same period last year was primarily driven by foreign excess tax benefits related to stock-based compensation expense.

# **Liquidity and Capital Resources**

We have financed our operations through subscription revenue from customers accessing our platform and services revenue, and in July 2021, we completed our IPO with net proceeds totaling \$214.9 million. We have incurred losses and generated negative cash flows from operations for the last several years, including fiscal 2023 and 2024 and the three months ended April 30, 2024. As of April 30, 2024, we had an accumulated deficit of \$511.7 million.

As of April 30, 2024, we had \$160.2 million in cash, cash equivalents and short-term investments. We maintain our cash and cash equivalents, restricted cash and short-term investments with high-quality financial institutions. For more information, see "Concentration of Credit Risk" in Note 2 of our notes to the condensed consolidated financial statements. We believe our existing cash, cash equivalents and short-term investments, our total available borrowing capacity under the Credit Facility with MUFG Bank, Ltd., which is described in Note 7 of our notes to the condensed consolidated financial statements, and cash provided by sales of subscriptions to our platform and sales of our services will be sufficient to meet our projected operating requirements and cash expenditures for at least the next 12 months. As a result of our revenue growth plans, both domestically and internationally, we expect that losses and negative cash flows from operations may continue in the future. Our future capital requirements will depend on many factors, including our subscription revenue growth rate, subscription renewals, billing timing and frequency, the timing and extent of spending to support development efforts, the expansion of sales and marketing activities, the introduction of new and enhanced platform features and functionality and the continued market adoption of our platform. We may in the future pursue acquisitions of businesses, technologies, assets and talent.

We may be required to seek additional equity or debt financing. In the event that additional financing is required from outside sources, we may not be able to raise it on terms acceptable to us or at all. If we are unable to raise additional capital or generate cash flows necessary to expand our operations and invest in new technologies, our competitive position could weaken, and our business, financial condition and results of operations could be adversely affected.

We typically invoice our subscription customers annually in advance. Therefore, a substantial source of our cash is from such prepayments, which are included on our condensed consolidated balance sheets as deferred revenue. Deferred revenue consists of billed fees for our subscriptions, prior to satisfying the criteria for revenue recognition, which are subsequently recognized as revenue in accordance with our revenue recognition policy. As of April 30, 2024, remaining performance obligations, including both deferred revenue and non-cancelable contracted amounts, were \$220.0 million. We expect to recognize revenue of \$137.0 million on these remaining performance obligations over the next 12 months, with the remaining balance recognized thereafter.

#### **Cash Flows**

The following table shows a summary of our cash flows for the periods presented:

	,	Three Months Ended April 30,			
		2024	2023		
		(in thousands)			
Net cash provided by (used in):					
Operating activities	\$	1,559 \$	(7,181)		
Investing activities	\$	3,695 \$	10,314		
Financing activities	\$	5,089 \$	2,764		

## **Operating Activities**

Cash provided by operating activities for the three months ended April 30, 2024 of \$1.6 million primarily consisted of our net loss of \$21.0 million, adjusted for non-cash charges of \$19.3 million and net cash inflows of \$3.2 million from changes in our operating assets and liabilities. The primary drivers of the changes in operating assets and liabilities include a \$10.2 million decrease in accounts receivable related to strong collections and timing of billings and a \$7.7 million increase in deferred revenue due to timing of billings. These changes were partially offset by a \$9.2 million decrease in accrued compensation primarily due to timing of bonus and commissions payments, a \$3.1 million increase in deferred commissions related to increased sales during the period, a \$0.8 million decrease in lease liabilities driven by monthly rental payments for operating leases, a \$0.8 million decrease in accrued expenses and other liabilities due to the timing of accruals and payments, and a \$0.8 million decrease in accounts payable due to timing of payments.

Cash used in operating activities for the three months ended April 30, 2023 of \$7.2 million primarily consisted of our net loss of \$21.9 million, adjusted for non-cash charges of \$14.6 million and net cash inflows of \$0.1 million from changes in our operating assets and liabilities. The primary drivers of the changes in operating assets and liabilities include a \$6.4 million increase in deferred revenue due to timing of billings, a \$5.5 million increase in accounts payable due to timing of payments, and a \$1.4 million decrease in prepaid expenses and other assets due to the timing of prepaid operating expenses. These changes were partially offset by a \$4.8 million increase in deferred commissions related to increased sales during the period, a \$4.1 million decrease in accrued compensation primarily due to the timing of bonus payments and purchases under our employee stock purchase plan, a \$2.3 million increase in accounts receivable due to timing of billings and collections, a \$1.3 million decrease in accrued expenses and other liabilities due to payments related to restructuring activities, and a \$0.8 million decrease in lease liabilities driven by monthly rental payments for operating leases.

## **Investing Activities**

Cash provided by investing activities for the three months ended April 30, 2024 of \$3.7 million consisted of maturities of short-term investments net of purchases of \$4.7 million and additions to property and equipment of \$1.0 million.

Cash provided by investing activities for the three months ended April 30, 2023 of \$10.3 million consisted of maturities of short-term investments net of purchases of \$11.6 million and additions to property and equipment of \$1.3 million.

### Financing Activities

Cash provided by financing activities for the three months ended April 30, 2024 of \$5.1 million was primarily due to \$3.3 million in proceeds from the issuance of common stock upon exercises of stock options, and \$1.8 million in proceeds from the issuance of common stock under our employee stock purchase plan.

Cash provided by financing activities for the three months ended April 30, 2023 of \$2.8 million was primarily due to \$1.9 million in proceeds from the issuance of common stock upon exercises of stock options, and \$0.8 million in proceeds from the issuance of common stock under our employee stock purchase plan.

#### **Contractual Obligations and Commitments**

Our contractual obligations consist of purchase obligations and operating lease commitments. Purchase obligations include agreements to purchase goods or services that are enforceable and legally binding on us and that specify all significant terms, including fixed or minimum quantities to be purchased; fixed, minimum or variable price provisions and the approximate timing of the transaction. These obligations relate to third-party cloud infrastructure agreements and subscription arrangements. Our operating lease commitments relate primarily to our office facilities.

For further information on our commitments and contingencies, refer to Note 9 in our unaudited condensed consolidated financial statements contained within this Quarterly Report on Form 10-Q. There has been no material change in our contractual obligations and commitments other than in the ordinary course of business since our fiscal year ended January 31, 2024. See our Annual Report on Form 10-K for the fiscal year ended January 31, 2024, which was filed with the SEC on March 26, 2024, for additional information regarding our contractual obligations.

### **Indemnification Agreements**

In the ordinary course of business, we enter into agreements of varying scope and terms pursuant to which we agree to indemnify customers, vendors, lessors and other business partners with respect to certain matters, including, but not limited to, losses arising out of the breach of such agreements, services to be provided by us or from intellectual property infringement claims made by third parties. Additionally, we entered into indemnification agreements with our directors and officers that require us, among other things, to indemnify them against certain liabilities that may arise by reason of their status or service as directors or officers. To date, no demands have been made upon us to provide indemnification under such agreements, and there are no claims that we are aware of that could reasonably be expected to have a material effect on our financial condition, results of operations or cash flows

### **Critical Accounting Policies and Estimates**

Our condensed consolidated financial statements and the related notes thereto included elsewhere in this Quarterly Report on Form 10-Q are prepared in accordance with U.S. generally accepted accounting principles. The preparation of condensed consolidated financial statements also requires us to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue, costs and expenses, and related disclosures. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances. Actual results could differ significantly from the estimates made by management. To the extent that there are differences between our estimates and actual results, our future financial statement presentation, financial condition, results of operations, and cash flows will be affected.

There have been no significant changes to our critical accounting policies and estimates as compared to those described in "Management's Discussion and Analysis of Financial Condition and Results of Operations" set forth in our Annual Report on Form 10-K for the fiscal year ended January 31, 2024, which was filed with the SEC on March 26, 2024.

## **Recent Accounting Pronouncements**

See Note 2 to our condensed consolidated financial statements included in Part I, Item 1 of this Quarterly Report on Form 10-Q for a discussion of recent accounting pronouncements.

### **JOBS Act Accounting Election**

We are an "emerging growth company," as defined in the JOBS Act. The JOBS Act provides that an "emerging growth company" can take advantage of an extended transition period for complying with new or revised accounting standards. This provision allows an "emerging growth company" to delay the adoption of some accounting standards until those standards would otherwise apply to private companies. We have elected to use the extended transition period under the JOBS Act until the earlier of the date we (i) are no longer an "emerging growth company" or (ii) affirmatively and irrevocably opt out of the extended transition period provided in the JOBS Act. As a result, our financial statements may not be comparable to companies that comply with new or revised accounting pronouncements as of public company effective dates.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to market risks in the ordinary course of our business. Market risk represents the risk of loss that may impact our financial position due to adverse changes in financial market prices and rates. Our market risk exposure is primarily the result of fluctuations in interest rates and foreign currency exchange rates.

#### Interest Rate Risk

Our cash, cash equivalents and short-term investments primarily consist of highly liquid investments in money market funds, U.S. government treasury securities, commercial paper, U.S. government agency securities, corporate debt securities, and asset-backed securities. As of April 30, 2024, we had cash and cash equivalents of \$52.0 million and short-term investments of \$108.2 million. We do not enter into investments for trading or speculative purposes. Our investments are exposed to market risk due to fluctuations in interest rates, which may affect our interest income and the fair value of our investments. However, due to the short-term nature of our investment portfolio, we do not believe an immediate 10% increase or decrease in interest rates would have a material effect on our results of operations and cash flows. We therefore do not expect our results of operations or cash flows to be materially affected by a sudden change in market interest rates.

## Foreign Currency Risk

The functional currency of our foreign subsidiaries is the U.S. Dollar. Accordingly, each foreign subsidiary remeasures monetary assets and liabilities at period-end exchange rates, while nonmonetary items are remeasured at historical rates. Revenue and expense items are remeasured at the exchange rates in effect on the day the transaction occurred, except for those expenses related to non-monetary assets and liabilities, which are remeasured at historical exchange rates. Remeasurement adjustments are recognized in other income (expense), net in our condensed consolidated statement of operations.

The volatility of exchange rates depends on many factors that we cannot forecast with reliable accuracy. We have experienced and will continue to experience fluctuations in foreign exchange gains (losses) related to changes in foreign currency exchange rates. In the event our foreign currency denominated assets, liabilities, revenue or expenses increase, our results of operations and cash flows may be more greatly affected by fluctuations in the exchange rates of the currencies in which we do business. We have not engaged in the hedging of foreign currency transactions to date, although we may choose to do so in the future.

As of April 30, 2024, a hypothetical 10% change in the relative value of the U.S. Dollar to other currencies would not have a material impact on our results of operations and cash flows.

## Item 4. Controls and Procedures.

### **Evaluation of Disclosure Controls and Procedures**

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures as of the end of the period covered by this Quarterly Report on Form 10-Q. The term "disclosure controls and procedures," as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934 (the "Exchange Act"), means controls and other procedures of a company that are designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the rules and forms of the Securities and Exchange Commission (the "SEC"). Disclosure controls and procedures include, without limitation, controls and procedures designed to provide reasonable assurance that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is accumulated and communicated to the company's management, including its principal executive and principal financial officers, or persons performing similar functions, as appropriate to allow timely decisions regarding required disclosure.

#### Table of Contents

Based on such evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, as of the end of the period covered by this Quarterly Report on Form 10-Q, our disclosure controls and procedures were effective at the reasonable assurance level.

#### **Changes in Internal Control Over Financial Reporting**

There were no changes in our internal control over financial reporting identified in connection with the evaluation required by paragraph (d) of Exchange Act Rules 13a-15 or 15d-15 that occurred during the period covered by this Quarterly Report on Form 10-Q that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### Inherent Limitations on Effectiveness of Controls and Procedures

Our management, including our Chief Executive Officer and Chief Financial Officer, believes that our disclosure controls and procedures and internal control over financial reporting are designed to provide reasonable assurance of achieving their objectives and are effective at the reasonable assurance level. However, management does not expect that our disclosure controls and procedures or our internal control over financial reporting will prevent or detect all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within the company have been detected. The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Over time, controls may become inadequate because of changes in conditions, or the degree of compliance with the policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

#### **Part II Other Information**

#### **Item 1. Legal Proceedings**

From time to time, we may be subject to legal proceedings and claims that arise in the ordinary course of business, as well as governmental and other regulatory investigations and proceedings. In addition, third parties may from time to time assert claims against us in the form of letters and other communications. We are not currently a party to any legal proceedings that, if determined adversely to us, would, in our opinion, have a material and adverse effect on our business, financial condition, results of operations or cash flows. Future litigation may be necessary to defend ourselves, our partners and our customers, to determine the scope, enforceability and validity of third-party intellectual property and proprietary rights or to establish our intellectual property and proprietary rights. The results of any current or future litigation cannot be predicted with certainty and there can be no assurances that favorable outcomes will be obtained, and regardless of the outcome, litigation can have an adverse impact on us because of defense and settlement costs, diversion of management attention and resources and other factors.

#### Item 1A. Risk Factors

You should carefully consider the risks and uncertainties described below, together with all of the other information in this Quarterly Report on Form 10-Q, including our consolidated financial statements and related notes and the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations." Our business, financial condition, results of operations or prospects could also be harmed by risks and uncertainties not currently known to us or that we currently do not believe are material. If any of the risks actually occur, our business, financial condition, results of operations and prospects could be adversely affected. In that event, the market price of our common stock could decline, and you could lose part or all of your investment. The last day of our fiscal year is January 31. Our fiscal years ended January 31, 2023 and 2024 are referred to herein as fiscal 2023 and fiscal 2024, respectively.

## Risks Related to Our Industry and Business

#### We have a history of net losses and may not achieve or maintain profitability in the future.

We have incurred net losses since our inception, and we expect to continue to incur net losses in the future. We incurred net losses of \$80.2 million, \$68.5 million, \$21.0 million and \$21.9 million for fiscal 2024, fiscal 2023 and the three months ended April 30, 2024 and 2023, respectively. As of April 30, 2024, we had an accumulated deficit of \$511.7 million. We intend to continue investing significant resources to further develop our platform, and expand our sales, marketing, operations and infrastructure, both domestically and internationally. Any failure to increase our revenue sufficiently at a rate that exceeds the rate of increase in our investments and other expenses could prevent us from achieving or maintaining profitability.

## We may not continue to grow on pace with historical rates.

Our historical revenue, revenue growth, key business metrics or key business metrics growth should not be considered indicative of our future performance. Our revenue was \$180.0 million, \$154.8 million, \$51.3 million and \$41.0 million for fiscal 2024, fiscal 2023 and the three months ended April 30, 2024 and 2023, respectively. Our revenue growth rate has fluctuated in prior periods and we expect our revenue growth rate to continue to fluctuate. Our revenue growth rate may be impacted by a number of factors, including slowing adoption of or demand for our products and services, increasing competition, decreasing growth of our overall market, changes to technology or our failure to capitalize on growth opportunities, among others.

## If we fail to manage our growth effectively, our brand, business, financial condition and results of operations could be adversely affected.

We have experienced strong growth in our employee headcount, geographic reach and operations, and we expect to continue to grow in the future. Managing our growth effectively and integrating new employees, technologies and acquisitions into our existing business will require us to continue expanding our operational and financial infrastructure while maintaining the beneficial aspects of our culture. Continued growth could challenge our ability to develop and improve our operational, financial and management controls, enhance our reporting systems and procedures, recruit, train and retain highly skilled personnel, maintain customer satisfaction and manage our costs and operating expenses. Further, as our customers adopt our products and services for an increasing number of use cases, we have had to support more complex commercial relationships. We must continue to improve and expand our information technology ("IT"), and financial infrastructure, operating and administrative systems and relationships with various partners and other third parties. In addition, we operate globally and have established numerous international subsidiaries. Plans to continue doing so will place additional demands on our resources and operations. If we do not manage the growth of our business and operations effectively, the quality of our products and services and the efficiency of our operations could suffer. This could impair our ability to attract new customers, retain existing customers and expand their use of our products and services, any of which could adversely affect our brand, business, financial condition and results of operations.

# We face intense competition and if we are unable to compete effectively, our business, financial condition and results of operations would be adversely affected.

The database software market in which we operate is competitive and characterized by rapid changes in technology, customer requirements and industry standards and frequent introductions of new products and services. Many established businesses aggressively compete against us and have offerings with functionalities similar to those of our products and services. These competing offerings may also be complementary with ours and customers often deploy our platform alongside a competitor's product.

We primarily compete with established legacy database providers, such as Oracle, IBM and Microsoft, providers of NoSQL database offerings, such as MongoDB, and cloud infrastructure providers with database functionalities, such as Amazon, Microsoft and Google. In the future, other large software and internet companies with substantial resources, customers and brand power may also seek to enter our market. We would expect competition to increase with the entry of new companies and the introduction of innovative technologies.

Many of our existing and potential competitors have or could have, substantial competitive advantages, including but not limited to:

- greater name recognition and longer operating histories;
- broader distribution and established relationships with partners, customers and application developer communities;
- greater financial resources for sales and marketing, acquisitions and entering into strategic partnerships;
- competitive pricing and greater professional services and customer support resources;
- lower labor, research and development costs; and
- more mature intellectual property rights portfolios.

If we fail to compete effectively with respect to any of these competitive advantages, we may weaken our competitive position, fail to attract new customers or lose or fail to renew existing customers, which could adversely affect our business, financial condition and results of operations.

We expect competition to increase with the entry of new companies and the introduction of innovative technologies. Potential customers may believe that substitute technologies or ancillary solutions that address narrower segments overall are adequate for their needs. Further, we have made portions of our source code available on an open source and source available basis and face risks if others compete effectively using our code. Additionally, some of our current or potential competitors have made or could make acquisitions of businesses or establish cooperative relationships, among themselves or with others, that may allow them to offer more directly competitive and comprehensive offerings than were previously offered and adapt more quickly to new technologies and customer needs, devote greater resources to the promotion or sale of their products, initiate or withstand substantial price competition, take advantage of other opportunities more readily or develop and expand their offerings, including those that may incorporate AI, more quickly than we do. If we are unable to anticipate or effectively react to competitive challenges, our business, financial condition and results of operations could be adversely affected.

We may fail to cost-effectively acquire new customers or obtain renewals, upgrades or expansions from our existing customers, which would adversely affect our business, financial condition and results of operations.

Attracting and retaining sales and marketing personnel, developing partner relationships and raising awareness of our platform, including leveraging Community Edition and free trials, are critical for customer acquisition, and failure to cost-effectively acquire new customers could negatively impact our growth.

Our success depends, in part, on retaining existing customers through subscription renewals and expanding relationships, including broadening customers' use cases within our products and adopting additional Couchbase products and services. The non-cancelable term of our subscriptions are typically one to three years but may be longer or shorter in limited circumstances, and renewals or upgrades are not guaranteed. In addition, renewals might not have a similar contract period or differ in price, terms or capacity, or customers may decide to otherwise downgrade their subscriptions. For example, the impact of the macroeconomic environment has caused, and may in the future continue to cause, certain customers to request concessions including extended payment terms or better pricing, increased customer churn, a lengthening of our sales cycles with prospective customers, a delay of planned projects or expansions and reduced contract values with certain prospective and existing customers. Retention may fluctuate due to factors including our customers' satisfaction with our products and services, our licensing models, the prices, features or perceived value of competing offerings, changes to our offerings or general economic conditions, among other things.

Additionally, our success depends, in part, on our determination of which product features to include in our free versus paid versions of our products (which we call the Community Edition and Enterprise Edition, respectively, for our Server and Mobile suite of products) including the timing of when to incorporate Enterprise Edition features into our Community Edition products. Any failure on our part to determine the correct balance and timing may adversely affect our business. Existing or potential customers may determine that the functionality of our free versions is sufficient for their needs and as a result may not convert from the use of our Community Edition or free trials to a paid product or downgrade from our paid products. Further, users of our Enterprise Edition Server and Mobile products may violate our license terms by using our product without paying for a required subscription or by exceeding their subscription entitlements, and we may not always be able to determine when this occurs or enforce our license terms.

In addition, expanding our customer base in new industry verticals depends on effective organization, focus and training of our sales and marketing personnel, efficient pricing and product strategies and educating the enterprise architects and application developers in such industries about the benefits and features of our products and services. Inadequate returns on sales and marketing efforts and investments may harm our business.

## The market for our products and services is relatively new and evolving, and our future success depends on the growth and expansion of this market.

It is uncertain whether the market for our products and services will continue to grow, how rapidly it will grow, or whether our products and services will be more widely adopted. Our success will depend, in part, on market acceptance and the widespread adoption of our products and services as an alternative to legacy or other offerings and the selection of our products and services over competing offerings that may have similar functionality. Technologies related to database offerings are still evolving and we cannot predict market acceptance of our products and services or the development of other competing offerings based on entirely new technologies. For example, we derive a substantial majority of our revenue from subscriptions for, and services related to Couchbase Server and Couchbase Mobile. Demand for our platform is affected by a number of factors, many of which are beyond our control, including continued market acceptance by existing customers and potential customers, the ability to expand the product for different use cases, the timing of development and releases of new offerings by our competitors, technological change and the growth or contraction in the market in which we compete. It is possible that customer adoption of our new products, such as Couchbase Capella, may replace a portion of customer spend on our existing products. If the market for database solutions, and for NoSQL database solutions in particular, does not continue to grow as expected, or if we are unable to continue to efficiently and effectively respond to the rapidly evolving trends and meet the demands of our customers, achieve more widespread market awareness and adoption of our products and services or otherwise manage the risks associated with the introduction of new products and services, our competitive position would weaken and our business, financial condition, results of operations and prospects would be adversely affected.

# If we fail to innovate in response to changing customer needs, new technologies or other market requirements, our business, financial condition and results of operations could be harmed.

Our ability to attract new customers and expand our relationship with our existing customers depends, in part, on our ability to continue to enhance and improve our products and services, introduce compelling new features, address additional use cases and develop features that reflect the constantly evolving nature of technology, regulations, and our customers' needs. The success of any new or enhanced product or service features depends on several factors, including our anticipation of market changes and market demand, timely completion and delivery, adequate quality testing, integration with existing technologies and applications and competitive pricing. If our investments in new products and services, including Couchbase Capella, are not successful, our business, financial condition and results of operations would be adversely affected.

In addition, because our products and services are designed to operate with a variety of systems, applications, data and devices, we will need to continuously modify and enhance our products and services to keep pace with changes in such systems. We may not be successful in developing these modifications and enhancements. The addition of new features and solutions to our products and services may increase our research and development expenses, compliance, personnel, security, infrastructure and other expenses. We have adopted new features and may introduce others in the future, which may not be introduced in a timely or cost-effective manner or may not achieve the market acceptance necessary to generate sufficient revenue to justify the related research and development and other related expenses. It is difficult to predict customer adoption of new features. Such uncertainty limits our ability to forecast our future results of operations and subjects us to a number of challenges, including our ability to plan for and model future growth. If we are unable to manage the risks associated with the development of new products and features, our business would be adversely affected. If new technologies emerge that enable others to deliver competitive products and services at lower prices, more efficiently, more conveniently or more securely, such technologies could adversely impact our ability to compete effectively.

Generally, our professional services and training revenues have grown as our subscription revenue has grown. We believe our investment in services facilitates the adoption of our products. However, Couchbase Capella contains a service component and Couchbase Capella customers may not require standalone services at the same rate as for our enterprise Server and Mobile offerings. As a result, as customers migrate to Couchbase Capella, it is not clear if revenue from our services business will continue to grow at rates consistent with prior periods. Further, professional services and training may be seen as ancillary to our core product offerings. To the extent customers reduce spending due to macroeconomic conditions, customers may opt out of services, which may harm our results of operations.

#### We have a limited operating history, which makes it difficult to predict our future results of operations.

We were formed in 2011 with the merger of Membase, Inc. and CouchOne, Inc. Since our formation, we have frequently expanded our product features and services and evolved our pricing methodologies. Our limited operating history and our evolving business make it difficult to evaluate our future prospects and the risks and challenges we may encounter. These risks and challenges include, among other things, our ability to manage our costs, accurately forecast revenue, gain new customers, retain or expand existing customers, introduce successful products, services and features and compete effectively.

If we fail to address the risks and challenges that we face, including those above as well as those described elsewhere in this "Risk Factors" section, our business, financial condition, results of operations, key business metrics and prospects could be adversely affected. Further, because we have limited historical financial data and operate in a rapidly evolving market, any predictions about our future revenue and expenses may not be as accurate as they would be if we had a longer operating history or operated in a more predictable market. We have encountered in the past, and will encounter in the future, risks and uncertainties frequently experienced by growing companies with limited operating histories in rapidly changing industries. If our assumptions regarding these risks and uncertainties, which we use to plan and operate our business, are incorrect or change, or if we do not address these risks successfully, our results of operations could differ materially from our expectations and our business, financial condition and results of operations could be adversely affected.

Our future results of operations and key business metrics may fluctuate significantly, and if we fail to meet the expectations of analysts or investors, the market price of our common stock and the value of your investment could decline substantially.

Our results of operations and key business metrics may fluctuate from period to period as a result of a number of factors, many of which are outside of our control and may be difficult to predict, including:

- market acceptance and demand for our products and services, including new products and services;
- the quality and level of our execution of our business strategy and operating plan;
- the effectiveness of our sales and marketing programs;
- the length of our sales cycle, including the timing of renewals;
- our ability to attract new customers, particularly large enterprises;
- our ability to retain customers and expand their adoption of our products and services, particularly our largest customers;
- our ability to successfully expand internationally and penetrate key markets;
- a failure to maintain the level of service uptime and performance required by our customers with certain of our products;
- technological changes and the timing and success of new or enhanced product features by us or our competitors or any other change in the competitive landscape of our market;
- our product mix and the revenue recognition related to such products;
- · changes in the average contract term or the timing of revenue recognition, any of which may impact implied growth rates;
- · changes to our packaging and licensing models, which may impact the timing and amount of revenue recognized;
- increases in and the timing of operating expenses that we may incur to grow our operations and to remain competitive;
- pricing pressure as a result of competition or otherwise;
- seasonal buying patterns;
- the implementation of cost-saving activities, extra layers of scrutiny and approval, and customers electing to buy in smaller increments as a result of macroeconomic conditions;
- the impact and costs related to the acquisition and integration of businesses, talent, technologies or intellectual property rights;
- an inability to enforce our licenses associated with our products:

- our ability to successfully hire and retain employees and key members of our management team;
- changes in the legislative, litigation or regulatory environment; and
- general economic conditions in either domestic or international markets, including geopolitical uncertainty and instability, as well as the
  effects of foreign exchange fluctuations.

Any one or more of the factors above may result in significant fluctuations in our results of operations. We also intend to continue to invest significantly to grow our business in the near future. In addition, we generally experience seasonality based on when we enter into agreements with customers, and our quarterly results of operations generally fluctuate from quarter to quarter depending on customer buying habits. This seasonality is reflected to a lesser extent, and sometimes is not immediately apparent, in revenue, due to the fact that a substantial portion of our subscription revenue is recognized ratably over the term of the subscription, which typically ranges from one to three years. We expect that seasonality will continue to affect our results of operations in the future. The variability of our results of operations or other operating estimates could result in our failure to meet our expectations or those of securities analysts or investors; if so, the market price of our common stock could decline, and we could face costly lawsuits, including securities class action suits.

We recognize a significant portion of revenue from subscriptions over the term of the relevant subscription period, and as a result, downturns or upturns in sales are not immediately reflected in full in our results of operations.

Subscription revenue accounts for a significant portion of our revenue, comprising 96% and 94% of total revenue for the three months ended April 30, 2024 and 2023, respectively.

Sales of new or renewal subscriptions may fluctuate as a result of a number of factors, including customers satisfaction, pricing, the prices of competitors' products and reductions in our customers' spending levels or fluctuations in customer usage of consumption-based offerings. If sales decline or if consumption-based customers consume Couchbase Capella at a slower rate than expected, our total revenue and revenue growth rate may decline.

Under most of our contracts, we recognize a portion of subscription revenue upon transfer of the software license to the customer and the larger remainder of the transaction price ratably over the term of the arrangement. See Note 2 to our consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q for more information. As we significantly rely on subscription revenue, a significant portion of the revenue that we report in each period will be derived from the recognition of deferred revenue relating to agreements entered into in prior periods. Consequently, a decline in new sales or renewals in any one period and any downturn in sales, demand or market acceptance for our products may not be immediately reflected in our results of operations for such period but in future periods. Our subscription-based products also make it difficult to rapidly increase our revenue through additional sales in any period, as a significant portion of such revenue from customers will be recognized over the term of the applicable agreement.

Further, we intend to increase our investment in research and development, sales and marketing and general and administrative functions and other areas to grow our business. These costs are generally expensed as incurred in contrast to our revenue. Accordingly, we may recognize the costs associated with such increased investments earlier than some of the anticipated benefits and the return on these investments may be lower, or may develop more slowly, than we expect, which could adversely affect our financial condition and results of operations.

We depend on our sales force, and we may fail to attract, retain, motivate or train our sales force, which could adversely affect our business, financial condition and results of operations.

Our sales force obtains new customers and drives additional sales to existing customers. Our hiring, training and retention efforts have been, and may further be, hindered as a result of the intense competition for talent. New hires require significant training and may take significant time before they achieve full productivity, and our remote and online onboarding and training processes may be less effective or efficient than in-person training and take longer. Further, hiring sales personnel in new countries requires additional set up and upfront costs that we may not recover if the sales personnel fail to achieve full productivity. If we are unable to attract, retain, motivate and train sufficient numbers of effective sales personnel, our sales personnel do not reach significant levels of productivity in a timely manner or our sales personnel are not successful in bringing potential customers into the pipeline, converting them into new customers or increasing sales to our existing customer base, our business, financial condition and results of operations would be adversely affected.

Our sales strategy to target larger enterprises involves risks that may not be present or that are present to a lesser extent with respect to smaller enterprises, such as long and unpredictable sales cycles and sales efforts that require considerable time and expense, particularly in the current macroeconomic environment.

Sales to large customers involve risks that may not be present or that are present to a lesser extent with sales to smaller customers, such as longer and unpredictable sales cycles, more complex customer requirements and processes,

substantial upfront sales costs and less predictability in completing some of our sales. These risks may be enhanced in the current macroeconomic environment. A number of factors influence the length and variability of our sales cycles, including the need to educate potential customers about the uses and benefits of our products and services, the discretionary nature of purchasing and budget cycles and the competitive nature of evaluation and purchasing approval processes and the size of the customer. For example, large customers often require proof of concept deployments, free trials or begin to deploy our products on a limited basis but nevertheless negotiate pricing discounts, which all increase our upfront investment in the sales effort with no guarantee that sales to these customers will justify our substantial upfront investment. Large customer sales have, in some cases, occurred in periods subsequent to those we anticipated, or have not occurred at all, the result of which could affect our cash flows and results of operations for that fiscal period and for future periods.

## If we are not able to maintain and enhance our brand, especially among enterprise architects, application developers and other key functions that support them, our business and results of operations may be adversely affected.

We believe that maintaining and enhancing our brand and our reputation as a leader in the market for database solutions is critical to our relationship with our existing customers and partners and our ability to attract new customers and partners. The successful promotion of our brand will depend on a number of factors, including our marketing efforts, our ability to foster awareness among enterprise architects, application developers and other key functions that support them, our ability to continue to develop high-quality products and services, our ability to successfully differentiate our products and services from those of our competitors, our ability to maintain the reputation of our products and services for data security and our ability to obtain, maintain, protect, defend and enforce our intellectual property and proprietary rights. Our brand promotion activities may not be successful or yield increased revenue. In addition, independent industry analysts often provide reports of our products and services, as well as the offerings of our competitors, and perception of our products and services in the marketplace may be significantly influenced by these reports. Negative reports, or reports that are less positive as compared to those of our competitors, may adversely affect our reputation and brand. Additionally, the performance of our partners may affect our reputation and brand if customers do not have a positive experience with our partners.

Our registered or unregistered trademarks or trade names may be challenged, infringed, circumvented or declared generic or determined to be infringing on other marks and our competitors may adopt trade names or trademarks similar to ours leading to market confusion. If we are otherwise unable to establish name recognition based on our trademarks and trade names, then we may not be able to compete effectively and our business may be adversely affected. The maintenance and promotion of our brand requires us to make substantial expenditures, yet may not generate customer awareness or yield increased revenue, and even if they do, any increase in revenue from such brand promotion initiatives may not offset the increased expenses we incur. If so, we may have reduced pricing power relative to our competitors, which could materially and adversely affect our business, financial condition and results of operations.

# Real or perceived errors, failures or bugs in our products or interruptions or performance problems associated with our technology and infrastructure could adversely affect our growth prospects, business, financial condition and results of operations.

Our products are complex, and therefore, undetected errors, failures or bugs have occurred in the past and may occur in the future. Our products are used in IT environments with different operating systems, system management software, applications, devices, databases, servers, storage, middleware, custom and third-party applications and equipment and networking configurations. This diversity increases the likelihood of errors or failures in the IT environments into which our products are deployed. Additionally, we rely upon third-party cloud hosting infrastructure providers to host our cloud offering. Despite testing by us, real or perceived errors, failures or bugs in our customer solutions, software or technology or the technology or software we license from third parties, including open source software, may not be found until our customers use our products. This could result in negative publicity, security related incidents such as data breaches, data loss, unavailability or corruption, loss of or delay in market acceptance of our products, harm to our brand, weakening of our competitive position or complaints or claims by customers for losses sustained by them or our failure to meet the stated service level commitments in our customer agreements. In such an event, we may be required, or may choose, for customer relations or other reasons, to expend significant additional resources in order to help correct the problem. Any errors, failures or bugs in our products could impair our ability to attract new customers, retain existing customers or expand their use of our products, any of which could adversely affect our business, financial condition and results of operations.

For certain of our products, our success depends, in part, on the ability of our existing customers and potential customers to access such products at any time and within an acceptable amount of time. We may experience service disruptions, outages, capacity constraints and other performance problems due to a variety of factors, including infrastructure changes or failures, human or software errors, malicious acts, terrorism, denial of service attacks or other

security related incidents or capacity constraints. In some instances, we may not be able to identify or remedy the cause or causes of these performance problems within an acceptable period of time. It may become increasingly difficult to maintain and improve our performance, especially during peak usage times and as our products and customer implementations become more complex. If our products are unavailable or if our customers are unable to access our products within a reasonable amount of time or at all, or if other performance problems occur, we may experience a loss of customers, lost or delayed market acceptance of our platform and services, delays in payment to us by customers, injury to our reputation and brand, legal claims against us and the diversion of our resources. The foregoing risks associated with any outage or service disruptions are magnified by the fact that our platform is typically used by our customers to support mission-critical applications. In addition, to the extent that we do not effectively address capacity constraints, upgrade our systems as needed and continually develop our technology and network architecture to accommodate actual and anticipated changes in technology, our business, financial condition and results of operations could be adversely affected.

Some of our customer contracts contain service level commitments, which contain specifications regarding response times for support, performance of our products and availability of our services. Any failure of or disruption to our infrastructure could impact the performance of our products and the availability of services to customers. If we are unable to meet our stated service level commitments or if we suffer extended periods of poor performance or unavailability of our products or services, we may be contractually obligated to provide affected customers with service credits or potentially face contract termination with refunds of prepaid amounts related to unused subscriptions. If we suffer performance issues or downtime that exceeds the service level commitments under our contracts with our customers, our business, financial condition and results of operations could be adversely affected.

Our ability to maintain and increase sales with our existing customers depends, in part, on the quality of our customer support, and our failure to offer high-quality support would harm our reputation and adversely affect our business and results of operations.

Our ability to provide effective support is vital to our business as our products are often utilized by our customers for mission-critical applications and are often integrated with and dependent on other core technologies, which factors also increase the complexity and challenge of providing support. If we do not succeed in helping our customers quickly resolve issues or provide effective ongoing education related to our products, our reputation could be harmed, and our existing customers may not renew or expand their use of our products. To the extent that we are unsuccessful in hiring, training and retaining adequate customer support personnel, our ability to provide adequate and timely support to our customers and our customers' satisfaction with our products, would be adversely affected. Our failure to provide and maintain high-quality customer support would harm our reputation and brand and adversely affect our business, financial condition and results of operations.

Our international operations and planned continued international expansion subject us to additional costs and risks, which could adversely affect our business, financial condition and results of operations.

Our continued success and our growth strategy depend, in part, on our planned continued international expansion. We are continuing to adapt to and develop strategies to address international markets, but such efforts may not be successful and are subject to a number of risks, including, without limitation:

- greater difficulty in enforcing contracts and managing collections in countries where our recourse may be more limited, as well as longer collection periods;
- higher costs of doing business internationally, including costs incurred in establishing and maintaining office space and equipment for our international operations;
- differing labor regulations, especially in the European Union ("EU") where labor laws may be and often are more favorable to employees;
- challenges inherent to efficiently recruiting and retaining talented and capable employees in foreign countries and maintaining our company culture and employee programs across all of our offices;
- fluctuations in exchange rates between the U.S. Dollar and foreign currencies in markets where we do business;
- management communication and integration problems resulting from language and cultural differences and geographic dispersion;
- costs associated with language localization of our products and services;
- risks associated with trade restrictions and foreign legal requirements, including any importation, certification and localization of our products and services that may be required in foreign countries;

- greater risk of unexpected changes in regulatory requirements, tariffs and tax laws, trade laws, export quotas, customs duties, treaties and other trade restrictions;
- costs of compliance with foreign laws and regulations and the risks and costs of non-compliance with such laws and regulations, including, but not limited to laws and regulations governing our corporate governance, product licenses, data privacy, data protection and data security regulations, particularly in the EU:
- compliance with anti-bribery laws, including, without limitation, the U.S. Foreign Corrupt Practices Act of 1977, as amended (the "FCPA"), the U.S. Travel Act and the United Kingdom Bribery Act 2010, violations of which could lead to significant fines, penalties and collateral consequences for us;
- risks relating to the implementation of exchange controls, including restrictions promulgated by the Office of Foreign Assets Control ( "OFAC") and other similar trade protection regulations and measures;
- heightened risk of unfair or corrupt business practices in certain geographies and of improper or fraudulent sales arrangements that may impact our financial condition and result in restatements of, or irregularities in, financial statements;
- the uncertainty of protection for intellectual property rights in some countries, particularly, those countries where we operate through a professional employer organization and do not have a direct contractual relationship with our service providers in such countries;
- exposure to regional or global public health issues and restrictions on travel or other measures undertaken by governments in response;
- general economic and political conditions in these foreign markets, including inflation concerns, rising interest rates, political and economic instability in some countries, such as the Hamas-Israel or Russia-Ukraine conflicts, and any related political or economic response and counter responses or otherwise by various global actors or general effect on the global economy;
- foreign exchange controls or tax regulations that might prevent us from repatriating cash earned outside the U.S.; and
- double taxation of our international earnings and potentially adverse tax consequences due to changes in the tax laws of the U.S. or the foreign jurisdictions in which we operate.

If we are unable to address these or other problems encountered in connection with our international operations and expansion, our operations may be negatively impacted. Some of our business partners also have international operations and are subject to the risks described above. These and other factors could harm our ability to generate revenue outside of the U.S. and, consequently, adversely affect our business.

In addition, compliance with evolving foreign regulations may increase operational costs. Failure to comply with these laws and regulations could have adverse effects on our business. In many foreign countries it is common for others to engage in business practices that are prohibited by our internal policies and procedures or U.S. regulations applicable to us. Although we have implemented policies and procedures designed to ensure compliance with these laws and policies, there can be no assurance that all of our employees, contractors, partners and third-party service providers will comply with these laws and policies. Violations of laws or key control policies by our employees, contractors, partners or third-party service providers could result in delays in revenue recognition, financial reporting misstatements, fines, penalties or the prohibition of the importation or exportation of our products and services and could have an adverse effect on our business, financial condition and results of operations.

We track certain key business metrics with internal systems and tools and do not independently verify such metrics. Certain of these metrics are subject to inherent challenges in measurement, and any real or perceived inaccuracies in such metrics may adversely affect our business and reputation.

We track certain metrics, including ARR, dollar-based net retention rate and number of customers, with internal systems and tools that are not independently verified by any third party and which may differ from estimates or similar metrics published by third parties due to differences in sources, methodologies or the assumptions on which we rely. Our internal systems and tools have a number of limitations, and our methodologies for tracking these metrics may change over time, which could result in unexpected changes to our metrics, including the metrics we publicly disclose. In addition, our ARR and dollar-based net retention rate calculations assume our customers will renew unless we receive notification of non-renewal and are no longer in negotiations prior to a measurement date, and will not increase or reduce, their subscriptions for our platform and services. If these assumptions are incorrect, our actual ARR and dollar-based net retention rate may differ significantly from the metrics presented in this Quarterly Report on Form 10-Q. If the internal systems and tools we use to track these metrics undercount or overcount performance or contain algorithmic or other technical errors, the data we report may not be accurate. While these numbers are based on what we believe to be reasonable estimates of our metrics for the applicable period of measurement, there are inherent challenges in measuring these metrics. Limitations or errors with respect to the data or how we measure data may affect our understanding of certain details of our business, which could affect our long-term strategies. If our key business metrics are not accurate representations of our business, if investors do not perceive our key business metrics to be accurate or if we discover material inaccuracies with respect to these figures, we expect that our business, reputation, financial condition and results of operations would be adversely affected.

We depend on our management team and other highly skilled personnel, and we may fail to attract, retain, motivate or integrate highly skilled personnel, which could adversely affect our business, financial condition and results of operations.

We depend on the continued contributions of our management team, key employees and other highly skilled personnel. Most of our management team and key employees are at-will employees, which means they may terminate their relationship with us at any time. We are also substantially dependent on the continued service of our existing engineering personnel because of the complexity of our products. The competition for top management, engineering talent and other highly skilled personnel is high, and the loss of their services or delays in hiring required personnel, particularly within our research and development and engineering organizations, could adversely affect our business, financial condition and results of operations.

Our future success also depends, in part, on continuing to attract and retain highly skilled personnel. Competition for these personnel in the San Francisco Bay Area, where our headquarters are located, and in other locations, is intense, and our industry faces significant competition for skilled personnel.

Additionally, the former employers of our new employees have, and in the future may assert that our new employees or we have breached legal obligations, which may be time-consuming, distracting to management and may divert our resources. Current and potential personnel also often consider the value of equity awards they receive in connection with their employment, and if the perceived value of our equity awards declines relative to those of our competitors, our ability to attract and retain highly skilled personnel may be harmed. If we fail to attract and integrate new personnel or retain and motivate our current personnel, our business, financial condition and results of operations could be adversely affected.

## Our company culture has contributed to our success and if we cannot maintain this culture as we grow, our business could be harmed.

We believe that our company culture, which promotes being valued and creating value, has been critical to our success. We face a number of challenges that may affect our ability to sustain our corporate culture, including:

- ability to identify, attract, reward and retain people in leadership positions in our organization who share and further our culture, values and mission;
- the increasing size and geographic diversity of our workforce;
- the continued challenges of a rapidly-evolving industry; and
- the integration of new personnel and businesses from acquisitions.

If we are not able to maintain our culture, our business could be adversely affected.

#### We may require additional capital, which may not be available on terms acceptable to us, or at all.

Historically, we have funded our operations and capital expenditures primarily through equity issuances, debt instruments and cash generated from our operations. To support our growing business, we must have sufficient capital to continue to make significant investments in our products and services. If we raise additional funds through the issuance of equity, equity-linked or debt securities, those securities may have rights, preferences or privileges senior to those of our common stock, and our existing stockholders may experience dilution. Any debt financing secured by us in the future could involve restrictive covenants relating to our capital-raising activities and other financial and operational matters, which may make it more difficult for us to obtain additional capital and to pursue business opportunities.

We evaluate financing opportunities from time to time, and our ability to obtain financing will depend on, among other things, our development efforts, business plans and operating performance and the condition of the capital markets at the time we seek financing. We cannot be certain that additional financing will be available to us on favorable terms, or at all. If we are unable to obtain adequate financing or financing on terms satisfactory to us, when we require it, our ability to grow and to respond to business challenges could be significantly limited and our business could be adversely affected.

## Future debt obligations could materially and adversely affect our business, financial condition or results of operations.

We maintain a revolving line of credit (the "Credit Facility") with MUFG Bank Ltd. ("MUFG"). Our ability to pay interest and repay the principal for any indebtedness, and maintain compliance with covenants as part of our agreement with MUFG, is dependent upon our ability to manage our business operations, generate sufficient cash flows to service such debt and the other factors discussed in this "Risk Factors" section.

In the event we draw on the Credit Facility or otherwise incur indebtedness, our debt obligations could adversely impact us. For example, these obligations could:

- require us to use a large portion of our cash flow to pay principal and interest on debt, which will reduce the amount of cash flow available to fund working capital, capital expenditures, acquisitions, research and development expenditures and other business activities;
- limit our future ability to raise funds for capital expenditures, strategic acquisitions or business opportunities, research and development and other general corporate requirements;
- restrict our ability to incur additional indebtedness and to create or incur certain liens;
- increase our vulnerability to adverse economic and industry conditions; and
- increase our exposure to interest rate risk from variable rate indebtedness.

There can be no assurance that we will be able to manage any of these risks successfully.

We may also need to refinance a portion of any of our outstanding indebtedness as it matures. There is a risk that we may not be able to refinance existing debt, including the Credit Facility, or that the terms may not be as favorable as the terms of our existing indebtedness. Furthermore, if prevailing interest rates or other factors at the time of refinancing result in higher interest rates upon refinancing, then the interest expense relating to that refinanced indebtedness would increase.

We may be unable to make acquisitions and investments or successfully integrate acquired companies and assets into our business, and our acquisitions and investments may not meet our expectations, any of which could adversely affect our business, financial condition and results of operations.

We may in the future acquire or invest in businesses, offerings, technologies or talent that we believe could complement or expand our products and services, enhance our technical capabilities or otherwise offer growth opportunities. However, we may not be able to fully realize the anticipated benefits of such acquisitions or investments due to inherent risks, including, without limitation:

- unanticipated costs or liabilities associated with the acquisition, including claims related to the acquired company, its offerings or technology,
  or potential violations of applicable law or industry rules and regulations, arising from prior or ongoing acts or omissions by the acquired
  businesses that are not discovered by due diligence during the acquisition process;
- incurrence of acquisition-related expenses, including those related to identifying, investigating and pursuing suitable acquisitions, whether or not they are consummated, which would be recognized as a current period expense;
- inability to generate sufficient revenue to offset acquisition or investment costs;

- inability to maintain relationships with customers and partners of the acquired business;
- challenges with incorporating acquired technology and rights into our products and services and maintaining quality and security standards
  consistent with our brand;
- inability to identify security vulnerabilities in acquired technology prior to integration with our technology and products and services;
- inability to achieve anticipated synergies or unanticipated difficulty with integration into our corporate culture;
- delays in customer purchases due to uncertainty related to any acquisition;
- the need to integrate or implement additional controls, procedures and policies;
- challenges caused by distance, language and cultural differences;
- harm to our existing business relationships with partners and customers as a result of the acquisition;
- potential loss of key employees;
- use of resources that are needed in other parts of our business and diversion of management and employee resources; and
- inability to recognize acquired deferred revenue in accordance with our revenue recognition policies.

Acquisitions also require the use of substantial portions of our available cash or the incurrence of debt to consummate the acquisition. Each of these could adversely affect our financial condition or the market price of our common stock. The sale of equity or issuance of equity-linked debt to finance any future acquisitions could result in dilution to our stockholders. The use of cash to finance any future acquisitions may limit other potential uses of our cash, including the retirement of outstanding indebtedness. The incurrence of indebtedness would result in increased fixed obligations and could also include covenants or other restrictions that would impede our ability to manage our operations. We may have to delay or forego a substantial acquisition if we cannot obtain the necessary financing to complete such acquisition in a timely manner or on favorable terms. Any of the foregoing could adversely affect our business, financial condition and results of operations.

## Use of AI, including in our products and services, combined with an uncertain regulatory environment, may result in reputational harm, liability, or other adverse consequences to our business.

We are developing and have launched new product features that use AI technologies, including, for example, Capella iQ, our generative AI-powered developer's coding assistant built into Couchbase Capella. AI technologies are complex and rapidly evolving, and we face significant competition from other companies as well as an evolving regulatory landscape. We may be unsuccessful in developing, integrating or maintaining product features using AI technologies that gain market traction, or in doing so in a cost-effective manner. The introduction of AI into our products, or the use of our products as part of AI technologies and applications, may result in new or enhanced governmental or regulatory scrutiny, litigation, intellectual property risks, confidentiality or security risks, ethical concerns or other complications that could harm our business, reputation or financial condition. New or modified legal and regulatory developments regulating AI such as the EU AI Act can impact the incorporation of AI technologies into our offerings and business, and any actual or perceived failure by us to comply with any such legal and regulatory developments or other actual or asserted obligations may lead to significant fines, penalties, regulatory investigations, lawsuits, significant costs for remediation, damage to our reputation or other liabilities.

Further, our use of AI tools and technologies may require additional investment and development of appropriate protections and safeguards for handling the use of our data, including customer data processed by us, with AI technologies in our product offerings or our tools. AI may create content that appears correct but is inaccurate or flawed, which may expose us to brand or reputational harm, competitive harm or legal liability if we, our customers or others rely on or use this flawed content to their detriment.

## Our business could be adversely affected by economic downturns.

Prolonged economic uncertainties or downturns could adversely affect our business, financial condition and results of operations and key business metrics. Negative conditions in the general economy in either the U.S. or abroad, including inflation and rising interest rate concerns, conditions resulting from financial and credit market fluctuations, changes in economic policy, trade uncertainty including changes in tariffs, sanctions, international treaties and other trade restrictions, the occurrence of a natural disaster, global public health crisis or armed conflicts, could continue to cause a decrease in corporate spending on IT offerings in general and negatively affect the growth of our business. We cannot

#### Table of Contents

predict the timing, strength or duration of any economic slowdown, instability or recovery, generally or within any particular industry or geography.

These conditions could make it extremely difficult for our customers and us to forecast and plan future business activities accurately and could cause our customers to reevaluate their decision to purchase our products and services, which could delay and lengthen our sales cycles or result in cancellations of planned purchases. In the current macroeconomic environment, we continue to see deal cycles that are consistent with previous quarters, along with an elevation in degree of budget scrutiny, slower than expected product migrations, lower than expected expansions, and customers electing to buy in smaller increments. Further, during challenging economic times, our customers may face issues in gaining timely access to sufficient credit, which could result in an impairment of their ability to make timely payments to us, if at all. If that were to occur, we may be required to increase our allowance for credit losses, which would adversely affect our results of operations.

A substantial downturn in any of the industries in which our customers operate may cause firms to react to worsening conditions by reducing their capital expenditures in general or by specifically reducing their spending on IT offerings. Customers in these industries may delay or cancel projects or seek to lower their costs by renegotiating vendor contracts. To the extent subscriptions to our products or expenditures on our services are perceived by existing customers or potential customers to be discretionary, our revenue may be disproportionately affected by delays or reductions in general IT spending.

### We face fluctuations in currency exchange rates, which could adversely affect our financial condition and results of operations.

To the extent we continue to expand internationally, we will become more exposed to fluctuations in currency exchange rates. The strengthening of the U.S. Dollar relative to foreign currencies increases the real cost of our products and services for our customers outside of the U.S. which could lead to the lengthening of our sales cycles or reduced demand for our products and services. Additionally, increased international sales may result in foreign currency denominated sales, increasing our foreign currency risk. Moreover, such continued expansion would increase operating expenses incurred outside the U.S. and denominated in foreign currencies. To date, we have not entered into any hedging transactions in an effort to reduce our exposure to foreign currency exchange risk. While we may decide to enter into hedging transactions in the future, the availability and effectiveness of these hedging transactions may be limited and we may not be able to successfully hedge our exposure, which could adversely affect our financial condition and results of operations.

## Our business could be adversely affected by pandemics, natural disasters, political crises or other unexpected events.

A significant natural disaster, such as an earthquake, fire, hurricane, tornado or flood, or a significant power outage or telecommunications failure, could disrupt our operations, mobile networks, the internet or the operations of our third-party service and technology providers. In particular, our corporate headquarters are located in the San Francisco Bay Area, a region known for seismic activity and wildfires. In addition, any unforeseen public health crises, political crises, such as terrorist attacks, war and other political instability or other catastrophic events, whether in the U.S. or abroad, including the Hamas-Israel and Russia-Ukraine conflicts and any related political or economic response and counter responses or otherwise by various global actors or general effect on the global economy, can continue to adversely affect our operations or the economy as a whole. The impact of any natural disaster, act of terrorism or other disruption to us or our third-party providers' abilities could result in decreased demand for our products and services or a delay in the provision of our products and services or could negatively impact consumer and business spending in the impacted regions or globally depending on the severity, any of which would adversely affect our business, financial condition and results of operations. All of the aforementioned risks would be further increased if our disaster recovery plans prove to be inadequate.

### Risks Related to Our Dependence on Third Parties

If we are unable to maintain successful relationships with our partners, our business, financial condition and results of operations could be harmed.

We employ a go-to-market business model whereby a portion of our revenue is generated by sales through or with our partners, including CSPs, independent software vendors, systems integrators, technology partners, original equipment manufacturers, marketplaces, managed service providers and resellers, that further expand the reach of our direct sales force into additional geographies, sectors, industries and channels. We have entered, and intend to continue to enter, into reseller relationships in certain international markets where we do not have a local presence. We provide certain partners with specific training and programs to assist them in selling our products and services, but our efforts to provide training and build relationships may be ineffective. In addition, if our partners are unsuccessful in marketing and selling our products and services, it would limit our planned expansion into certain geographies, sectors, industries and channels. If we are unable to develop and maintain effective sales incentive programs for our partners, we may not be able to successfully incentivize these partners to sell our products and services to customers.

Some of our partners may also market, sell and support offerings that are competitive with ours, may devote more resources to the marketing, sales and support of such competitive offerings, may have incentives to promote our competitors' offerings to the detriment of our own or may cease selling our products and services altogether. Our partners could also subject us to lawsuits, potential liability and reputational harm if, for example, any of our partners misrepresents the functionality of our products and services to customers, violate laws or violate our or their corporate policies. Our ability to achieve revenue growth in the future will depend, in part, on our success in maintaining successful relationships with our partners, identifying additional partners and training our partners to independently sell our products and services. Any shortcomings of the foregoing by us or our partners could adversely affect our business, financial condition, results of operations and growth prospects.

We rely on third-party service providers for many aspects of our business, and any failure to maintain these relationships could harm our business.

Our success depends, in part, on our relationships with third-party service providers, including providers of cloud hosting infrastructure, customer relationship management systems, financial reporting systems, human resource management systems, credit card processing platforms, marketing automation systems, payroll processing systems and data centers, among others. In particular, cloud hosting infrastructure is becoming increasingly important as customers adopt Couchbase Capella. If any of these third parties experience difficulty meeting our requirements or standards, become unavailable due to extended outages or interruptions, temporarily or permanently cease operations, face financial distress or other business disruptions or increase their fees, or if our relationships with any of these providers deteriorate or if any of the agreements we have entered into with such third parties are terminated or not renewed without adequate transition arrangements, or if we are unsuccessful in managing or negotiating cost-effective relationships with them, we could suffer increased costs and delays in our ability to provide customers with our products and services, our ability to manage our finances could be interrupted, receipt of payments from customers may be delayed, our ability to generate and manage sales leads could be weakened or our business operations could be disrupted. Any of such disruptions may adversely affect our business, financial condition, results of operations or cash flows until we replace such providers or develop replacement technology or operations.

Certain estimates and information we refer to publicly are based on information from third-party sources and we do not independently verify the accuracy or completeness of the data contained in such sources or the methodologies for collecting such data, and any real or perceived inaccuracies in such estimates and information may harm our reputation and adversely affect our business.

Certain estimates and information we refer to publicly, including general expectations concerning our industry and the market in which we operate and market size, are based to some extent on information provided by third-party providers. This information involves a number of assumptions and limitations, and although we believe the information from such third-party sources is reliable, we have not independently verified the accuracy or completeness of the data contained in such third-party sources or the methodologies for collecting such data. If investors do not perceive such data or methodologies to be accurate, or if we discover limitations or material inaccuracies with respect to such data or methodologies, our reputation, financial condition and results of operations could be adversely affected.

### Risks Related to Our Open Source and Intellectual Property

Our use of third-party open source software in our solutions, the availability of core portions of our source code on an open source or source available basis and contributions to our open source projects could negatively affect our ability to sell our products and provide our services, subject us to possible litigation and allow third parties to access and use software and technology that we use in our business, all of which could adversely affect our business and results of operations.

Our products include software that is licensed to us by third parties under "open source" licenses. Use and distribution of open source software may entail greater risks than use of third-party commercial software, because open source projects may have vulnerabilities and architectural instabilities, and also because open source licensors generally provide their software on an "as-is" basis and do not provide support, warranties, indemnification or other contractual protections regarding infringement claims or the quality of the code even though our customers may insist on such protections in our contracts with them. We have historically elected to make core portions of our source code available on an open source basis to facilitate adoption as well as collaboration and participation from our application developer communities. However, we may not be successful in this strategy, and our move toward source-available licensing, as well as the continued availability of our source code, may enable others to compete more effectively against us. In addition, the public availability of the source code for such software may make it easier for others to compromise our products. We expect to continue to incorporate such open source software in our products and allow core portions of our source code to be available on an open source or source-available basis in the future.

Although most of our code is developed in-house, we also receive a limited amount of contributions from our open source developer communities. We require third parties who provide contributions to us to assign ownership of all intellectual property rights in their contributions to us, or provide us with a perpetual license to their works, and represent that their contributions are original works and that they are entitled to assign or license these rights to us. However, we cannot be sure that we can use all contributions without obtaining additional licenses from third parties, and may be subject to intellectual property infringement or misappropriation claims as a result of our use of these contributions.

Some open source licenses contain requirements that we make available source code for modifications or derivative works we create based upon the type of open source software we use, or grant other licenses to our intellectual property. We seek to ensure that our closed-source proprietary software is not combined with, and does not incorporate, open source software in ways that would require the release of the source code of our closed-source proprietary software to the public. However, we cannot ensure that our processes for controlling our use of open source software in our products will be effective. If we are held to have failed to comply with the terms of applicable licenses or our current policies and procedures or otherwise combine our closed-source proprietary software with open source software in a certain manner, we could, under certain open source licenses, be required to release the source code of our closed-source proprietary software to the public at no cost under the terms of applicable open source licenses. This would allow our competitors to create similar offerings with lower development effort and time and ultimately could result in a loss of our competitive advantages.

Additionally, the terms of many open source licenses have not been interpreted by U.S. or foreign courts, and these licenses could be construed in a way that could impose unanticipated conditions or restrictions on our ability to provide or distribute our products. From time to time, there have been claims challenging the ownership of open source software against companies that incorporate open source software into their products, and the licensors of such open source software provide no warranties or indemnities with respect to such claims. As a result, we and our customers could be subject to lawsuits by parties claiming ownership of what we believe to be open source software. If an author or other third party that distributes such open source software were to allege that we had not complied with applicable license conditions, or if we are required to defend a customer in the event of such a claim, we could be required to incur significant legal expenses defending against such allegations. If we are unable to successfully defend against such allegations, we could be subject to significant damages or other liability, including being enjoined from the sale of our products and services. We could also be required to seek licenses from third parties to continue providing our products on terms that are not economically feasible, re-engineer our products, discontinue or delay the provision of our products if re-engineering cannot be accomplished on a timely basis or make generally available, in source code form, our proprietary code. Any of the foregoing would adversely affect our business, financial condition and results of operations.

## Our distribution and licensing model could negatively affect our ability to monetize and protect our intellectual property rights.

Many of our products are available for free on the internet, including a substantial portion of our source code on open source or source available terms. Also, we may have limited or no direct visibility into who may be using our software or to what extent or purpose, so our ability to detect violations of our product licenses is extremely limited. If we

are unable to manage the risks related to our licensing and distribution model, our business could be adversely affected. Additionally, we have adopted BSL 1.1, a source-available license for certain of our publicly available source code. We believe BSL 1.1 enables us to fairly and transparently control commercialization of our source code, however such licensing strategy may not prevent misuse of our source code. Additionally, BSL 1.1 is not an open source license, which may negatively impact adoption of the source code, reduce our brand and product awareness and negatively impact our ability to compete.

Because of the rights accorded to third parties under open source licenses, there may be fewer technology barriers to entry in the markets in which we compete and it may be relatively easy for new and existing competitors, some of whom may have greater resources than we have, to compete with us.

Open source software license terms generally allow liberal modifications and distribution of the code. We have historically elected to make core portions of our source code available on an open source basis. The continued availability of our source code, notwithstanding our move toward source-available licensing, among other things, may enable others to develop new software products or services that are competitive to ours without the same degree of overhead and lead time required by us, particularly if customers do not value the differentiation of our proprietary components. It is possible for new and existing competitors, including those with greater resources than ours, to develop their own open source software or hybrid proprietary and open source software offerings. In addition, some competitors make open source software available for free download or use or may position competing open source software as a loss leader. We cannot guarantee that we will be able to compete successfully against current and future competitors or that competitive pressure or the availability of open source software will not result in price reductions, reduced revenue and gross margins and loss of market share, any one of which could adversely affect our business. Our use of open source software may also limit our ability to assert certain of our intellectual property and proprietary rights against third parties, including competitors, who access or use software or technology that we have contributed to such open source projects.

We could incur substantial costs in obtaining, maintaining, protecting, defending and enforcing our intellectual property rights and any failure to obtain, maintain, protect, defend or enforce our intellectual property rights could reduce the value of our software and brand.

Our success depends, in part, upon our ability to obtain, maintain, protect, defend and enforce our intellectual property rights, including our proprietary technology, know-how and our brand. We rely on a combination of patents, copyrights, trademarks, service marks, trade secret laws, other intellectual property laws, confidentiality procedures and contractual provisions in an effort to establish and protect our proprietary rights. However, the steps we take to obtain, maintain, protect, enforce and defend our intellectual property rights may be inadequate if, for example, we are unable to enforce our rights or if we do not detect unauthorized use of our intellectual property, and as a result our competitors might gain access to our proprietary technology and develop and commercialize similar or substantially identical products, services or technologies, and our business, financial condition, results of operations or prospects could be adversely affected. While we have patent applications pending in the U.S., there can be no assurance that they will result in issued patents. As of April 30, 2024, we owned five issued U.S. patents, ten U.S. non-provisional patent applications, three Patent Cooperation Treaty, or PCT, patent applications, and two foreign patent applications. Pending PCT patent applications are not eligible to become issued patents until, among other things, we file national stage patent applications within 30 months in the countries in which we seek patent protection. If we do not timely file such national stage patent applications, we may lose our priority date with respect to our PCT patent applications and any patent protection on the inventions disclosed in such applications.

Even if we continue to seek patent protection in the future, we may be unable to obtain or maintain patent protection for our technology. In addition, any patents issued or licensed to us in the future may not provide us with competitive advantages, or may be successfully challenged by third parties. In addition, defending our intellectual property rights might entail significant expenses. Any of our patents, trademarks or other intellectual property rights that we have or may obtain may be challenged or circumvented by others or invalidated or held unenforceable through administrative process, including re-examination, *inter partes* review, interference and derivation proceedings and equivalent proceedings in foreign jurisdictions or litigation in the U.S. or in foreign jurisdictions. Others may infringe on our patents, trademarks or other intellectual property rights, independently develop similar, substantially identical or superior offerings, duplicate any of our offerings or design around our patents or other intellectual property rights or use information that we regard as proprietary to create products and services that compete with ours. Further, legal standards relating to the validity, enforceability and scope of protection of intellectual property rights are uncertain. Various courts, including the U.S. Supreme Court, have rendered decisions that affect the scope of patentability of certain inventions or discoveries relating to software and business methods. These decisions state, among other things, that a patent claim that recites an abstract idea, natural phenomenon or law of nature are not themselves patentable. Precisely what constitutes a law of nature or abstract idea is uncertain, and it is possible that certain aspects of our technology could be considered abstract ideas. Accordingly,

#### Table of Contents

the evolving case law in the U.S. may adversely affect our ability to obtain patents and may facilitate third-party challenges to any future owned or licensed patents. As we expand our international activities, our exposure to unauthorized copying and use of our services and platform capabilities and proprietary information will likely increase. Intellectual property protection may not be available to us in every country in which our services are available, or the laws of or mechanisms for enforcement in some foreign countries may not be as protective of intellectual property rights as those in the U.S. Policing unauthorized use of our technologies, trade secrets and intellectual property may thus be difficult, expensive and time-consuming. Accordingly, despite our efforts, we may be unable to prevent third parties from infringing upon, misappropriating or otherwise violating our intellectual property rights.

In addition, we have made core portions of our own software available under open source or source-available licenses, and we include third-party open source software in our products. We have also occasionally contributed source code to open source projects. Because the source code for any software we distribute under open source or source-available licenses or contribute to open source projects is publicly available, our ability to protect our intellectual property rights with respect to such source code may be limited or lost entirely.

We protect and rely, in part, on trade secrets, proprietary know-how and other confidential information to maintain our competitive position. While we generally enter into confidentiality and invention assignment agreements with our employees and consultants and enter into confidentiality agreements with other third parties, including suppliers, vendors and the parties with whom we have strategic relationships and business alliances, the assignment of intellectual property rights may not be self-executing or may be breached, and we may be forced to bring claims against third parties, or defend claims that they may bring against us, to determine the ownership of what we regard as our intellectual property. Additionally, we cannot guarantee that we have entered into such agreements with each party that has or may have created or developed intellectual property on our behalf or had access to our proprietary information, know-how or trade secrets. We cannot guarantee that these agreements will be effective in controlling access to, distribution, use, misuse, misappropriation, reverse engineering or disclosure of our proprietary information, know-how and trade secrets. Further, these agreements may not prevent our competitors or partners from independently developing offerings that are substantially equivalent or superior to ours. We may not have adequate remedies for any breach of these agreements. Further, we have experienced and may in the future experience unauthorized access of our proprietary source code, confidential information and know-how. We have and may in the future initiate litigation regarding trade secret misappropriation, but enforcing a claim that a party illegally disclosed or misappropriated a trade secret or know-how is difficult, expensive and time-consuming, and the outcome is unpredictable. In addition, some courts in and outside the U.S. are less willing or unwilling to protect trade secrets and know-how.

We may be required to spend significant resources in order to monitor and protect our intellectual property rights, and some violations may be difficult or impossible to detect. Any litigation to protect and enforce our intellectual property rights could be costly, time-consuming and distracting to management, and could result in the impairment or loss of portions of our intellectual property. Our efforts to enforce our intellectual property rights may be met with defenses, counterclaims and countersuits attacking the validity and enforceability of our intellectual property rights that, if successful, could cost us valuable intellectual property rights. Our inability to protect our intellectual property and proprietary technology against unauthorized copying or use, as well as any costly litigation or diversion of our management's attention and resources, could impair the functionality of our products, delay introductions of enhancements to our products, result in our substituting inferior or more costly technologies into our products or harm our reputation and brand. In addition, we may be required to license additional technology from third parties to develop and market new product features, which may not be on commercially reasonable terms, or at all, and would adversely affect our ability to compete.

We have been and may in the future become subject to intellectual property disputes which may be costly to defend, subject us to significant liability, require us to pay significant damages and limit our ability to use certain technologies.

We have been and may in the future become subject to intellectual property disputes. Our success depends, in part, on our ability to develop and commercialize our products and services without infringing, misappropriating or otherwise violating the intellectual property rights of third parties. However, we may not be aware if our products are infringing, misappropriating or otherwise violating third-party intellectual property rights, and such third parties may bring claims alleging such infringement, misappropriation or violation. Further, we have faced and may in the future face claims from third parties claiming ownership of, or demanding release of, the software or derivative works that we have developed, including works using third-party open source software, which could include our proprietary source code, or otherwise seeking to enforce the terms of the applicable open source license. In addition, certain companies and rights holders seek to enforce and monetize patents or other intellectual property rights they own, have purchased or otherwise obtained. Many potential litigants, including some of our competitors and patent-holding companies, have the ability to

dedicate substantial resources to assert their intellectual property rights and to defend claims that may be brought against them.

Certain of our agreements with our customers and other third parties include indemnification provisions under which we agree to indemnify or otherwise be liable to them for losses suffered or incurred as a result of claims of infringement, misappropriation or other violation of intellectual property rights. Any claim of infringement by a third party, even those without merit, against us or for which we are required to provide indemnification, are time consuming, could cause us to incur substantial costs defending against the claim, could distract our management from our business and could require us to cease use of such intellectual property. Further, because of the substantial amount of discovery required in connection with intellectual property litigation, we risk compromising our confidential information during this type of litigation. We may be required to make substantial payments for legal fees, settlement fees, damages (including treble damages and attorneys' fees if we are found to have willfully infringed a party's rights), royalties or other fees in connection with a claimant securing a judgment against us and we may be subject to an injunction or other restrictions that cause us to cease selling or using products or services that incorporate the intellectual property rights that we allegedly infringe, misappropriate or violate, including subscriptions to our products. We may also agree to a settlement that prevents us from distributing our products or a portion thereof, any of which could adversely affect our business, financial condition and results of operations.

With respect to any intellectual property rights claim, we may have to seek out a license to continue operations found to be in violation of such rights, which may not be available on favorable or commercially reasonable terms and may significantly increase our operating expenses. Some licenses may be non-exclusive, and therefore our competitors may have access to the same technology licensed to us. If a third party does not offer us a license to its intellectual property on commercially reasonable terms, or at all, we may be required to develop alternative, non-infringing technology, which could require significant time (during which we would be unable to continue to offer our affected product features), effort and expense and may ultimately not be successful. Any of these events would adversely affect our business, financial condition and results of operations.

Even if the claims do not result in litigation or are resolved in our favor, these claims and the time and resources necessary to resolve them, could divert the resources of our management. Moreover, there could be public announcements of the results of hearings, motions or other interim proceedings or developments, and if securities analysts or investors perceive these results to be negative, it would have a substantial adverse effect on our business, results of operations or the market price of our common stock. We expect that the occurrence of infringement claims is likely to grow as the market for platform and services grows. Accordingly, our exposure to damages resulting from infringement claims could increase and this could further exhaust our financial and management resources.

## Risks Related to Our Legal and Regulatory Environment

Our business is subject to a wide range of laws and regulations, many of which are evolving, and failure to comply with such laws and regulations could harm our business, financial condition and results of operations.

Our business is subject to regulation by various federal, state, local and foreign governmental agencies, including for data privacy and cybersecurity laws and regulations, intellectual property, employment and labor laws, workplace safety, consumer protection laws, anti-bribery laws, import and export controls, immigration laws, federal securities laws and tax laws and regulations. Further, emerging tools and technologies we may utilize in our products and services, like AI, may become subject to regulation under new laws or new applications of existing laws. In certain foreign jurisdictions, these regulatory requirements may be more stringent than in the U.S. These laws and regulations impose added costs on our business, noncompliance with which could subject us to:

- investigations, enforcement actions, orders and sanctions;
- mandatory changes to our products and services;
- · disgorgement of profits, fines and damages;
- · civil and criminal penalties or injunctions;
- claims for damages by our customers or partners;
- termination of contracts;
- · loss of intellectual property rights; and
- temporary or permanent debarment from sales to heavily regulated organizations and governments.

#### Table of Contents

If any governmental sanctions are imposed, enforcement actions are taken, or if we do not prevail in any possible civil or criminal litigation, our business, financial condition and results of operations could be adversely affected. In addition, responding to any action will likely result in a significant diversion of management's attention and resources and an increase in professional fees.

In addition, we must comply with laws and regulations relating to the formation, administration and performance of contracts with customers in heavily regulated industries and the public sector, including U.S. federal, state and local governmental organizations when selling our product to them directly or through partners, which affect how we and our partners do business with such customers. Failure to comply with these requirements by either us or our partners could subject us to investigations, fines, suspension or debarment from future government or other contracting opportunities and other penalties, which would adversely affect our business, financial condition, results of operations and growth prospects.

If our security measures, or those of our service providers or customers, are breached or unauthorized parties otherwise obtain access to our or our customers' data or software, our products and services may be perceived as not being secure, customers may reduce or terminate their use of our products and services and we may face claims, litigation, regulatory investigations, significant liability and reputational damage.

We collect, use, store and transmit or otherwise process data as part of our business operations, including personal data in and across multiple jurisdictions. We also use third-party service providers to collect, use, store, transmit, maintain and otherwise process such information. Increasingly, threats from computer malware, ransomware, viruses, social engineering (including phishing attacks), denial of service, application programming interface attacks or other attacks, employee theft or misuse and general hacking have become more prevalent in our industry and our customers' industries. Any of these security incidents could result in unauthorized access or damage to, or disablement, encryption, use, disclosure, modification, destruction, loss or other processing of, our data or customer data (including personal data), software or systems or disrupt our ability to provide our products and services. Any actual or perceived security incident could interrupt our operations, harm our reputation and brand, result in significant remediation and cybersecurity protection costs (including deploying additional personnel and modifying or enhancing our protection technologies and investigating and remediating any information security vulnerabilities), result in lost revenue, lead to regulatory investigations and orders, litigation, disputes, indemnity obligations, damages for breach of contract, penalties for violations of law or regulation and other legal risks, increase our insurance premiums, result in any other financial exposure, lead to loss of customer confidence in us or decreased use of our products and services and otherwise adversely affect our reputation, competitiveness, business, financial condition and results of operations.

We have taken steps to protect the data on our systems and offerings, but our security measures or those of our customers or third-party service providers could be insufficient and breached as a result of third-party action, employee or user errors, technological limitations, defects or vulnerabilities in our systems or offerings or those of our third-party service providers, malfeasance, fraud or malice on the part of employees or third parties, including state-sponsored organizations with significant financial and technological resources, or from failure in technological resources, failure to comply with policies or otherwise. We have experienced and may continue to experience security incidents and attacks of varying types and degrees, including instances where our third-party providers have been impacted by a supply-chain attack and instances where there has been exposure and unauthorized use of credentials of our personnel. In addition, we have identified and been required to remediate or mitigate vulnerabilities in our code and in third-party code. We could be impacted by these and other security incidents and vulnerabilities in the future, and our internal controls and operations regarding security may not be effective in eliminating the risk of compromise of our systems, data and software. Additionally, with our employees and many employees of our third-party service providers working remotely, we may be exposed to increased risks of security breaches or incidents. For example, we have seen an increase in phishing attempts and spam emails over time and it is possible this trend will continue. Also, in connection with geopolitical events and conflicts such as the Hamas-Israel and Russia-Ukraine conflicts, we and our third-party service providers may be vulnerable to a heightened risk of cybersecurity attacks, phishing attacks, viruses, malware, ransomware, hacking or similar breaches or incidents.

#### Table of Contents

There can be no assurance that any security measures that we or our customers or third-party service providers have implemented will be effective against current or future security threats. We have developed systems and processes to protect the integrity, confidentiality, availability and security of our systems, data and software, but our security measures or those of our customers or third-party service providers could fail and result in unauthorized access or damage to, or disablement, encryption, use, disclosure, modification, destruction or loss of, such systems, data and software. Through contractual provisions and third-party risk management processes, we take steps to require that our third-party providers and their subcontractors protect our data, but we cannot ensure the security measures they take will be sufficient to protect our data. A vulnerability in a third-party provider's or a customer's software or systems, a failure of our customers' or third-party providers' safeguards, policies or procedures or a breach or incident of or impacting a customer's or third-party provider's software or systems could result in the compromise of the confidentiality, integrity or availability of our offerings or systems, or our customers' data. Further, security breach techniques are varied and continue to evolve, including through the use of AI to launch more automated, targeted, sophisticated and coordinated attacks, and these attacks may not be detected until after an incident has occurred. We may be unable to implement adequate preventative measures, anticipate, prevent or detect attempted security breaches or other security incidents or react in a timely manner.

We have contractual and other obligations to notify customers, regulators, impacted individuals or others of certain security incidents. We have made such notifications in the past and may be required to do so in the future. Such disclosures or the failure to comply with relevant requirements could lead to adverse consequences. Any security breach or incident that we or our third-party service providers experience, or the perception that one has occurred, could result in a loss of revenue or customer confidence in the security of our products and services, lead to negative publicity or otherwise harm our reputation and brand, reduce the demand for our products and services, disrupt normal business operations, divert management's attention and resources, require us to spend material resources to investigate, correct existing or prevent future security breaches and incidents (including deploying additional personnel and modifying or enhancing our protection technologies and investigating and remediating any information security vulnerabilities), increase our insurance premiums or expose us to legal liabilities, including claims, litigation, regulatory enforcement and orders, disputes, investigations, indemnity obligations, damages for contract breach, penalties and significant costs for remediation, any of which could adversely affect our results of operations. In addition, our remediation efforts may not be successful. We cannot ensure that any limitation of liability provisions in our customer, partner, vendor and other contracts would be enforceable or adequate with respect to any security lapse or breach or other security incident or protect us from any liabilities or damages with respect to any particular claim. These risks will increase as we continue to grow and evolve our offerings to collect, host, store, transmit, and otherwise process increasing volumes of data.

Further, if a security incident or breach occurs with respect to us or a competitor or third-party service provider, our customers and potential customers may lose trust in the security of our products or services or database software generally, which could adversely impact our ability to retain existing customers or attract new customers, which could adversely affect our business, financial condition and results of operations.

Moreover, our insurance coverage, subject to applicable deductibles, may not be adequate for liabilities incurred or cover any indemnification claims against us relating to any security incident or breach or an insurer may deny or exclude from coverage certain types of claims. In the future, we may not be able to secure insurance for such matters on commercially reasonable terms, or at all. The successful assertion of one or more claims against us that exceed available insurance coverage, or the occurrence of changes in our insurance policies, including premium increases or the imposition of large deductible or co-insurance requirements, could adversely affect our business, financial condition and results of operations.

If we are not able to comply with, or are perceived to not comply with U.S. and foreign laws, rules, regulations, industry standards, contractual obligations and other requirements relating to data protection, information security and privacy, our business, financial condition and results of operations could be harmed.

We are subject to a variety of federal, state, local and international laws, rules and regulations, as well as industry standards, internal and external privacy policies and contractual obligations, relating to the collection, use, retention, security, disclosure, transfer, storage and other processing of personal information and other data. The regulatory framework governing these matters worldwide is rapidly evolving and is likely to remain uncertain for the foreseeable future, and it is possible that these or other actual or asserted obligations may be interpreted and applied in manners inconsistent between jurisdictions and in conflict with other rules or our practices. Any actual or perceived failure by us, our suppliers or other third parties with whom we do business to comply with laws, regulations, contractual obligations, or other actual or asserted obligations could result in proceedings against us by governmental entities or others. In many jurisdictions, including the U.S., enforcement actions and consequences for noncompliance are rising. Further, we have had and may in the future receive assertions of noncompliance by private actors. Such assertions may result in fines, investigations or settlement costs. In addition, security advocates and industry groups have regularly proposed, and may propose in the future, self-regulatory standards with which we must legally comply or that contractually apply to us. If we fail to follow these standards even if no personal information is compromised, we may incur significant fines or experience a significant increase in costs.

Internationally, virtually every jurisdiction in which we operate has established its own data security and privacy legal framework with which we or our customers must comply, including but not limited to the United Kingdom ("UK"), Switzerland and the EU. The EU has adopted the General Data Protection Regulation ("GDPR"), which went into effect in May 2018, and together with national legislation, regulations and guidelines of the EU member states, contains numerous requirements relating to the processing of personal data of EU data subjects, including the increased jurisdictional reach of the European Commission ("EC") and more robust compliance obligations. Among other requirements, the GDPR regulates transfers of personal data subject to the GDPR to third countries that have not been found to provide adequate protection to such personal data. The GDPR also introduced numerous data processing and notification requirements and increased fines. In particular, under the GDPR, fines of up to 20 million euros or 4% of the annual global revenue of the noncompliant company, whichever is greater, could be imposed for violations of certain of the GDPR's requirements. Such penalties are in addition to any civil litigation claims by customers and data subjects.

While we have taken steps to mitigate the impact on us with respect to transfers of personal data, the efficacy and longevity of transfer mechanisms upon which we rely remains uncertain. We have in the past and may in the future be required to take additional steps to legitimize any personal data transfers impacted by legal or regulatory developments and be subject to increasing costs of compliance and limitations on our customers and us. More generally, we may find it necessary or desirable to modify our personal data handling practices, and the outcomes of legal challenges relating to cross-border personal data transfer may serve as a basis for our personal data handling practices, or those of our customers and vendors, to continue to be challenged, which may adversely affect our business.

The UK has adopted a version of the GDPR (combining the GDPR and the Data Protection Act 2018), exposing us to two parallel regimes, each of which potentially authorizes similar fines and other potentially divergent enforcement actions. Furthermore, there will be increasing scope for divergence in application, interpretation and enforcement of data protection law between the UK and the European Economic Area. We continue to monitor and review the impact of any resulting changes to EU or UK law that could affect our operations. We may incur liabilities, expenses, costs and other operational losses under the GDPR and data protection laws of the applicable EU member states and the UK in connection with any measures we take to comply with them. Other countries have also passed or are considering passing laws requiring local data residency or restricting the international transfer of personal data.

In addition, domestic data privacy laws continue to evolve and could require us to modify our data processing practices and policies and expose us to further regulatory or operational burdens. For example, the California Consumer Privacy Act ("CCPA") took effect in January 2020 and was subsequently modified by the California Privacy Rights Act ("CPRA"). The CCPA imposes obligations on companies that process California residents' personal information, including an obligation to provide certain new disclosures to such residents and creates new consumer rights. The CCPA provides for civil penalties for violations, as well as a private right of action for certain data breaches that result in the loss of personal information. This private right of action may increase the likelihood of, and risks associated with, data breach litigation. The CPRA also created a new state agency vested with authority to implement and enforce the CCPA and the CPRA.

Additionally, numerous other states have enacted privacy laws that have gone or will go into effect in the near future. While these new privacy laws may share similarities with each other, as well as with the CPRA and CCPA, they differ in many ways and we must comply with each if our operations fall within their scopes. Similar laws have been proposed in other states and at the federal level, and certain states have enacted other privacy-focused legislation such as Washington's enactment of the My Health, My Data Act, which provides for a private right of action. All of these developments reflect a trend toward more stringent privacy legislation in the U.S. We expect that this increase in legislation and regulatory scrutiny will continue to add additional complexity, variation in requirements, restrictions and potential legal risk, require additional investment of resources in compliance programs, impact strategies and the availability of previously useful data and could result in increased compliance costs or changes in business practices and policies.

Additionally, in connection with Couchbase Capella, we may receive higher volumes of data, including sensitive and regulated data, which may require us to comply with additional legal or regulatory requirements. For example, we may store and process protected health information on behalf of our customers, which may subject us to a number of data protection, security and privacy requirements under our contracts and under the Health Insurance Portability and Accountability Act of 1996 ("HIPAA") and other laws and regulations. We may sign business associate agreements with certain of our customers and be directly subject to provisions of HIPAA applicable to business associates, as well as additional contractual requirements. We may also be subject to additional data protection, security and privacy requirements relating to cardholder data, including the Payment Card Industry Data Security Standard. Increased customer adoption of Couchbase Capella may result in further increases in such requirements. If we are, or are perceived to be, unable to maintain the privacy and security of such sensitive and regulated data, we could be subject to claims and demands by private parties, investigations and other proceedings by regulatory authorities, and significant fines, civil and criminal penalties and other liabilities.

Complying with these laws, regulations, contractual or other obligations relating to privacy, security, data protection, transfer or localization and information security may require us to modify our products and services, incur substantial operational costs, modify our data practices and policies and restrict our business operations. Any actual or perceived failure by us to comply with these laws, regulations or other actual or asserted obligations may lead to significant fines, penalties, regulatory investigations, lawsuits, significant costs for remediation, damage to our reputation or other liabilities. The interpretation and application of many privacy, security, and data protection laws, regulations and standards are uncertain, and it is possible that these laws and regulations may be interpreted and applied in a manner that is inconsistent with our data management practices or the features of our services and platform capabilities. If so, in addition to the possibility of fines, lawsuits, regulatory enforcement or orders, investigations, and other proceedings, imprisonment of company officials and public censure, other claims and penalties, significant costs for remediation and damage to our reputation, we could be required to fundamentally change our business activities and practices or modify our services and platform capabilities, any of which could require significant additional expense and have an adverse effect on our business, including impacting our ability to innovate, delaying our product development roadmap and adversely affecting our relationships with customers and our ability to compete. We may be unable to make such changes and modifications in a commercially reasonable manner, or at all, and our ability to develop new product features and services could be limited.

In addition to government activity, privacy advocacy and other industry groups have established or may establish self-regulatory standards that may place additional burdens on our ability to provide our products and services. Our customers expect us to meet certain voluntary certification and other standards established by third parties. If we are unable to maintain these certifications or meet these standards, it could adversely affect our ability to attract new customers or continue providing our services to certain customers and could harm our business. Further, the uncertain and shifting regulatory environment may cause concerns regarding privacy, data protection or security, and may cause our customers to resist providing data that could improve our products and services, or limit the use and adoption of our products and services.

These laws, regulations, rules, industry standards and contractual or other obligations relating to privacy, security, data protection, transfers or localization and information security could require us to take on more onerous obligations in our contracts, restrict our ability to store, transfer and otherwise process data or, in some cases, impact our ability to offer our products and services, to reach existing and potential customers or to derive insights from customer data. The costs of compliance with, and other burdens imposed by, these laws, regulations, standards and obligations, or any inability to adequately address privacy, data protection or security-related concerns, even if unfounded, may limit the use and adoption of our products and services, reduce overall demand for our products and services, make it more difficult to meet expectations from or commitments to customers, impact our reputation or slow the pace at which we close sales transactions, any of which could harm our business, financial condition and results of operations.

### Any future litigation against us could be costly and time-consuming to defend.

In addition to litigation regarding intellectual property, employment, governmental and regulatory investigations, and other claims discussed above, from time to time, we may become involved in various legal proceedings relating to matters incidental to the ordinary course of our business, including commercial, product liability, class action, whistleblower and other litigation, claims and proceedings. Such proceedings can be time-consuming and difficult to estimate, divert management's attention and resources, cause us to incur significant expenses or liability, require us to change our business practices or adversely affect our business, financial condition and results of operations. Because of the potential risks, expenses and uncertainties of litigation, we may, from time to time, settle disputes, even where we have meritorious claims or defenses. In addition, we cannot be sure that our existing insurance coverage for errors and omissions will be adequate or available or continue to be available on acceptable terms.

## Indemnity provisions in various agreements to which we are party potentially expose us to substantial liability for intellectual property infringement, misappropriation or other violation and other losses.

Our agreements with our customers, partners and other third parties may include capped or uncapped indemnification provisions, which may survive termination or expiration of the applicable agreement and under which we agree to indemnify or otherwise be liable to them for losses suffered or incurred from claims of infringement, misappropriation or violation of intellectual property rights, data breaches, damages or other liabilities caused by us, or relating to or arising from our products and services, our acts or omissions under such agreements or other contractual obligations. Large indemnity payments could harm our business, financial condition and results of operations. Although we attempt to contractually limit indemnity obligations, we are not always successful and may still incur substantial liability related to such claims and we may be required to halt certain functions of our products or services. Moreover, even claims that ultimately are unsuccessful could result in expenses in litigation, divert management's time and other resources and harm our business and reputation.

In addition, although we carry general liability insurance, our insurance against this liability may not be adequate to cover a potential claim, and such coverage may not be available to us on acceptable terms, or at all. Any dispute with respect to such obligations could have adverse effects on our relationship with customers, channel parties or other third parties or other existing or potential customers, harm our reputation or reduce demand for our products and services.

#### A portion of our revenue is generated by sales to heavily regulated organizations, which are subject to a number of challenges and risks.

We provide our products and services to heavily regulated organizations, and at times to federal, state and local governments and non-U.S. governments directly and through our partners. Selling to these entities can be highly competitive, expensive and time-consuming, often requiring significant upfront time and expense without any assurance of sales. For instance, highly regulated entities and government customers often require contract terms that differ from our standard arrangements, impose complex compliance requirements, require preferential pricing or "most favored nation" terms and conditions or are otherwise time-consuming and expensive to satisfy. If we undertake to meet special standards or requirements and do not meet them, we could be subject to increased liability. Even if we do meet them, the additional costs associated with providing our services to such customers could harm our financial condition and results of operations.

We have been and are increasingly doing more business in heavily regulated industries. Customers in these industries, may be required to comply with more stringent regulations in connection with subscribing to and implementing our products and services. In addition, regulatory agencies may impose requirements toward third-party vendors generally, or to us in particular, that we may not be able to, or may not choose to, meet. Any changes in the underlying regulatory conditions that affect these types of customers could harm our ability to efficiently provide our products and services to them and to grow or maintain our customer base. Moreover, customers in these heavily regulated areas often have a right to conduct audits of our systems, products and practices. If one or more of such customers determine that some aspect of our business does not meet contractual or regulatory requirements, we may be limited in our ability to continue or expand our business. Each of these difficulties could adversely affect our business and results of operations.

## Failure to comply with anti-bribery, anti-corruption, anti-money laundering and similar laws could subject us to penalties and other adverse consequences.

We are subject to the FCPA, the U.S. domestic bribery statute contained in 18 U.S.C. § 201 and the U.S. Travel Act and other anti-bribery and anti-money laundering laws in countries outside of the U.S. where we conduct our activities. Anti-corruption and anti-bribery laws have been enforced aggressively recently and are interpreted broadly to generally prohibit companies, their employees, agents, representatives, partners and third-party intermediaries from authorizing, offering or providing, directly or indirectly, improper payments or benefits to recipients in the public or private sector.

#### Table of Contents

We may have direct or indirect interactions with officials and employees of government agencies or state-owned or affiliated entities through third parties that sell our products and services and conduct our business abroad or through our employees, agents, representatives, partners and third-party intermediaries. We may be held liable for their corrupt or other illegal activities even if we do not explicitly authorize such activities. These laws also require that we keep accurate books and records and maintain internal controls and compliance procedures designed to prevent any such actions. While we have policies and procedures designed to address compliance with such laws, we cannot ensure that none of our employees, agents, representatives, partners or third-party intermediaries will take actions in violation of our policies and applicable law, for which we may be ultimately held responsible.

Any allegations or violation of the FCPA or other applicable anti-bribery, anti-corruption and anti-money laundering laws could result in whistleblower complaints, sanctions, settlements, prosecution, enforcement actions, fines, damages, adverse media coverage, investigations, loss of export privileges, severe criminal or civil sanctions or suspension or debarment from federal contracts. Responding to any investigation or action will likely result in a materially significant diversion of management's attention and resources and significant defense costs and other professional fees. In addition, the U.S. government may seek to hold us liable for successor liability for FCPA violations committed by companies in which we invest or that we acquire. As a general matter, any of the foregoing could harm our reputation, business, financial condition and results of operations.

We are subject to governmental export control, trade sanctions and import controls that could impair our ability to compete in international markets or subject us to liability if we violate the controls.

Certain of our business activities are subject to the U.S. export control laws and regulations, including the Export Administration Regulations (the "EAR") and the U.S. trade and economic sanctions maintained by the U.S. Department of Treasury's OFAC as well as the U.S. import laws and regulations. The U.S. export control laws and economic sanctions prohibit the export, re-export and in-country transfer of our offerings, including software and services, to certain U.S. embargoed or sanctioned countries and territories, governments and persons, as well as for prohibited end-uses. Further, we incorporate encryption functionality into certain of our products. As a result, we submit reports about certain of our products to the U.S. Department of Commerce's Bureau of Industry and Security to ensure that our exports, re-exports and transfers are in accordance with the EAR. Also, in certain cases, it is possible that a license may be required to export or re-export our products to certain countries and end-users. Obtaining the necessary export license for a particular sale or offering may be time-consuming or unfeasible and may result in the delay or loss of sales opportunities. In addition, various countries regulate the import of certain encryption technology, including through import permit and license requirements, and have enacted laws that could limit our ability to distribute or our customers' ability to implement our products in those countries.

If we do not comply with such U.S. export controls, economic sanctions and import laws and regulations or other similar laws, we could be subject to both civil and criminal penalties, including substantial fines, possible incarceration for employees and managers for willful violations and the possible loss of our export or import privileges. We take precautions designed to ensure that we and our partners comply with all relevant export control, sanctions and import laws and regulations, but we cannot ensure that our measures will always succeed since such laws and regulations are very detailed and technical.

In addition, changes in our products or services or changes in export and import regulations in various countries may create delays in the introduction of our products and services into international markets, prevent the deployment of our products and services globally or, in some cases, prevent or delay the export or import of our products and services to certain countries, governments or persons altogether. Any change in export or import laws or regulations, economic sanctions or related legislation, or in their scope or targets, could result in decreased use of our products and services by or in our decreased ability to export or sell our products and services to, existing or potential end-customers with international operations. Any of the foregoing would adversely affect our business, financial condition and results of operations.

### Our international operations may subject us to greater than anticipated tax liabilities.

Our corporate structure and associated transfer pricing policies contemplate future growth in international markets and consider the functions, risks and assets of the various entities involved in intercompany transactions, the amount of taxes we pay in different jurisdictions, including the U.S., our international business activities, changes in tax rates, new or revised tax laws or interpretations of existing tax laws and policies and our ability to operate our business in a manner consistent with our corporate structure and intercompany agreements. The relevant taxing authorities may challenge our methodologies for pricing intercompany transactions pursuant to intercompany arrangements or disagree with our determinations as to the income and expenses attributable to specific jurisdictions. If such a disagreement were to occur, and our position were not sustained, we could be required to pay additional taxes, interest and penalties, which could result in one-time tax charges, higher effective tax rates, reduced cash flows and lower overall profitability of our operations. Our financial statements could fail to reflect adequate reserves to cover such a contingency.

## Changes in tax laws could materially affect our financial condition, results of operations and cash flows.

The tax regimes we are subject to or operate under, including income and non-income taxes, are unsettled and may be subject to significant change. For example, the Inflation Reduction Act of 2022 imposes a 15% minimum tax for large corporations on global adjusted financial statement income for tax years beginning after December 31, 2022, and a 1% excise tax on certain share repurchases occurring after December 31, 2022. We do not currently expect that the IRA will have a material impact on our income tax liability, but will continue to monitor this change in future periods. We are unable to predict what changes to the tax laws of the U.S. and other jurisdictions may be proposed or enacted in the future or what effect such changes would have on our business. Any significant increase in our future effective tax rate could have a material adverse impact on our business, financial condition, results of operations, or cash flows.

There is also a high level of uncertainty in today's tax environment stemming from both global initiatives put forth by the Organisation for Economic Co-operation and Development (the "OECD") and unilateral measures being implemented by various countries such as Pillar Two and the global minimum tax. If these proposals are passed, it is likely that we will have to pay higher income taxes in countries where such rules are applicable.

As we expand the scale of our international business activities, any changes in the U.S. or foreign taxation of such activities may increase our worldwide effective tax rate and harm our business, financial condition and results of operations. Such changes may also apply retroactively to our historical operations and result in taxes greater than the amounts estimated and recorded in our financial statements.

## Our ability to use our net operating losses may be limited.

As of January 31, 2024, we had federal and state net operating losses ("NOLs") of \$335.6 million and \$193.2 million, respectively, which may be available to offset taxable income in the future. A lack of future taxable income would adversely affect our ability to utilize these NOLs before they expire. Unused U.S. federal NOLs for taxable years beginning before January 1, 2018, may be carried forward to offset future taxable income, if any, until such unused NOLs expire. Under the Tax Cuts and Jobs Act, U.S. federal NOLs arising in tax years beginning after December 31, 2017 can be carried forward indefinitely, but the deductibility of such U.S. federal NOLs is limited to 80% of current year taxable income.

Of our U.S. federal NOLs, \$166.9 million may be carried forward indefinitely with utilization limited to 80% of taxable income. The remaining \$168.7 million will begin to expire in 2028. Our state NOLs carryforwards begin to expire in 2026.

Under Section 382 of the Internal Revenue Code of 1986, as amended, if a corporation undergoes an "ownership change," the corporation's ability to use its pre-change NOLs to offset its post-change income may be limited. In general, an "ownership change" will occur if there is a cumulative change in our ownership by "5-percent shareholders" that exceeds 50 percentage points over a rolling three-year period. Similar rules may apply under state tax laws. Subsequent ownership changes and changes to the U.S. tax rules in respect of the utilization of NOLs may further affect the limitation in future years.

There is also a risk that due to U.S. federal or state regulatory changes, such as suspensions on the use of NOLs, our existing NOLs could expire or otherwise be unavailable to offset future income tax liabilities.

## Adverse outcomes in tax audits or changes in our effective tax rate or tax liability may have an adverse effect on our results of operations.

We are, and expect to continue to be, subject to review and audit by the U.S. Internal Revenue Service and other tax authorities in various domestic and foreign jurisdictions. As a result, we may receive assessments in multiple jurisdictions on various tax-related assertions. Taxing authorities have made inquiries of us and may in the future

investigate or challenge our tax positions and methodologies on various matters, including our positions regarding the collection of sales and use taxes and the jurisdictions in which we are subject to taxes, which could expose us to additional taxes. We assess the likelihood of adverse outcomes resulting from any ongoing tax examinations to determine the adequacy of our provision for income taxes. These assessments can require considerable judgments and estimates. The calculation of our tax liabilities involves uncertainties in the application of complex tax laws and regulations in a variety of jurisdictions. There can be no assurance that our tax positions and methodologies or calculation of our tax liabilities are accurate or that the outcomes from tax examinations will not have an adverse effect on our financial condition and results of operations. A difference in the ultimate resolution of tax uncertainties from what is currently estimated could adversely affect our financial condition and results of operations.

Taxing authorities may successfully assert that we should have collected or in the future should collect sales and use, value added or similar taxes and we could be subject to liability with respect to past or future sales, which could adversely affect our results of operations.

We collect sales tax in a number of jurisdictions. Sales and use, value added and similar tax laws and rates vary greatly by jurisdiction. Certain jurisdictions in which we do not collect such taxes may assert that such taxes are applicable, which could result in tax assessments, penalties and interest, and we may be required to collect such taxes in the future. Such tax assessments, penalties, interest or future requirements would adversely affect our financial condition and results of operations.

## Our reported financial results may be adversely affected by changes in accounting principles generally accepted in the U.S.

Generally accepted accounting principles in the U.S. ("GAAP") are subject to interpretation by the Financial Accounting Standards Board, the SEC and other various bodies formed to promulgate and interpret appropriate accounting principles. Changes in accounting principles applicable to us, or varying interpretations of current accounting principles, in particular with respect to revenue recognition, could have a significant effect on our reported results of operations and could affect the reporting of transactions completed before the announcement of the change. Further, any difficulties in the implementation of changes in accounting principles, including the ability to modify our accounting systems, could cause us to fail to meet our financial reporting obligations, which could result in regulatory discipline and harm investors' confidence in us.

# If our estimates or judgments relating to our critical accounting policies prove to be incorrect, our results of operations could be adversely affected.

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in our financial statements and accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, as described in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations." The results of these estimates form the basis for making judgments about the recognition and measurement of certain assets and liabilities and revenue and expenses that is not readily apparent from other sources. Our accounting policies that involve judgment include standalone selling prices for each distinct performance obligation, capitalized internal-use software costs, expected period of benefit for deferred commissions, valuation of our common stock prior to our IPO, valuation of stock-based awards, determination of allowance for credit losses, incremental borrowing rate used to measure operating lease liabilities, and accounting for income taxes. If our assumptions change or if actual circumstances differ from those in our assumptions, our results of operations could be adversely affected or fall below the expectations of securities analysts and investors, resulting in a decline in the market price of our common stock.

## We are obligated to maintain proper and effective internal control over financial reporting, and any failure to maintain the adequacy of these internal controls may adversely affect investor confidence in our company and, as a result, the value of our common stock.

We are subject to the reporting requirements of the Exchange Act, the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), and the listing standards of the Nasdaq Global Select Market. Our management and other personnel devote a substantial amount of time to comply with these requirements. Moreover, these laws, regulations and standards are subject to varying interpretations and revisions. Such changes could result in continuing uncertainty regarding compliance matters and higher legal and financial costs. We continue to invest resources to comply with evolving laws, regulations and standards, which may result in increased general and administrative expenses and a diversion of management's time and attention. If our compliance efforts differ from the activities intended by regulatory or governing bodies due to ambiguities related to their application and practice, regulatory authorities may initiate legal proceedings against us and our business may be adversely affected.

The Sarbanes-Oxley Act requires, among other things, that we maintain effective disclosure controls and procedures and internal control over financial reporting. We are continuing to refine and improve our disclosure controls, internal controls and other procedures to ensure disclosures required in SEC filings are timely recorded, processed, summarized and reported and that disclosures under the Exchange Act are accumulated and communicated to our principal executive and financial officers. We have expended, and anticipate that we will continue to expend, significant resources to maintain and improve the effectiveness of our disclosure controls and procedures and internal control over financial reporting.

Our current controls and any new controls that we develop may become inadequate because of changes in the conditions in our business, including increased complexity resulting from any international expansion. Further, weaknesses in our disclosure controls or our internal control over financial reporting may be discovered in the future. Any failure or difficulties to develop, maintain, implement or improve effective controls could harm our results of operations or cause us to fail to meet our reporting obligations and may result in a restatement of our financial statements for prior periods. Any failure to implement and maintain effective internal control over financial reporting could also adversely affect the results of periodic management evaluations and annual independent registered public accounting firm attestation reports regarding the effectiveness of our internal control over financial reporting that we will eventually be required to include in our periodic reports that will be filed with the SEC. Any of the foregoing could have an adverse effect on our business, financial condition and results of operations and could cause investors to lose confidence in our reported financial and other information, which would likely adversely affect the market price of our common stock. In addition, if we are unable to continue to meet these requirements, we may not be able to remain listed on the Nasdaq Global Select Market.

Our independent registered public accounting firm will be required to formally attest to the effectiveness of our internal control over financial reporting when we are no longer an "emerging growth company." At such time, our independent registered public accounting firm may issue a report that is adverse in the event it is not satisfied with the level at which our internal control over financial reporting is documented, designed or operating.

### Risks Related to Ownership of Our Common Stock and Governance Matters

## Operating as a public company has and will require us to incur substantial costs and will require substantial management attention.

As a public company, we incur substantial legal, accounting and other expenses that we did not incur as a private company. Compliance with the rules and regulations of the Exchange Act, the applicable requirements of the Sarbanes-Oxley Act, the Dodd-Frank Wall Street Reform and Consumer Protection Act, the rules and regulations of the SEC and the listing standards of the Nasdaq Global Select Market have increased and may further increase our legal and financial compliance costs, and increase demand on our systems, particularly after we are no longer an "emerging growth company." In addition, as a public company, we may be subject to stockholder activism, which can lead to additional substantial costs, distract management and impact the manner in which we operate our business in unanticipated ways. As a result of disclosure of information in filings required of a public company, our business and financial condition become more visible, which may result in threatened or actual litigation, including by competitors.

There is also an increasing focus from regulators, certain investors, and other stakeholders concerning environmental, social and governance ("ESG") matters. Our actual or perceived failure to achieve some or all of our ESG-related initiatives, goals, or commitments or maintain ESG practices that meet evolving stakeholder expectations or regulatory requirements could harm our reputation, adversely impact our ability to attract and retain employees or customers, and expose us to increased scrutiny from ESG-focused investors, regulatory authorities, and others, or subject us to liabilities, which may further adversely impact our business, financial condition, or results of operations.

Our management team may not successfully or efficiently manage the significant regulatory oversight and reporting obligations under the federal or state securities laws and the continuous scrutiny of securities analysts and investors. These obligations and constituents require significant attention from our senior management and could divert their attention away from the day-to-day management of our business, which could adversely affect our business, financial condition and results of operations.

We are an "emerging growth company" and the reduced disclosure requirements applicable to emerging growth companies may make our common stock less attractive to investors.

We are an "emerging growth company," as defined in the JOBS Act, and we intend to take advantage of certain available exemptions, including not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. In addition, we have elected to take

advantage of the extended transition period under the JOBS Act to delay adoption of new or revised accounting pronouncements applicable to public companies until such pronouncements are made applicable to private companies. Any difficulties in implementing these pronouncements could cause failure to meet our financial reporting obligations, regulatory discipline or harm investor confidence. We may take advantage of these exemptions for so long as we are an "emerging growth company," which could be for as long as five full fiscal years following the completion of our IPO. If some investors find our common stock less attractive as a result, there may be a less active trading market and a more volatile market price for our common stock.

Our executive officers, directors and holders of 5% or more of our common stock continue to have substantial control over us, which will limit your ability to influence the outcome of important transactions, including a change in control.

Our executive officers, directors and our stockholders who own 5% or more of our outstanding common stock and their affiliates, in the aggregate, beneficially own a substantial portion of the outstanding shares of our common stock. As a result, these stockholders, if acting together, will be able to influence or control matters requiring stockholders' approval, including the election of directors and the approval of mergers, acquisitions or other extraordinary transactions and may vote in a way with which you disagree and which may be adverse to your interests. This concentration of ownership may have the effect of delaying, preventing or deterring a change in control of our company, could deprive our stockholders of an opportunity to receive a premium for their common stock as part of a sale of our company, and might ultimately affect the market price of our common stock.

## The market price of our common stock may continue to be volatile, and you could lose all or part of your investment.

The market price of our common stock may continue to be volatile and subject to fluctuations in response to various factors, some of which are beyond our control. These fluctuations could cause you to lose all or part of your investment in our common stock and include the following:

- price and volume fluctuations and investor confidence in the overall stock market and in technology stocks or those in our industry in particular;
- sales or short selling of our common stock or related derivative securities;
- failure of securities analysts to maintain coverage of us or publish inaccurate or unfavorable research about our business, changes in financial estimates by securities analysts or our failure to meet these estimates or the expectations of investors;
- any changes in the financial projections we may provide to the public or our failure to meet those projections;
- announcements by us or our competitors of new offerings or platform features and market acceptance of such new offerings or platform features;
- the public's reaction to our press releases, other public announcements and SEC filings;
- rumors and market speculation in our industry, whether or not involving us;
- actual or anticipated changes or fluctuations in our results of operations or key business metrics;
- actual or anticipated developments in our business, our competitors' businesses or the competitive landscape generally;
- announced or completed acquisitions of businesses, offerings or technologies by us or our competitors;
- developments or disputes concerning our intellectual property or other proprietary rights;
- litigation involving us, our industry, or both, or investigations by regulators into our operations or those of our competitors, or securities class action litigation against us;
- new laws, regulations, rules or industry standards or new interpretations of existing laws, regulations, rules or industry standards applicable to our business:
- changes in accounting standards, policies, guidelines, interpretations or principles;
- · any significant change in our management; and
- general economic conditions and slow or negative growth of our markets and other geopolitical developments.

Stock markets in general, and the markets for technology stocks in particular, have previously experienced and may in the future experience extreme volatility, including as a result of global economic conditions. Furthermore, the market price of our common stock may be adversely affected by third parties, such as short sellers, trying to drive down the price of our common stock. These broad market and industry factors may seriously harm the market price of our common stock, regardless of our operating performance.

Sales of substantial amounts of shares of our common stock in the public market, or the perception that such sales might occur, could cause the market price of our common stock to decline or impair our ability to raise capital through the sale of additional equity securities.

If our stockholders sell, or the market perceives that our stockholders intend to sell, a substantial number of shares of our common stock in the public market, the market price of our common stock could decline and our ability to raise capital through the sale of additional equity securities could be impaired. Many of our existing equity holders have substantial unrecognized gains on the value of the equity they hold, and may take, or attempt to take, steps to sell, directly or indirectly, their shares or otherwise secure the value of their unrecognized gains on those shares.

In addition, certain of our stockholders are entitled, under our investors' rights agreement, to require us to register their shares for public sale in the U.S. Sales of our common stock pursuant to registration rights may make it more difficult for us to sell equity securities in the future at an appropriate time. These sales also could cause the market price of our common stock to fall and make it more difficult for you to sell our common stock.

The issuance of additional stock in connection with financings, acquisitions, investments, our equity compensation plans or otherwise will dilute all other stockholders.

Subject to applicable rules and regulations and our amended and restated certificate of incorporation, we may issue additional common stock or securities convertible into common stock from time to time in connection with a financing, acquisition, investment, our equity compensation plans or otherwise. Any such issuance could result in substantial dilution to our existing stockholders and cause the market price of our common stock to decline.

Delaware law and provisions in our amended and restated certificate of incorporation and amended and restated bylaws could make a merger, tender offer or proxy contest difficult, thereby depressing the market price of our common stock.

Our status as a Delaware corporation and the anti-takeover provisions of the Delaware General Corporation Law may discourage, delay or prevent a change in control by prohibiting us from engaging in a business combination with an interested stockholder for a period of three years after the date of the transaction in which the person became an interested stockholder, regardless of the potential benefit to our existing stockholders. In addition, our amended and restated certificate of incorporation and amended and restated bylaws contain provisions that may make the acquisition of our company more difficult, including the following:

- our board of directors will be classified into three classes of directors with staggered three-year terms, and directors will only be able to be removed from office for cause;
- certain amendments to our amended and restated certificate of incorporation will require the approval of at least 2/3 of our then-outstanding common stock;
- our stockholders will only be able to take action at a meeting of stockholders and not by by written consent;
- our amended and restated certificate of incorporation will not provide for cumulative voting;
- vacancies on our board of directors will be able to be filled only by our board of directors and not by stockholders;
- a special meeting of our stockholders may only be called by the chairperson of our board of directors, our Chief Executive Officer or a majority of our board of directors;
- certain litigation against us can only be brought in Delaware;
- our amended and restated certificate of incorporation authorizes undesignated preferred stock, the terms of which may be established and shares of which may be issued without further action by our stockholders; and
- advance notice procedures apply for stockholders to nominate candidates for election as directors or to bring matters before an annual meeting of stockholders.

These provisions, alone or together, could discourage, delay or prevent a transaction involving a change in control of our company. These provisions could also discourage proxy contests, election of stockholders' director nominees and other corporate actions our stockholders may desire, any of which, under certain circumstances, could limit the opportunity for our stockholders to receive a premium as part of a sale of our company and could also affect the price of our common stock in a change of control.

Our amended and restated bylaws designate a state or federal court located within the State of Delaware as the exclusive forum for substantially all disputes between us and our stockholders, which could limit our stockholders' ability to choose the judicial forum for disputes with us or our directors, officers or employees.

Our amended and restated bylaws provide that, unless we consent in writing to the selection of an alternative forum, to the fullest extent permitted by law, the sole and exclusive forum for (i) any derivative action or proceeding brought on our behalf, (ii) any action asserting a claim of breach of a fiduciary duty owed by any of our directors, officers or other employees to us or our stockholders, (iii) any action arising pursuant to any provision of the Delaware General Corporation Law, our amended and restated certificate of incorporation or our amended and restated bylaws or (iv) any other action asserting a claim that is governed by the internal affairs doctrine shall be the Court of Chancery of the State of Delaware (or, if the Court of Chancery does not have jurisdiction, the federal district court for the District of Delaware), in all cases subject to the court having jurisdiction over indispensable parties named as defendants.

Section 22 of the Securities Act of 1933, as amended (the "Securities Act"), creates concurrent jurisdiction for federal and state courts over all such Securities Act actions. Accordingly, both state and federal courts have jurisdiction to entertain such claims. To prevent having to litigate claims in multiple jurisdictions and the threat of inconsistent or contrary rulings by different courts, among other considerations, our amended and restated bylaws further provide that the federal district courts of the U.S. will be the exclusive forum for resolving any complaints asserting a cause of action arising under the Securities Act. We note, however, that investors cannot waive compliance with the federal securities laws and the rules and regulations thereunder, and that there is uncertainty as to whether a court would enforce this exclusive forum provision. If a court were to find either exclusive-forum provision in our amended and restated bylaws to be inapplicable or unenforceable in an action, we may incur additional costs associated with resolving the dispute in other jurisdictions, which could harm our results of operations.

If securities or industry analysts do not publish research or publish inaccurate or unfavorable research about us, our business or our market, or if they change their recommendations regarding our common stock adversely, the market price and trading volume of our common stock could decline.

The trading market for our common stock depends, in part, on the research and reports that securities or industry analysts publish about us, our business, our market or our competitors. The analysts' estimates are based upon their own opinions and are often different from our estimates or expectations. If any of the analysts who cover us change their recommendation regarding our common stock adversely, provide more favorable relative recommendations about our competitors or publish inaccurate or unfavorable research about our business, the market price of our common stock would likely decline. If few securities analysts commence coverage of us, or if one or more of these analysts cease coverage of us or fail to publish reports on us regularly, we could lose visibility in the financial markets and demand for our securities could decrease, which could cause the market price and trading volume of our common stock to decline.

## We do not intend to pay dividends for the foreseeable future.

We have never declared nor paid cash dividends on our capital stock. We currently intend to retain any future earnings to finance the operation and expansion of our business, and we do not expect to declare or pay any dividends in the foreseeable future. As a result, stockholders must rely on sales of their common stock after price appreciation, if any, as the only way to realize any future gains on their investment in our common stock.

## Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

#### **Unregistered Sales of Equity Securities**

None.

#### **Use of Proceeds**

On July 26, 2021, we completed our IPO. The offer and sale of the shares in the IPO were registered under the Securities Act pursuant to a registration statement on Form S-1 (File No. 333-257205), which was declared effective by the SEC on July 21, 2021. There has been no material change in the use of the IPO proceeds as described in our final prospectus filed with the SEC on July 22, 2021, pursuant to Rule 424(b) of the Securities Act and other periodic reports previously filed with the SEC.

## Item 3. Defaults Upon Senior Securities

None.

## **Item 4. Mine Safety Disclosures**

Not Applicable.

#### **Item 5. Other Information**

## Securities Trading Plans of Directors and Executive Officers

On March 11, 2024 during our last fiscal quarter, Jeff Epstein, a member of our board of directors adopted a "Rule 10b5-1 trading arrangement" as defined in Regulation S-K Item 408. The trading arrangement provides for the sale from time to time of an aggregate of up to 121,623 shares of our common stock. The trading arrangement is intended to satisfy the affirmative defense in Rule 10b5-1(c). The duration of the trading arrangement is until June 7, 2025, or earlier if all transactions under the trading arrangement are completed.

On April 8, 2024 during our last fiscal quarter, Carol Carpenter, a member of our board of directors adopted a "Rule 10b5-1 trading arrangement" as defined in Regulation S-K Item 408. The trading arrangement provides for the sale from time to time of an aggregate of up to 10,800 shares of our common stock. The trading arrangement is intended to satisfy the affirmative defense in Rule 10b5-1(c). The duration of the trading arrangement is until July 30, 2025, or earlier if all transactions under the trading arrangement are completed.

No other directors or officers, as defined in Rule 16a-1(f), adopted, modified and/or terminated a "Rule 10b5-1 trading arrangement" or a "non-Rule 10b5-1 trading arrangement," as defined in Regulation S-K Item 408, during the last fiscal quarter.

## Item 6. Exhibits

Exhibit Number	Description	Form	File No.	Exhibit	Filing Date	Filed Herewith			
10.1†+	Loan and Security Agreement, dated as of February 7, 2024, between the Company and MUFG Ltd.					X			
31.1	Certification of Principal Executive Officer Pursuant to Rules 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant								
31.2	to Section 302 of the Sarbanes-Oxley Act of 2002. Certification of Principal Financial Officer Pursuant to Rules 13a-14(a) and					X			
31.2	15d-14(a) under the Securities Exchange Act of 1934, as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.					X			
32.1*	Certifications of Principal Executive Officer and Principal Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.					X			
101.INS	Inline XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the					Α			
	Inline XBRL document.					X			
101.SCH	Inline XBRL Taxonomy Extension Schema Document.					X			
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.					X			
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.					X			
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document.					X			
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.					X			
104	Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibits 101) - the cover page interactive data is embedded within the								
	Inline XBRL document or included within the Exhibit 101 attachments.					X			
*	The certifications attached as Exhibit 32.1 that accompanies this Quarterly Report on Form 10-Q are deemed furnished and not filed with the SEC and are not to be incorporated by reference into any filing of Couchbase, Inc. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date of this Quarterly Report on Form 10-Q, irrespective of any general incorporation language contained in such filing.								
†	Certain portions of the Loan and Security Agreement have been omitted to preswill furnish copies of any such information to the SEC upon request.		·						
+	The schedules to the Loan and Security Agreement have been omitted from this Company will furnish copies of any such schedules to the SEC upon request.	s filing pur	suant to Item	601(a)(5) o	of Regulation	S-K. The			

### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

### COUCHBASE, INC.

D (	r	-	202	
Date: .	lune	h.	2024	٠

By:

/s/ MATTHEW M. CAIN

Matthew M. Cain

Chair, President and Chief Executive Officer
(Principal Executive Officer)

By:

/s/ GREG HENRY

Greg Henry

Senior Vice President and Chief Financial Officer
(Principal Financial Officer)

By:

/s/ BILL CAREY

Bill Carey

Vice President and Chief Accounting Officer
(Principal Accounting Officer)

CERTAIN CONFIDENTIAL INFORMATION, MARKED BY [\*\*\*], HAS BEEN EXCLUDED FROM THIS EXHIBIT BECAUSE (I) IT IS NOT MATERIAL AND (II) THE REGISTRANT CUSTOMARILY AND ACTUALLY TREATS THE INFORMATION AS PRIVATE AND CONFIDENTIAL.

### LOAN AND SECURITY AGREEMENT

This **LOAN AND SECURITY AGREEMENT** (this "**Agreement**"), dated as of February 7, 2024 (the "**Effective Date**"), between **MUFG BANK, LTD.** ("**Bank**"), and **COUCHBASE, INC.**, a Delaware corporation ("**Borrower**"), provides the terms on which Bank shall lend to Borrower and Borrower shall repay Bank.

### 1 ACCOUNTING AND OTHER TERMS

- 1.1 Accounting Terms. Except as otherwise expressly provided herein, all accounting terms not otherwise defined herein shall be construed in conformity with GAAP. Financial statements and other information required to be delivered by Borrower to Bank shall be prepared in accordance with GAAP as in effect at the time of such preparation, except with respect to unaudited financial statements (a) noncompliance with FAS 123R, and (b) for the absence of footnotes and subject to year-end audit and purchase accounting adjustments. Notwithstanding the foregoing, (i) for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, Indebtedness of Borrower and its Subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded, (ii) if Borrower notifies Bank that Borrower requests an amendment to any provision hereof to eliminate the effect of any change occurring after the date hereof in GAAP or in the application thereof on the operation of such provision (or if Bank notifies Borrower that Bank requests an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such change in GAAP or in the application thereof, then such provision shall be interpreted on the basis of GAAP as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance herewith and (iii) any obligations of any Person that are or would have been treated as operating leases for purposes of GAAP prior to the issuance by the Financial Accounting Standards Board on February 25, 2016 of an Accounting Standards Update (the "ASU") shall continue to be accounted for as operating leases for purposes of all financial definitions, calculations and covenants for purpose of this Agreement (whether or not such operating lease obligations were in effect on such date) notwithstanding the fact that such obligations are required in accordance with the ASU (on a prospective or retroactive basis or otherwise) to be treated as capitalized lease obligations in accordance with GAAP.
- 1.2 Rates. Bank does not warrant or accept responsibility for, and shall not have any liability with respect to (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Prime Rate, the Term SOFR Reference Rate or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, or any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement) will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Prime Rate, the Term SOFR Reference Rate, Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. Bank and its affiliates or other related entities may engage in transactions that affect the calculation of the Prime Rate, the Term SOFR Reference Rate, Term SOFR, any alternative, successor or

replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto, in each case, in a manner adverse Borrower. Bank may select information sources or services in its reasonable discretion to ascertain the Prime Rate, the Term SOFR Reference Rate, Term SOFR or any other Benchmark, or any component definition thereof or rates referred to in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to Borrower or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

#### 2 LOAN AND TERMS OF PAYMENT

**2.1 Promise to Pay**. Borrower hereby unconditionally promises to pay Bank the outstanding principal amount of all Credit Extensions and accrued and unpaid interest thereon as and when due in accordance with this Agreement.

## 2.2 Revolving Line.

- (a) <u>Availability</u>. Subject to the terms and conditions of this Agreement, Bank shall make Advances not exceeding the Availability Amount. Amounts borrowed under the Revolving Line may be repaid and, prior to the Revolving Line Maturity Date, reborrowed, subject to the applicable terms and conditions precedent herein. Each Advance shall be comprised entirely of SOFR Advances. Each SOFR Advance shall be in an aggregate amount of at least \$1,000,000 or a larger multiple of \$100,000; provided, that a SOFR Advance may be in an aggregate amount that is equal to the entire unused balance of the Availability Amount. There shall not be more than a total of seven (7) SOFR Advances outstanding at any time.
- (b) Letter of Credit Sublimit. Subject to the terms and conditions of this Agreement, Bank shall issue for the account of Borrower such Letters of Credit as Borrower may request from time to time by delivering to Bank a duly executed letter of credit application on Bank's standard form; provided, however, that the outstanding and undrawn amounts under all such Letters of Credit (i) shall not at any time exceed the Letter of Credit Sublimit, and (ii) shall be deemed to constitute Advances for the purpose of calculating availability under the Availability Amount; provided further, that the maturity date of any such Letter of Credit shall not be later than the Revolving Line Maturity Date. Any drawn but unreimbursed amounts under any Letters of Credit shall be charged as Advances against the Revolving Line. All Letters of Credit shall be in form and substance and shall include terms (including, without limitation, the expiration date thereof) acceptable to Bank in its sole discretion and shall be subject to the terms and conditions of Bank's form of letter of credit application and agreement (as such form may be amended, restated, supplemented, replaced, or otherwise modified from time to time) and any other agreement that Borrower may enter into with Bank in connection with such Letters of Credit.
- (c) <u>Termination</u>. The Revolving Line terminates on the Revolving Line Maturity Date, when the outstanding principal amount of all Advances, the accrued and unpaid interest thereon, and all other Obligations relating to the Revolving Line shall be immediately due and payable.
- **2.3 Overadvances**. If, at any time, the outstanding principal amount of any Advances exceeds the lesser of either the Revolving Line or the Availability Amount, Borrower shall immediately pay to Bank in cash the amount of such excess (such excess, the "Overadvance"). Without limiting Borrower's obligation to repay Bank any Overadvance, Borrower agrees to pay Bank interest on the outstanding

amount of any Overadvance, on demand, at a per annum rate equal to the rate that is otherwise applicable to Advances plus three percent (3.0%).

#### 2.4 Procedures for Borrowing.

- (a) Subject to the prior satisfaction of all other applicable conditions to the making of an Advance set forth in this Agreement, each Advance shall be made upon Borrower's irrevocable notice to Bank. Each such notice shall be in the form of a written Borrowing Request, appropriately completed and signed by a Authorized Signer of Borrower, or may be given by telephone to Bank (if promptly confirmed by such a written Borrowing Request consistent with such telephonic notice) and must be received by Bank not later than 2:00 p.m. Pacific time three (3) U.S. Government Securities Business Days prior to the date of the requested Borrowing.
- (b) Each Borrowing Request for a Borrowing pursuant to this Section shall specify the following information: (i) the aggregate amount of the requested Advance; (ii) the date of such Advance (which shall be a Business Day); and (iii) the Interest Period therefor. If no Interest Period is specified with respect to any requested SOFR Advance, Borrower shall be deemed to have selected an Interest Period of one month's duration.
- (c) Bank shall have received satisfactory evidence that the Board has approved that such Authorized Signer may provide such notices and request Advances. Bank shall credit proceeds of an Advance to an account designated by Borrower to Bank in writing. Bank may make Advances under this Agreement based on instructions from an Authorized Signer or without instructions if the Advances are necessary to meet Obligations which have become due.
- 2.5 Interest Elections. Borrower may elect to convert any Advance to an Advance with a different Interest Period or to continue as the same type of such Advance. Each such election pursuant to this Section shall be made upon Borrower's irrevocable notice to Bank. Each such notice shall be in the form of a written Interest Election Request, appropriately completed and signed by a Authorized Signer of Borrower, or may be given by telephone to Bank (if promptly confirmed in writing by delivery of such a written Interest Election Request consistent with such telephonic notice) and must be received by Bank not later than the time that a Borrowing Request would be required under Section 2.4 if Borrower were requesting an Advance of the type resulting from such election to be made on the effective date of such election. Each Interest Election Request shall specify the Advance to which such Interest Election Request applies and, if different options are being elected with respect to different portions thereof, the portions thereof to be allocated to each resulting Advance, the effective date of the election made pursuant to such Interest Election Request, which shall be a Business Day, and the Interest Period therefor after giving effect to such election. If Borrower fails to deliver a timely and complete Interest Election Request with respect to a SOFR Advance prior to the end of the Interest Period therefor, then, unless such SOFR Advance is repaid as provided herein, Borrower shall be deemed to have selected that such SOFR Advance shall automatically be continued as a SOFR Advance with an Interest Period of one month at the end of Default is continuing, at Bank's election, (i) no outstanding Advance may be converted to or continued as a SOFR Advance and (ii) unless repaid as provided herein,

each SOFR Advance shall automatically be converted to a Prime Rate Advance at the end of the Interest Period therefor.

#### 2.6 Payment of Interest on the Credit Extensions.

- (a) Interest Rate. Subject to Section 2.6(b), the principal amount outstanding under the Revolving Line shall accrue interest at a floating per annum rate equal to (i) for each Prime Rate Advance, the greater of (A) one-half of one percent (0.50%) above the Prime Rate, and (B) three and three-quarters of one percent (3.75%), or (ii) for each SOFR Advance, three percent (3.0%) above the Term SOFR for the Interest Period therefor.
- (b) <u>Default Rate</u>. Upon the occurrence and during the continuance of an Event of Default, at Bank's election, Obligations shall bear interest at a rate per annum which is two percent (2.0%) above the rate that is otherwise applicable thereto (the "**Default Rate**"). Fees and expenses which are required to be paid by Borrower pursuant to the Loan Documents (including, without limitation, Bank Expenses) but are not paid when due shall bear interest until paid at a rate equal to the highest rate applicable to the Obligations. Payment or acceptance of the increased interest rate provided in this Section 2.6(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Bank.
- (c) <u>Payment</u>. Accrued interest on each Loan shall be payable in arrears on each Interest Payment Date applicable thereto and at such other times as may be specified herein; provided that (i) interest accrued pursuant to paragraph (b) of this Section shall be payable on demand, (ii) in the event of any repayment or prepayment of any Loan, accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment and (iii) in the event of any conversion of any SOFR Advance prior to the end of the Interest Period therefor, accrued interest on such Advance shall be payable on the effective date of such conversion.
- (d) <u>Interest Computation</u>. All interest hereunder shall be computed on the basis of a year of three hundred sixty (360) days (or in the case of interest computed by reference to the Prime Rate, such interest shall be computed on the basis of a year of 365 days (or 366 days in a leap year)), and in each case shall be payable for the actual number of days elapsed (including the first day but excluding the last day). All interest hereunder on any Advance shall be computed on a daily basis based upon the outstanding principal amount of such Advance as of the applicable date of determination. The applicable Prime Rate or Term SOFR shall be determined by Bank, and such determination shall be conclusive absent manifest error.
- (e) <u>Term SOFR Conforming Changes</u>. In connection with the use or administration of Term SOFR, Bank will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document. Bank will promptly notify Borrower of the effectiveness of any Conforming Changes in connection with the use or administration of Term SOFR.

# **2.7 Fees**. Borrower shall pay to Bank:

- (a) [Reserved];
- (b) <u>Unused Revolving Line Facility Fee</u>. Payable quarterly in arrears on the last day of each calendar quarter occurring thereafter prior to the Revolving Line Maturity Date, and on the Revolving

Line Maturity Date, a fee (the "Unused Revolving Line Facility Fee") in an amount equal to one-quarter of one percent (0.25%) per annum of the average unused portion of the Revolving Line, as determined by Bank, computed on the basis of a year with the applicable number of days as set forth in Section 2.6(d). The unused portion of the Revolving Line, for purposes of this calculation, shall be calculated on a calendar year basis and shall equal the difference between (i) the Revolving Line, and (ii) the average for the period of the daily closing balance of the Revolving Line outstanding;

- (c) Letter of Credit Fees. Borrower shall pay Bank (i) a letter of credit fee (the "Letter of Credit Fee") equal to two percent (2.00%) per annum times the daily face amount of each outstanding Letter of Credit for the period from and excluding the date of issuance of such Letter of Credit to and including the date of expiration or termination of such Letter of Credit, and (ii) without duplication of costs, expenses and fees otherwise payable to Bank hereunder, all reasonable and documented out-of-pocket costs and expenses incurred by Bank on account of such Letters of Credit and any other standard issuance or other fees that Bank notifies Borrower it will charge for issuing and processing Letters of Credit. The Letter of Credit Fee shall be payable in arrears on the last day of each calendar quarter and on the Revolving Line Maturity Date. All Letter of Credit Fees shall be computed on the basis of a year of three hundred sixty (360) days and shall be payable for the actual number of days elapsed (including the first day but excluding the last day). Any other fees payable to Bank pursuant to this paragraph shall be payable within ten (10) days after demand.
- (d) <u>Bank Expenses</u>. All Bank Expenses (including reasonable and documented attorneys' fees and expenses for documentation and negotiation of this Agreement) incurred through and after the Effective Date, when due (or, if no stated due date, upon demand by Bank).
- (e) <u>Fees Fully Earned</u>. Unless otherwise provided in this Agreement or in a separate writing by Bank, Borrower shall not be entitled to any credit, rebate, or repayment of any fees earned by Bank pursuant to this Agreement notwithstanding any termination of this Agreement or the suspension or termination of Bank's obligation to make loans and advances hereunder. Bank may deduct amounts owing by Borrower under the clauses of this Section 2.7 pursuant to the terms of Section 2.8(c). Bank shall provide Borrower written notice of deductions made from any of Borrower's deposit accounts pursuant to the terms of the clauses of this Section 2.7.

### 2.8 Repayment; Prepayments; Payments Generally; Application of Payments; Debit of Accounts.

- (a) <u>Repayment</u>. The outstanding principal amount of all Advances, the accrued and unpaid interest thereon, and all other Obligations relating to the Revolving Line shall be immediately due and payable on the Revolving Line Maturity Date.
- (b) <u>Prepayments</u>. Borrower may prepay any Advances at any time from time to time without penalty or premium. Borrower shall provide Bank prior notice of any prepayment in the form of a Prepayment Notice, appropriately completed and signed by a Responsible Officer of the Borrower. Such Prepayment Notice must be received by Bank (i) in the case of prepayment of a SOFR Advance, not later than 2:00 p.m. Pacific time three (3) U.S. Government Securities Business Days before the date of prepayment, or (ii) in the case of prepayment of a Prime Rate Advance, not later than 2:00 p.m. Pacific time one (1) Business Day before the date of prepayment. Each Prepayment Notice shall specify (x) the prepayment date and (y) the principal amount of each Advance or portion thereof to be prepaid. Each Prepayment Notice shall be irrevocable; provided that such Prepayment Notice may state that such prepayment is contingent upon the effectiveness of other credit facilities, securities offerings or other

transactions, the proceeds of which will be used to refinance in full this Agreement, in which case such notice may be revoked by Borrower (by notice to the Bank on or prior to the specified effective date) if such condition is not satisfied.

- (c) <u>Payments Generally</u>. All payments (including prepayments) to be made by Borrower under any Loan Document shall be made in immediately available funds in Dollars, without setoff, counterclaim or deduction before 2:00 p.m. Pacific time on the date when due. Payments of principal and/or interest received after 2:00 p.m. Pacific time are considered received at the opening of business on the next Business Day. Unless otherwise provided in this Agreement, when a payment is due on a day that is not a Business Day, the payment shall be due the next Business Day, and additional fees or interest, as applicable, shall continue to accrue until paid.
- (d) <u>Application of Payments</u>. Prior to the existence of an Event of Default, payments shall be applied as directed by Borrower. Upon the occurrence and during the continuance of an Event of Default, Bank has the exclusive right to determine the order and manner in which all payments with respect to the Obligations may be applied and Borrower shall have no right to specify the order or the accounts to which Bank shall allocate or apply any payments required to be made by Borrower to Bank or otherwise received by Bank under this Agreement when any such allocation or application is not specified elsewhere in this Agreement.
- (e) <u>Debit of Accounts</u>. Bank may debit any of Borrower's deposit accounts, for principal and interest payments or any other amounts Borrower owes Bank when due. These debits shall not constitute a set-off.
- 2.9 Compensation for Losses. In the event of (a) the payment of any principal of any SOFR Advance other than on the last day of the Interest Period applicable thereto (including as a result of an Event of Default), (b) the conversion of any SOFR Advance other than on the last day of the Interest Period applicable thereto (including as a result of an Event of Default), or (c) the failure to borrow, convert, continue or prepay any SOFR Advance on the date specified in any notice delivered pursuant hereto, then, in any such event, Borrower shall compensate Bank for any loss, cost and expense attributable to such event, including any loss, cost or expense arising from the liquidation or redeployment of funds or from any fees payable. A certificate of Bank setting forth any amount or amounts that Bank is entitled to receive pursuant to this Section shall be delivered to Borrower and shall be conclusive absent manifest error. Borrower shall pay Bank the amount shown as due on any such certificate within 10 days after receipt thereof.

#### 2.10 Increased Costs.

- (a) <u>Increased Costs Generally</u>. If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve (including pursuant to regulations issued from time to time by the Federal Reserve Board for determining the maximum reserve requirement (including any emergency, special, supplemental or other marginal reserve requirement) with respect to eurocurrency funding (currently referred to as "Eurocurrency liabilities" in Regulation D)), special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, Bank;
- (ii) subject Bank to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) and (d) of the definition of Excluded Taxes and (C) Connection Income Taxes)

on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto; or

(iii) impose on Bank any other condition, cost or expense (other than Taxes) affecting this Agreement or Advances;

and the result of any of the foregoing shall be to increase the cost to Bank of making, converting to, continuing or maintaining any Advance or of maintaining its obligation to make any such Advance, or to increase the cost to Bank, or to reduce the amount of any sum received or receivable by Bank hereunder (whether of principal, interest or any other amount) then, upon request of Bank, Borrower will pay to Bank, such additional amount or amounts as will compensate Bank for such additional costs incurred or reduction suffered.

- (b) <u>Capital Requirements</u>. If Bank determines that any Change in Law affecting Bank or any lending office of Bank or Bank's holding company, if any, regarding capital or liquidity requirements, has or would have the effect of reducing the rate of return on Banks's capital or on the capital of Bank's holding company, if any, as a consequence of this Agreement, the Revolver Commitments of Bank or the Advances made by Bank to a level below that which Bank or Bank's holding company could have achieved but for such Change in Law (taking into consideration Bank's policies and the policies of Bank's holding company with respect to capital adequacy), then from time to time Borrower will pay to Bank such additional amount or amounts as will compensate Bank or Bank's holding company for any such reduction suffered.
- (c) <u>Certificates for Reimbursement</u>. A certificate of Bank setting forth the amount or amounts necessary to compensate Bank or its holding company, as the case may be, as specified in paragraph (a) or (b) of this Section and delivered to Borrower, shall be conclusive absent manifest error. Borrower shall pay Bank the amount shown as due on any such certificate within 10 days after receipt thereof.
- (d) <u>Delay in Requests</u>. Failure or delay on the part of Bank to demand compensation pursuant to this Section shall not constitute a waiver of Bank's right to demand such compensation; provided that Borrower shall not be required to compensate Bank pursuant to this Section for any increased costs incurred or reductions suffered more than nine months prior to the date that Bank notifies Borrower of the Change in Law giving rise to such increased costs or reductions, and of Bank's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the nine-month period referred to above shall be extended to include the period of retroactive effect thereof).

## 2.11 Taxes.

- (a) <u>Defined Terms</u>. For purposes of this Section, the term "Applicable Law" includes FATCA.
- (b) <u>Payments Free of Taxes</u>. Any and all payments by or on account of any obligation of Borrower under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by Applicable Law. If any Applicable Law (as determined in the good faith discretion of Borrower or Bank) requires the deduction or withholding of any Tax from any such payment by Borrower or Bank, then such party shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with Applicable Law and, if such Tax is an Indemnified Tax, then the sum payable by Borrower shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and

withholdings applicable to additional sums payable under this section) Bank receives an amount equal to the sum it would have received had no such deduction or withholding been made.

- (c) <u>Payment of Other Taxes by Borrower</u>. Borrower shall timely pay to the relevant Governmental Authority in accordance with Applicable Law, or at the option of Bank timely reimburse it for the payment of, any Other Taxes.
- (d) <u>Indemnification by Borrower</u>. Borrower shall indemnify Bank, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this section) payable or paid by Bank or required to be withheld or deducted from a payment to Bank and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to Borrower by Bank shall be conclusive absent manifest error.
- (e) Evidence of Payments. As soon as practicable after any payment of Taxes by Borrower to a Governmental Authority pursuant to this Section, Borrower shall deliver to Bank the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Bank

### (f) Status of Lenders

- (i) Bank or any other lender under any Loan Document that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to Borrower, at the time or times reasonably requested by Borrower, such properly completed and executed documentation reasonably requested by Borrower as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, Bank or any other lender under this Agreement or any Loan Document, if reasonably requested by Borrower, shall deliver such other documentation prescribed by Applicable Law or reasonably requested by Borrower as will enable Borrower to determine whether or not such lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in paragraphs (f)(ii)(A), (ii)(B) and (ii)(D) of this section) shall not be required if in Bank or any other such lender's reasonable judgment such completion, execution or submission would subject Bank or any such lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such lender,
  - (ii) Without limiting the generality of the foregoing, in the event that Borrower is a U.S. Person,
- (A) Bank or any lender that is a U.S. Person shall deliver to Borrower on or about the date on which bank or any such lender becomes a lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower), executed copies of IRS Form W-9 certifying that Bank or any such lender is exempt from U.S. federal backup withholding tax
- (B) Bank or any foreign lender shall, to the extent it is legally entitled to do so, deliver to Borrower (in such number of copies as shall be requested by Borrower) on or about the date on which Bank or any such foreign lender becomes a lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower), whichever of the following is applicable:

(1) in the case of Bank or any foreign lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed copies of IRS Form W-8BEN or IRS Form W-8BEN-E establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN or IRS Form W-8BEN-E establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;

#### (2) executed copies of IRS Form W-8ECI;

(3) in the case of Bank or any foreign lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the U.S. Code, (x) a certificate to the effect that such Bank or foreign lender is not a "bank" within the meaning of U.S. Code Section 881(c)(3)(A), a "10 percent shareholder" of Borrower within the meaning of Section 871(h)(3)(B) of the U.S. Code, or a "controlled foreign corporation" related to Borrower as described in U.S. Code Section 881(c)(3)(C) (a "U.S. Tax Compliance Certificate") and (y) executed copies of IRS Form W-8BEN or IRS Form W-8BEN-E; or

(4) to the extent Bank or foreign lender is not the beneficial owner, executed copies of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN, IRS Form W-8BEN-E, a U.S. Tax Compliance Certificate and/or other certification documents from each beneficial owner, as applicable; <u>provided</u> that if Bank or the foreign lender is a partnership and one or more direct or indirect partners of Bank or such foreign lender are claiming the portfolio interest exemption, Bank or such foreign lender may provide a U.S. Tax Compliance Certificate on behalf of each such direct and indirect partner;

(C) Bank or any foreign lender shall, to the extent it is legally entitled to do so, deliver to Borrower (in such number of copies as shall be requested by Borrower) on or about the date on which Bank or such foreign lender becomes a lender under this Agreement (and from time to time thereafter upon the reasonable request of Borrower), executed copies of any other form prescribed by Applicable Law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by Applicable Law to permit to determine the withholding or deduction required to be made; and

(D) if a payment made to Bank or any lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in U.S. Code Section 1471(b) or 1472(b), as applicable), Bank or any such lender shall deliver to Borrower at the time or times prescribed by law and at such time or times reasonably requested by Borrower such documentation prescribed by Applicable Law (including as prescribed by U.S. Code Section 1471(b)(3)(C)(i)) and such additional documentation reasonably requested by Borrower as may be necessary for Borrower to comply with their obligations under FATCA and to determine that Bank or any such lender has complied with Bank or any such lender's obligations under FATCA or to determine the amount, if any, to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

(E) Bank and any other lender under any Loan Document agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect,

it shall update such form or certification or promptly notify Borrower in writing of its legal inability to do so.

- (g) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this section (including by the payment of additional amounts pursuant to this section), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (h) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this paragraph, in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.
- (h) <u>Survival</u>. Each party's obligations under this Section shall survive any assignment of rights by, or the replacement of, Bank, the termination of the commitments hereunder and the repayment, satisfaction or discharge of all obligations under any Loan Document.
  - **2.12 Inability to Determine Rates**. Subject to Section 2.13, if, on or prior to the first day of any Interest Period for any SOFR Advance:
- (a) Bank determines (which determination shall be conclusive and binding absent manifest error) that "Term SOFR" cannot be determined pursuant to the definition thereof, or
- (b) Bank determine that for any reason in connection with any request for a SOFR Advance or a conversion thereto or a continuation thereof that Term SOFR for any requested Interest Period with respect to a proposed SOFR Advance does not adequately and fairly reflect the cost to Bank of making and maintaining such Advance,

then, in each case, Bank will promptly so notify Borrower.

Upon notice thereof by Bank to Borrower, any obligation of Bank to make SOFR Advances, and any right of Borrower to continue SOFR Advances shall be suspended (to the extent of the affected SOFR Advances or affected Interest Periods) until Bank revokes such notice. Upon receipt of such notice, (i) Borrower may revoke any pending request for a borrowing of, conversion to or continuation of SOFR Advances (to the extent of the affected SOFR Advances or affected Interest Periods) or, failing that, Borrower will be deemed to have converted any such request into a request for a Borrowing of or conversion to Prime Rate Advances in the amount specified therein and (ii) any outstanding affected SOFR Advances will be deemed to have been converted into Prime Rate Advances at the end of the applicable Interest Period. Upon any such conversion, Borrower shall also pay accrued interest on the amount so converted, together with any additional amounts required pursuant to Section 2.9.

### 2.13 Benchmark Replacement Setting.

- (a) <u>Benchmark Replacement</u>. Notwithstanding anything to the contrary herein or in any other Loan Document (and any interest rate swap or other funding arrangement entered by Bank to make or maintain any Advance shall be deemed not to be a "Loan Document" for purposes of this paragraph), upon the occurrence of a Benchmark Transition Event, Bank and Borrower may amend this Agreement to replace the then-current Benchmark with a Benchmark Replacement. Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. (New York City time) on the fifth Business Day after Bank has posted such proposed amendment to Borrower. No replacement of a Benchmark with a Benchmark Replacement pursuant to this section will occur prior to the applicable Benchmark Transition Start Date.
- (b) <u>Benchmark Replacement Conforming Changes</u>. In connection with the use, administration, adoption or implementation of a Benchmark Replacement, Bank will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (c) <u>Notices</u>; Standards for Decisions and Determinations. Bank will promptly notify Borrower of (i) the implementation of any Benchmark Replacement, and (ii) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. Bank will notify Borrower of (x) the removal or reinstatement of any tenor of a Benchmark and (y) the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by Bank under the section, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action, will be conclusive and binding absent manifest error and may be made in Bank's sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this section.
- (d) <u>Unavailability of Tenor of Benchmark</u>. Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by Bank in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then Bank may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then Bank may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.
- (e) <u>Benchmark Unavailability Period</u>. Upon Borrower' receipt of notice of the commencement of a Benchmark Unavailability Period, Borrower may revoke any pending request for a

borrowing of SOFR Advances, or a conversion to or a continuation of any SOFR Advances to be made during any Benchmark Unavailability Period and, failing that, Borrower will be deemed to have converted any such request into a request for a borrowing of Prime Rate Advances and any outstanding SOFR Advances will automatically convert to Prime Rate Advances at the end of the applicable Interest Period.

**2.14 Illegality**. If Bank determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for Bank or its applicable lending office to make, maintain or fund Advances whose interest is determined by reference to SOFR, the Term SOFR Reference Rate or Term SOFR, or to determine or charge interest based upon SOFR, the Term SOFR Reference Rate or Term SOFR, then, upon notice thereof by Bank to Borrower (an "**Illegality Notice**"), (a) any obligation of Bank to make SOFR Advances, and any right of Borrower to continue SOFR Advances shall be suspended until Bank notifies Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of an Illegality Notice, Borrower shall, if necessary to avoid such illegality, upon demand from Bank, prepay or, if applicable, convert all SOFR Advances to Prime Rate Advances, on the last day of the Interest Period therefor, if all Bank may lawfully continue to maintain such SOFR Advances to such day, or immediately, if Bank may not lawfully continue to maintain such SOFR Advances to such day. Upon any such prepayment or conversion, Borrower shall also pay accrued interest on the amount so prepaid or converted, together with any additional amounts required pursuant to Section 2.9.

#### 2.15 Accordion.

(a) Increase in Revolver Commitments. Subject to the conditions set forth in Section 2.15(b), Borrower may request that the Revolver Commitments be increased by an aggregate amount of up to Twenty-Five Million Dollars (\$25,000,000) (each such increase, an "Increase"; and each date on which an Increase is made, an "Increase Date"). Borrower may submit a request in writing in accordance with the notice provisions of this Agreement (each, an "Increase Request") not later than 2:00 p.m. Pacific time three (3) U.S. Government Securities Business Days prior to the date of the proposed Increase that Bank increase its Revolver Commitments in connection with a proposed Increase (it being agreed and understood that Bank shall not be obligated to increase its Revolver Commitments). If Bank does not agree to increase its Revolver Commitments in connection with such proposed Increase, then Borrower may consult with Bank to invite one or more prospective lenders satisfactory to Bank in its reasonable discretion to become a lender under this Agreement in connection with a proposed Increase. Any Increase shall be in an amount of at least \$1,000,000 and integral multiples of \$100,000 in excess thereof. In no event may the Revolver Commitments be increased pursuant to this Section 2.15 on more than three (3) occasions in the aggregate for all such Increases. Submission by Borrower of an Increase Request shall be deemed to be a certification that the representations and warranties in this Agreement are true, accurate, and correct in all material respects on the date of such submission; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and correct in all material respects as of such specific date. For the avoidance of doubt, it is understood and agreed that in no event shall the aggregate amount of the Increases to the Revolver Commitments exceed Twenty-Five Million Dollars (\$25,000,000).

(b) <u>Conditions to Increase</u>. Each of the following shall be conditions precedent to any Increase of the Revolver Commitments: (i) Borrower has obtained the commitment of either Bank or one or more other prospective lenders reasonably satisfactory to Bank to provide the applicable Increase; (ii)

if such Increase is not provided by Bank in full, any new lenders shall duly execute and deliver a joinder agreement or amendment to this Agreement, in form and substance reasonably satisfactory to Bank, to which such lenders, Borrower and Bank are party; (iii) each of the conditions precedent set forth in Section 3.2(b) and (c) have been satisfied; (iv) Borrower shall have paid all increase and closing fees as may be required by Bank or any other applicable lender as a condition to such Increase; (v) Borrower shall arrange for the delivery of all Loan Documents, certificates and other items as Bank or other lender may reasonably require in connection with such Increase; and (vi) no Event of Default has occurred and is continuing.

(c) <u>Miscellaneous</u>. The Revolver Commitments and Advances established pursuant to this Section 2.15 shall constitute Revolver Commitments and Advances under, and shall be entitled to all the benefits afforded by, this Agreement and the other Loan Documents, and shall, without limiting the foregoing, benefit equally and ratably from any guarantees and the security interests created by the Loan Documents. Borrower shall take any actions reasonably required by Bank to ensure and demonstrate that the Liens and security interests granted by the Loan Documents continue to be perfected under the UCC or otherwise after giving effect to the establishment of any such new Revolver Commitments.

### 3 CONDITIONS OF LOANS

- **3.1 Conditions Precedent to Initial Credit Extension**. Bank's obligation to make the initial Credit Extension is subject to the condition precedent that Bank shall have received, in form and substance satisfactory to Bank, such documents, and completion of such other matters, as Bank may request, including, without limitation:
  - (a) duly executed Loan Agreement;
- (b) the Operating Documents of Borrower and (i) a long-form good standing certificate of Borrower certified by the Secretary of State of the State of Delaware and (ii) good standing/foreign qualification certificates certified by the Secretary of State (or equivalent agency) of each other jurisdiction in which Borrower is qualified to conduct business and the failure of so qualified in such jurisdiction would reasonably result in a Material Adverse Change, in each case, as of a date no earlier than thirty (30) days prior to the Effective Date;
- (c) certificate duly executed by a Responsible Officer or secretary of Borrower with respect to Borrower's (i) Operating Documents, (ii) good standing certificates, (iii) Borrowing Resolutions and (iv) incumbency;
- (d) certified copies, dated as of a recent date, of searches for financing statements filed in the central filing office of the State of Delaware against Borrower, accompanied by written evidence (including any UCC termination statements) that the Liens indicated in any such financing statements constitute Permitted Liens;
- (e) evidence satisfactory to Bank that (i) that certain Amended and Restated Loan and Security Agreement, dated as of January 29, 2021, between Borrower and Silicon Valley Bank, has been terminated and all amounts thereunder shall have been paid in full, and (ii) all Liens granted in connection therewith have been terminated;
  - (f) duly executed Perfection Certificate of Borrower;

- (g) evidence satisfactory to Bank that the insurance policies and endorsements required by Section 6.7 hereof are in full force and effect, together with appropriate evidence showing lender loss payable and/or additional insured clauses or endorsements in favor of Bank; and
  - (h) payment of the fees and Bank Expenses then due as specified in Section 2.7(a) and (d) hereof.
- **3.2** Conditions Precedent to all Credit Extensions. Bank's obligation to make each Credit Extension, including the initial Credit Extension, is subject to the following conditions precedent:
  - (a) timely receipt of the Credit Extension request and any materials and documents required by Section 2.4;
- (b) the representations and warranties in this Agreement shall be true, accurate, and correct in all material respects on the date of the proposed Credit Extension and on the Funding Date of each Credit Extension; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and correct in all material respects as of such date, and no Event of Default shall have occurred and be continuing or result from the Credit Extension. Each Credit Extension is Borrower's representation and warranty on that date that the representations and warranties in this Agreement remain true, accurate, and correct in all material respects; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and correct in all material respects as of such date; and
  - (c) Bank determines to its reasonable satisfaction that there has not been a Material Adverse Change.
- **3.3 Covenant to Deliver**. Borrower agrees to deliver to Bank each item required to be delivered to Bank under this Agreement as a condition precedent to any Credit Extension. Borrower expressly agrees that a Credit Extension made prior to the receipt by Bank of any such item shall not constitute a waiver by Bank of Borrower's obligation to deliver such item, and the making of any Credit Extension in the absence of a required item shall be in Bank's sole discretion.

### 4 CREATION OF SECURITY INTEREST

- **4.1 Grant of Security Interest**. Borrower hereby grants Bank, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Bank, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. Borrower acknowledges that it previously has entered, and/or may in the future enter, into Bank Services Agreements with Bank. Regardless of the terms of any Bank Services Agreement, Borrower agrees that any amounts Borrower owes Bank thereunder shall be deemed to be Obligations hereunder and that it is the intent of Borrower and Bank to have all such Obligations secured by the first priority perfected security interest in the Collateral granted herein (subject only to Permitted Liens).
- **4.2 Termination**. If this Agreement is terminated, Bank's Lien in the Collateral shall continue until the Obligations (other than inchoate indemnity obligations) are repaid in full in cash. Upon payment in full in cash of the Obligations (other than inchoate indemnity obligations) and at such time as Bank's obligation to make Credit Extensions has terminated, Bank shall, at the sole cost and expense of

Borrower, release its Liens in the Collateral and all rights therein shall revert to Borrower. In the event (x) all Obligations (other than inchoate indemnity obligations), except for Bank Services, are satisfied in full, and (y) this Agreement is terminated, Bank shall terminate the security interest granted herein upon Borrower providing cash collateral acceptable to Bank in its good faith business judgment for Bank Services, if any. In the event such Bank Services consist of outstanding Letters of Credit, Borrower shall provide to Bank cash collateral in an amount equal to (x) if such Letters of Credit are denominated in Dollars, then at least one hundred five percent (105.0%); and (y) if such Letters of Credit are denominated in a Foreign Currency, then at least one hundred ten percent (110.0%), of the Dollar Equivalent of the face amount of all such Letters of Credit plus all interest, fees, and costs due or to become due in connection therewith (as estimated by Bank in its business judgment), to secure all of the Obligations relating to such Letters of Credit.

- **4.3 Priority of Security Interest**. Borrower represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject only to Permitted Liens). If Borrower shall acquire a commercial tort claim valued in excess of Two Million Five Hundred Thousand Dollars (\$2,500,000), individually or in the aggregate, Borrower shall promptly notify Bank in a writing signed by Borrower of the general details thereof and grant to Bank in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Bank.
- **4.4 Authorization to File Financing Statements**. Borrower hereby authorizes Bank to file financing statements, without notice to Borrower, with all appropriate jurisdictions to perfect or protect Bank's interest or rights hereunder, including a notice that any disposition of the Collateral, by either Borrower or any other Person, shall be deemed to violate the rights of Bank under the Code. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Bank's discretion.

# REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants as follows:

**5.1 Due Organization, Authorization; Power and Authority.** Borrower is duly existing and in good standing as a Registered Organization in its jurisdiction of formation and is qualified and licensed to do business and is in good standing in any jurisdiction in which the conduct of its business or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to have a material adverse effect on Borrower's business or operations. In connection with this Agreement, Borrower has delivered to Bank a completed certificate signed by Borrower, entitled "Perfection Certificate" (the "**Perfection Certificate**"). Borrower represents and warrants to Bank that except as otherwise disclosed to Bank from time to time pursuant to Section 7.2, (a) Borrower's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Borrower is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Borrower's organizational identification number or accurately states that Borrower has none; (d) the Perfection Certificate accurately sets forth Borrower's place of business, or, if more than one, its chief executive office as well as Borrower's mailing address (if different than its chief executive office); (e) Borrower (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Borrower and each of its Subsidiaries is accurate and correct in all material respects (it being understood and agreed that Borrower may from time to time update certain information in the Perfection

Certificate after the Effective Date to the extent permitted by one or more specific provisions in this Agreement). If Borrower is not now a Registered Organization but later becomes one, Borrower shall promptly notify Bank of such occurrence and provide Bank with Borrower's organizational identification number.

The execution, delivery and performance by Borrower of the Loan Documents to which it is a party have been duly authorized, and do not (i) conflict with any of Borrower's organizational documents, (ii) contravene, conflict with, constitute a default under or violate any material Requirement of Law, (iii) contravene, conflict or violate any applicable order, writ, judgment, injunction, decree, determination or award of any Governmental Authority by which Borrower or any of its Subsidiaries or any of their property or assets may be bound or affected, (iv) require any action by, filing, registration, or qualification with, or Governmental Approval from, any Governmental Authority (except such Governmental Approvals which have already been obtained and are in full force and effect and filings necessary to perfect Liens granted under the Loan Documents), or (v) conflict with, contravene, constitute a default or breach under, or result in or permit the termination or acceleration of, any material agreement by which Borrower is bound. Borrower is not in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a material adverse effect on Borrower's business or operations.

**5.2 Collateral**. The security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject to Permitted Liens). Borrower has good title to, rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien hereunder, free and clear of any and all Liens except Permitted Liens. As of the Effective Date, Borrower has no Collateral Accounts at or with any bank or financial institution other than Bank or Bank's Affiliates except for the Collateral Accounts described in the Perfection Certificate delivered to Bank in connection herewith. The Accounts are bona fide, existing obligations of the Account Debtors.

As of the Effective Date, the Collateral is not in the possession of any third party bailee (such as a warehouse) or maintained at locations except as otherwise provided in the Perfection Certificate.

All Inventory is in all material respects of good and marketable quality, free from material defects. Borrower is the sole owner of the Intellectual Property which it owns or purports to own except for (a) non- exclusive licenses granted to its customers in the ordinary course of business and licenses that could not result in a legal transfer of title of the licensed property but that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States, (b) over-the-counter software that is commercially available to the public, (c) material Intellectual Property licensed to Borrower and noted on the Perfection Certificate or as otherwise disclosed to Bank and (d) open-source software. Each Patent which it owns or purports to own and which is material to Borrower's business is valid and enforceable, and no part of the Intellectual Property which Borrower owns or purports to own and which is material to Borrower's business has been judged invalid or unenforceable, in whole or in part. To the best of Borrower's knowledge, no claim has been made that any part of the Intellectual Property violates the rights of any third party except to the extent such claim would not reasonably be expected to have a material adverse effect on Borrower's business or operations.

As of the Effective Date, except as noted on the Perfection Certificate, Borrower is not a party to, nor is it bound by, any Restricted License.

## 5.3 [Reserved].

- **5.4 Litigation**. Except as disclosed on the Perfection Certificate or as required to be disclosed pursuant to Section 6.2 (such disclosure shall be deemed to update the applicable provision of the Perfection Certificate), there are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing against Borrower or any of its Subsidiaries that could reasonably be expected to result in damages payable by Borrower or any of its Subsidiaries of more than, individually or in the aggregate, Two Million Five Hundred Thousand Dollars (\$2,500,000).
- **5.5 Financial Statements; Financial Condition**. All consolidated financial statements for Borrower and any of its Subsidiaries delivered to Bank fairly present in all material respects (subject to normal fiscal year-end adjustments) Borrower's consolidated financial condition and Borrower's consolidated results of operations. There has not been any material deterioration in Borrower's consolidated financial condition since the date of the most recent financial statements submitted to Bank.
- **5.6 Solvency**. The fair salable value of Borrower's consolidated assets (including goodwill minus disposition costs) exceeds the fair value of Borrower's liabilities; Borrower is not left with unreasonably small capital after the transactions in this Agreement; and Borrower is able to pay its debts (including trade debts) as they mature.
- 5.7 Regulatory Compliance. Borrower is not an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act of 1940, as amended. Borrower is not engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors). Borrower (a) has complied in all material respects with all Requirements of Law, and (b) has not violated any Requirements of Law the violation of which could reasonably be expected to have a material adverse effect on its business. None of Borrower's or any of its Subsidiaries' properties or assets has been used by Borrower or any Subsidiary or, to the best of Borrower's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than legally. Borrower and each of its Subsidiaries have obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue their respective businesses as currently conducted.
- **5.8 Subsidiaries; Investments**. Borrower does not own any stock, partnership, or other ownership interest or other equity securities except for Permitted Investments.
- **5.9 Tax Returns and Payments; Pension Contributions.** Borrower has timely filed all required income and other material tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except (a) to the extent such taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as such reserve or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made therefor, or (b) if such taxes, assessments, deposits and contributions do not, individually or in the aggregate, exceed Two Million Five Hundred Thousand Dollars (\$2,500,000).

To the extent Borrower defers payment of any contested taxes, Borrower shall post bonds or take any other steps required to prevent the Governmental Authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien." Borrower is unaware of any claims or adjustments proposed for any of Borrower's prior tax years which could result in additional taxes

becoming due and payable by Borrower in excess of Two Million Five Hundred Thousand Dollars (\$2,500,000). Borrower has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and Borrower has not withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

- **5.10 Use of Proceeds**. Borrower shall use the proceeds of the Credit Extensions solely as working capital and to fund its general business requirements and not for personal, family, household or agricultural purposes.
- **5.11 Full Disclosure**. To the best of Borrower's knowledge, no written representation, warranty or other statement of Borrower or any of its Subsidiariesin any report, certificate or written statement by submission to Bank, as of the date such representation, warranty, or other statement was made, taken together with all such written reports, written certificates and written statements submitted to Bank, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the reports, certificates or written statements not misleading (it being recognized by Bank that the projections and forecasts provided by Borrower in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).
- **5.12 Definition of "Knowledge**." For purposes of the Loan Documents, whenever a representation or warranty is made to Borrower's knowledge or awareness, to the "best of" Borrower's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer.
- **5.13 Sanctions**. Neither Borrower nor any of its Subsidiaries is: (a) in violation of any Sanctions; or (b) a Sanctioned Person. Neither Borrower nor any of its Subsidiaries, directors, or officers, or, to the knowledge of Borrower, any of its employees, agents or Affiliates: (i) conducts any business or engages in any transaction or dealing with any Sanctioned Person, including making or receiving any contribution of funds, goods or services to or for the benefit of any Sanctioned Person; (ii) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to any Sanctions; (iii) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Sanctions; or (iv) otherwise engages in any transaction that could cause Bank to violate any Sanctions.

### **6** AFFIRMATIVE COVENANTS

Borrower shall do all of the following:

### 6.1 Government Compliance.

(a) Except as permitted by Section 7.1 or Section 7.3, maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Borrower's business or operations. Borrower shall comply, and have each Subsidiary comply, in all material respects, with all laws, ordinances and regulations to which it is subject.

(b) Obtain all of the Governmental Approvals necessary for the performance by Borrower of its obligations under the Loan Documents to which it is a party and the grant of a security interest to Bank in all of its property. Upon Bank's reasonable request, Borrower shall promptly provide copies of any such obtained Governmental Approvals to Bank.

### **6.2 Financial Statements, Reports**. Provide Bank with the following by submitting to Bank:

- (a) for each of the first three fiscal quarters of Borrower's fiscal year, as soon as available, but no later than the earlier of (i) forty-five (45) days after the end of each such fiscal quarter, or (ii) five (5) days after filing with the SEC, a consolidated balance sheet and income statement covering Borrower's and each of its Subsidiary's operations for such quarter in a form of presentation reasonably acceptable to Bank (the "Quarterly Financial Statements");
- (b) together with the Quarterly Financial Statements delivered pursuant to Section 6.2(a) and the Annual Financial Statements delivered pursuant to Section 6.2(d), a completed Compliance Statement, confirming that, as of the end of such fiscal period, Borrower was in full compliance with all of the terms and conditions of this Agreement, and setting forth calculations showing compliance with the financial covenants set forth in this Agreement and such other information as Bank may reasonably request;
- (c) within the earlier of (i) fifteen (15) days after approval by the Board or (ii) sixty (60) days after each fiscal year of Borrower, and contemporaneously with any updates or amendments thereto, (A) annual budgeted income statement for the then-current fiscal year of Borrower, and (B) annual financial projections for the then-current fiscal year, in each case as approved by the Board;
- (d) as soon as available, but no later than the earlier of (i) one hundred twenty (120) days following the end of Borrower's fiscal year, or (ii) five (5) days after filing with the SEC, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified (other than a qualification with respect to "going concern" or like qualification or exception solely as a result of the final maturity date of any Advance being scheduled to occur within twelve (12) months from the date of such opinion) opinion on the financial statements from an independent certified public accounting firm reasonably acceptable to Bank (the "Annual Financial Statements");

### (e) [reserved];

- (f) within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower and/or any Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be:
- (g) within five (5) days of delivery, copies of all statements, reports and notices made available to all of Borrower's security holders (in their capacity as such) or to any holders of Subordinated Debt;
- (h) prompt report of any legal actions pending or threatened in writing against Borrower or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, Two Million Five Hundred Thousand Dollars (\$2,500,000) or more;
  - (i) prompt written notice of the occurrence of a Default or Event of Default; and

(j) promptly, from time to time, such other information regarding Borrower or compliance with the terms of any Loan Documents as reasonably requested by Bank.

Notwithstanding the foregoing, information required to be delivered pursuant to clauses (a), (b), (d), (f), (g), (h) and (i) of this Section 6.2 shall be deemed to have been delivered if such information, or one or more annual, quarterly or other periodic reports containing such information, shall be available on the website of the SEC at <a href="http://www.sec.gov">http://www.sec.gov</a>. In the event that the Borrower is no longer subject to the reporting requirements under the Exchange Act, Borrower shall furnish Bank the information required to be delivered pursuant to clauses (a), (b) and (d) of this Section 6.2 in accordance with the same timeframes as were required when Borrower was a company subject to the reporting requirements under the Exchange Act.

Any submission by Borrower of a Compliance Statement or any other financial statement submitted to Bank shall be deemed to be a representation by Borrower that (a) as of the date of such Compliance Statement or other financial statement, the information and calculations set forth therein are true, accurate and correct in all material respects, (b) as of the end of the compliance period set forth in such submission, Borrower is in complete compliance with all required covenants except as noted in such Compliance Statement or other financial statement, as applicable; (c) as of the date of such submission, no Events of Default have occurred or are continuing; (d) all representations and warranties other than any representations or warranties that are made as of a specific date in Article 5 remain true and correct in all material respects as of the date of such submission except as noted in such Compliance Statement or other financial statement, as applicable; (e) as of the date of such submission, Borrower and each of its Subsidiaries has timely filed all required tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except as otherwise permitted pursuant to the terms of Section 5.9; and (f) as of the date of such submission, no Liens have been levied or claims made against Borrower or any of its Subsidiaries relating to unpaid employee payroll or benefits of which Borrower has not previously provided written notification to Bank.

6.3 [Reserved].

6.4 [Reserved].

**6.5 Taxes; Pensions**. Timely file, and require each of its Subsidiaries to timely file, all required income and other material tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all material foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower and each of its Subsidiaries, except for deferred payment of any taxes contested pursuant to the terms of Section 5.9 hereof, and shall deliver to Bank, on demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms.

**6.6 Access to Collateral; Books and Records**. At reasonable times, on five (5) Business Days' notice (provided no notice is required if an Event of Default has occurred and is continuing), Bank, or its agents, shall have the right to inspect the Collateral and the right to audit and copy Borrower's Books. Such inspections and audits shall be conducted no more often than once every twelve (12) months, unless an Event of Default has occurred and is continuing in which case such inspections and audits shall occur as often as Bank shall determine is necessary. The foregoing inspections and audits shall be conducted at Borrower's expense and the charge therefor shall be One Thousand Dollars (\$1,000) per person per day (or such higher amount as shall represent Bank's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Borrower and Bank schedule an audit more than ten (10)

days in advance, and Borrower cancels or seeks to or reschedules the audit with less than ten (10) days written notice to Bank, then (without limiting any of Bank's rights or remedies) Borrower shall pay Bank a fee of One Thousand Dollars (\$1,000) plus any out-of-pocket expenses incurred by Bank to compensate Bank for the anticipated costs and expenses of the cancellation or rescheduling.

#### 6.7 Insurance.

- (a) Keep its business and the Collateral insured for risks and in amounts standard for companies in Borrower's industry and location and as Bank may reasonably request. Insurance policies shall be in a form, with financially sound and reputable insurance companies that are not Affiliates of Borrower, and in amounts that are satisfactory to Bank. All property policies shall have Bank's loss payable endorsement showing Bank as lender loss payee. All liability policies shall show, or have endorsements showing, Bank as an additional insured. Bank shall be named as lender loss payee and/or additional insured with respect to any such insurance providing coverage in respect of any Collateral.
- (b) Ensure that proceeds payable under any property policy are, at Bank's option, payable to Bank on account of the Obligations. Notwithstanding the foregoing, (a) so long as no Event of Default has occurred and is continuing, Borrower shall have the option of applying the proceeds of property and casualty policies up toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property (i) shall be of equal or like value as the replaced or repaired Collateral, (ii) shall be made within 365 days of date of receipt of such proceeds, and (iii) shall be deemed Collateral in which Bank has been granted a first priority security interest (subject to Permitted Liens), and (b) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Bank, be payable to Bank on account of the Obligations.
- (c) At Bank's request, Borrower shall deliver certified copies of insurance policies and evidence of all premium payments. Each provider of any such insurance required under this Section 6.7 shall agree, by endorsement upon the policy or policies issued by it or by independent instruments furnished to Bank, that it will give Bank thirty (30) days prior written notice before any such policy or policies shall be canceled. If Borrower fails to obtain insurance as required under this Section 6.7 or to pay any amount or furnish any required proof of payment to third persons and Bank, Bank may make all or part of such payment or obtain such insurance policies required in this Section 6.7, and take any action under the policies Bank deems prudent.
- **6.8 Accounts.** Borrower shall provide Bank five (5) days prior written notice before establishing any Collateral Account at or with any bank or financial institution other than Bank or Bank's Affiliates. For each Collateral Account that Borrower at any time maintains, Borrower shall cause the applicable bank or financial institution (other than Bank) at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Bank's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Bank. The provisions of this Section 6.8 shall not apply to the Excluded Accounts; provided that Borrower shall provide Bank with written notice promptly (and in any event within five (5) days) upon the time that any of Borrower's accounts shall no longer constitute an Excluded Account as defined herein.
- **6.9 Minimum LTM Consolidated EBITDA**. Maintain, to be tested as of the last day of each quarter and calculated on a trailing 12-month basis, Consolidated EBITDA of at least the amounts shown in the table immediately below for the corresponding reporting periods (the "Financial Covenant"):

Period Ending	Minimum LTM Consolidated EBITDA
LTM 4Q'24	\$ (***)
LTM 1Q'25	\$ (***)
LTM 2Q'25	\$ (***)
LTM 3Q'25	\$ (***)
LTM 4Q'25	\$ (***)
LTM 1Q'26	\$ (***)
LTM 2Q'26	\$ (***)
LTM 3Q'26	\$ (***)
LTM 4Q'26	\$ (***)
LTM 1Q'27	\$ (***)
LTM 2Q'27	\$ (***)
LTM 3Q'27	\$ (***)

### 6.10 Protection of Intellectual Property Rights.

- (a) (i) Protect, defend and maintain the validity and enforceability of Borrower's and each Subsidiary's Intellectual Property material to Borrower's business or operations; (ii) promptly advise Bank in writing of material infringements or any other event that could reasonably be expected to materially and adversely affect the value of Borrower's and each Subsidiary's Intellectual Property; and (iii) not allow any Intellectual Property material to Borrower's business or operations to be abandoned, forfeited or dedicated to the public without Bank's written consent.
- (b) Provide written notice to Bank, concurrently with the required delivery of a Compliance Statement pursuant to Section 6.2, of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). Borrower shall take such commercially reasonable steps as Bank requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (i) any Restricted License to be deemed "Collateral" and for Bank to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (ii) Bank to have the ability in the event of a liquidation of any Collateral to dispose of such Collateral in accordance with Bank's rights and remedies under this Agreement and the other Loan Documents.
- **6.11 Litigation Cooperation**. From the date hereof and continuing through the termination of this Agreement, make available to Bank, without expense to Bank, Borrower and its officers, employees and agents and Borrower's books and records, to the extent that Bank may deem them reasonably necessary to prosecute or defend any third-party suit or proceeding instituted by or against Bank with respect to any Collateral or relating to Borrower.

### 6.12 Reserved.

**6.13 Formation or Acquisition of Subsidiaries**. Notwithstanding and without limiting the negative covenants contained in Sections 7.3 and 7.7 hereof, at the time that Borrower or any Guarantor forms any direct or indirect Subsidiary or acquires any direct or indirect Subsidiary after the Effective Date (including, without limitation, pursuant to a Division), Borrower and such Guarantor shall, within forty-five (45) days of such formation or acquisition, (a) cause such new Subsidiary (other than a Foreign

Subsidiary) to provide to Bank a joinder to this Agreement to become a co-borrower hereunder or a Guaranty to become a Guarantor hereunder, together with such appropriate financing statements and/or Control Agreements, all in form and substance reasonably satisfactory to Bank (including being sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Subsidiary), (b) provide to Bank appropriate certificates and powers and financing statements, pledging all of the direct or beneficial ownership interest in such new Subsidiary, in form and substance reasonably satisfactory to Bank, provided, however, that such pledge shall include no more than sixty-five percent (65%) of the presently existing and hereafter issued voting stock of any Foreign Subsidiary (as determined under U.S. federal tax principles); and (c) provide to Bank all other documentation in form and substance satisfactory to Bank, which in its opinion is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 6.13 shall be a Loan Document.

- **6.14 Use of Proceeds**. Cause the proceeds of the Credit Extensions to be used solely (a) as working capital or (b) to fund its general business purposes, and not for personal, family, household or agricultural purposes.
- **6.15 Further Assurances**. Execute any further instruments and take further action as Bank reasonably requests to perfect, protect, ensure the priority of or continue Bank's Lien in the Collateral or to effect the purposes of this Agreement. Upon Bank's reasonable request, deliver to Bank, within five (5) days after the same are sent or received, copies of all correspondence, reports, documents and other filings with any Governmental Authority regarding compliance with or maintenance of Governmental Approvals or Requirements of Law or that could reasonably be expected to have a material effect on any of the Governmental Approvals or otherwise on the operations of Borrower or any of its Subsidiaries.

# **6.16 Post-Closing Obligations**.

- (a) On or prior to the date that is sixty (60) days after the Closing Date, the Borrower shall have delivered to Bank a Control Agreement with respect to all Collateral Accounts as required by Section 6.8, in form and substance satisfactory to Agent, duly authorized, executed and delivered by the Borrower, Agent and the bank at which such Collateral Account is maintained.
- (b) On or prior to the date that is thirty (30) days after the Closing Date, the Borrower shall deliver to the Bank insurance endorsements, in form and substance reasonably satisfactory to Bank, with respect to additional insured, lenders loss payee and notice endorsements relating to the insurance policies required by Section 6.7 hereof.

## 7 <u>NEGATIVE COVENANTS</u>

Borrower shall not do any of the following without Bank's prior written consent:

**7.1 Dispositions**. Convey, sell, lease, transfer, assign, or otherwise dispose of (including, without limitation, pursuant to a Division) (collectively, "**Transfer**"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out, surplus or obsolete Equipment that is, in the reasonable judgment of Borrower, no longer economically practicable to maintain or useful in the ordinary course of business of Borrower; (c) consisting of Permitted Liens and Permitted Investments; (d) consisting of the sale or issuance of any stock of Borrower permitted under Section 7.2 of this Agreement; (e) consisting of Borrower's use or transfer of money or Cash Equivalents in the ordinary course of its business for the payment of ordinary

course business expenses in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents; (f) of non-exclusive licenses for the use of the property of Borrower or its Subsidiaries in the ordinary course of business and licenses that could not result in a legal transfer of title of the licensed property but that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States; (g) of accounts receivable and other claims which arise out of the sale of goods or services to United Parcel Service, Inc., a Delaware corporation ("UPS"), and/or its Subsidiaries or Affiliates, to JPMorgan Chase Bank, N.A. ("JPMorgan") and/or one (1) or more other investors, pursuant to the terms of a Master Receivables Purchase Acceptance Letter by and between Borrower and JPMorgan or to any other financial institution pursuant to any similar arrangement; or (h) not to exceed Two Million Five Hundred Thousand Dollars (\$2,500,000) in the aggregate in any fiscal year.

- **7.2 Changes in Business, Management, Control, or Business Locations.** (a) Engage in or permit any of its Subsidiaries to engage in any business other than the businesses currently engaged in by Borrower and such Subsidiary, as applicable, or reasonably related thereto; (b) liquidate or dissolve; (c) [reserved]; or (d) permit, allow or suffer any Change in Control. Borrower shall not, without at least thirty (30) days prior written notice to Bank: (1) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Two Million Five Hundred Thousand Dollars (\$2,500,000) in Borrower's assets or property) or deliver any portion of the Collateral (other than movable items of personal property such as laptop computers) valued, individually or in the aggregate, in excess of Two Million Five Hundred Thousand Dollars (\$2,500,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (2) change its jurisdiction of organization, (3) change its organizational type, (4) change its legal name, or (5) change any organizational number (if any) assigned by its jurisdiction of organization.
- **7.3 Mergers or Acquisitions**. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock, partnership, membership, or other ownership interest or other equity securities or property of another Person (including, without limitation, by the formation of any Subsidiary or pursuant to a Division), except for Permitted Acquisitions; provided that a Subsidiary may merge or consolidate into another Subsidiary or into Borrower.
- **7.4 Indebtedness**. Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness.
- **7.5 Encumbrance.** (a) Create, incur, allow, or suffer to exist any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except for Permitted Liens, (b) permit any Collateral not to be subject to the first priority security interest granted herein, or (c) enter into any agreement, document, instrument or other arrangement (except with or in favor of Bank) with any Person which directly or indirectly prohibits or has the effect of prohibiting Borrower or any Subsidiary from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Borrower's or any Subsidiary's Intellectual Property in favor of Bank, except as is otherwise permitted in Section 7.1 hereof and the definition of "Permitted Liens" herein.

7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.8 hereof.

- 7.7 Distributions; Investments. (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock, partnership, membership, or other ownership interest or other equity securities provided that Borrower may (i) convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof and pay cash in lieu of fractional shares in connection with such conversion, (ii) pay dividends or make distributions solely in common stock; (iii) repurchase stock pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of any such repurchase and would not exist after giving effect to any such repurchase, provided that the aggregate amount of all such repurchases does not exceed Two Million Five Hundred Thousand Dollars (\$2,500,000) per fiscal year and (iv) any Subsidiary of Borrower may make dividends or distributions ratably on account of its capital stock; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than Permitted Investments, or permit any of its Subsidiaries to do so.
- **7.8 Transactions with Affiliates**. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of Borrower, except for (a) transactions that are in the ordinary course of Borrower's business, upon fair and reasonable terms that are no less favorable to Borrower than would be obtained in an arm's length transaction with a non-affiliated Person, (b) transactions that are not otherwise prohibited by this Section 7, (c) transactions permitted by Section 7.7 hereof, (d) commercially reasonable and customary compensation arrangements with Borrower's employees, officers, directors and managers and commercially reasonable and customary indemnification arrangements with Borrower's directors and managers, in each case, approved by the Board, (e) the incurrence of Subordinated Debt, or (f) sales of equity securities in a bona fide financing transactions.
- **7.9 Subordinated Debt**. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof, provide for earlier or greater principal, interest, or other payments thereon, or adversely affect the subordination thereof to Obligations owed to Bank.
- **7.10 Compliance**. Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if such violation, individually or in the aggregate, could reasonably be expected to have a material adverse effect on Borrower's business or operations, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other Governmental Authority.
- **7.11 Negative Pledge of Intellectual Property**. Except for Permitted Liens, assign, pledge, grant a security interest in or upon, or encumber any of Borrower's or any Subsidiary's Intellectual Property existing now or in the future, in favor of any third party, without receiving Bank's prior written consent.

### **8** EVENTS OF DEFAULT

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

**8.1 Payment Default**. Borrower fails to (a) make any payment of principal or interest on any Credit Extension when due, or (b) pay any other Obligations within three (3) Business Days after such Obligations are due and payable (which three (3) Business Day cure period shall not apply to payments due on the Revolving Line Maturity Date). During the cure period, the failure to make or pay any payment specified under clause (b) hereunder is not an Event of Default (but no Credit Extension will be made during the cure period);

### 8.2 Covenant Default.

(a) Borrower fails or neglects to perform any obligation in Sections 6.2, 6.3, 6.4, 6.5, 6.6, 6.7, 6.8, 6.9, 6.10, 6.12, 6.13, 6.14, or 6.15 or violates any covenant in Section 7; or (b) Borrower fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement or any Loan Documents, and as to any default (other than those specified in this Section 8) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure the default within ten (10) Business Days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) Business Day period or cannot after diligent attempts by Borrower be cured within such ten (10) Business Day period, and such default is likely to be cured within a reasonable time, then Borrower shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period). Cure periods provided under this section shall not apply, among other things, to financial covenants or any other covenants set forth in clause (a) above;

### 8.3 [Reserved].

### 8.4 Attachment; Levy; Restraint on Business.

- (a) (i) The service of process seeking to attach, by trustee or similar process, any funds of Borrower or of any entity under the control of Borrower (including a Subsidiary) in excess of Two Million Five Hundred Thousand Dollars (\$2,500,000), or (ii) a notice of lien or levy is filed against any of Borrower's assets by any Governmental Authority, and the same under subclauses (i) and (ii) hereof are not, within ten (10) days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); provided, however, no Credit Extensions shall be made during any ten (10) day cure period; or
- (b) (i) any material portion of Borrower's assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents Borrower from conducting all or any material part of its business, provided, however, that the Event of Default under this Section 8.4(b) shall be cured or waived for purposes of this Agreement upon Bank receiving written evidence that the same under subclauses (i) and (ii) hereof have, within ten (10) days after the occurrence thereof, been discharged or stayed (whether through the posting of a bond or otherwise) and so long as Bank has not declared an Event of Default under any other provision of this Agreement and/or exercised any rights with respect thereto;

- **8.5 Insolvency**. (a) Borrower or any of its Subsidiaries is unable to pay its debts (including trade debts) as they become due or otherwise becomes insolvent; (b) Borrower or any of its Subsidiaries begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against Borrower or any of its Subsidiaries and is not dismissed or stayed within forty-five (45) days (but no Credit Extensions shall be made while any of the conditions described in clause (a) exist and/or until any Insolvency Proceeding is dismissed);
- **8.6 Other Agreements**. There is, under any agreement to which Borrower or any Guarantor is a party with a third party or parties, any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of Two Million Five Hundred Thousand Dollars (\$2,500,000);
- **8.7 Judgments; Penalties.** One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least Two Million Five Hundred Thousand Dollars (\$2,500,000) (not covered by independent third-party insurance as to which liability has been accepted by such insurance carrier) shall be rendered against Borrower by any Governmental Authority, and the same are not, within ten (10) days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay (provided that no Credit Extensions will be made prior to the satisfaction, payment, discharge, stay, or bonding of such fine, penalty, judgment, order or decree);
- **8.8 Misrepresentations.** Borrower or any Person acting for Borrower makes any representation, warranty, or other statement now or later in this Agreement, any Loan Document or in any writing delivered to Bank or to induce Bank to enter this Agreement or any Loan Document, and such representation, warranty, or other statement is incorrect in any material respect when made;
- **8.9 Subordinated Debt**. If (a) any document, instrument, or agreement evidencing any Subordinated Debt shall for any reason be revoked or invalidated or otherwise cease to be in full force and effect, any Person shall be in breach thereof or contest in any manner the validity or enforceability thereof or deny that it has any further liability or obligation thereunder; (b) a default or event of default (however defined) has occurred under any document, instrument, or agreement evidencing any Subordinated Debt, which default shall not have been cured or waived within any applicable grace period; or (c) the Obligations shall for any reason be subordinated or shall not have the priority contemplated by this Agreement or any applicable subordination or intercreditor agreement;
- **8.10 Lien Priority**. There is a material impairment (through no fault of Bank) in the perfection or priority of Bank's security interest in the Collateral:
- **8.11 Guaranty**. (a) Any guaranty of any Obligations terminates or ceases for any reason to be in full force and effect; (b) any Guarantor does not perform any obligation or covenant under any guaranty of the Obligations; (c) any circumstance described in Sections 8.4, 8.5, 8.6, 8.7, or 8.8 of this Agreement occurs with respect to any Guarantor, (d) the death, liquidation, winding up, or termination of existence of any Guarantor; or (e) a material impairment (through no fault of Bank) in the perfection or priority of Bank's Lien in the collateral provided by Guarantor; or
- **8.12 Governmental Approvals**. Any Governmental Approval shall have been revoked, rescinded, suspended, modified in an adverse manner or not renewed that could reasonably be expected to cause a Material Adverse Change.

### 9 BANK'S RIGHTS AND REMEDIES

- **9.1 Rights and Remedies**. Upon the occurrence and during the continuance of an Event of Default, Bank may, without notice or demand, do any or all of the following:
- (a) declare all Obligations immediately due and payable (but if an Event of Default described in Section 8.5 occurs all Obligations are immediately due and payable without any action by Bank);
- (b) stop advancing money or extending credit for Borrower's benefit under this Agreement or under any other agreement between Borrower and Bank;
- (c) demand that Borrower (i) deposit cash with Bank in an amount equal to at least (A) one hundred five percent (105.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in Dollars remaining undrawn, and (B) one hundred ten percent (110.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in a Foreign Currency remaining undrawn (plus, in each case, all interest, fees, and costs due or to become due in connection therewith (as estimated by Bank in its good faith business judgment)), to secure all of the Obligations relating to such Letters of Credit, as collateral security for the repayment of any future drawings under such Letters of Credit, and Borrower shall forthwith deposit and pay such amounts, and (ii) pay in advance all letter of credit fees scheduled to be paid or payable over the remaining term of any Letters of Credit;
- (d) terminate any FX Contracts (it being understood and agreed that (i) Bank is not obligated to deliver the currency which Borrower has contracted to receive under any FX Contract, and Bank may cover its exposure for any FX Contracts by purchasing or selling currency in the interbank market as Bank deems appropriate; (ii) Borrower shall be liable for all losses, damages, costs, margin obligations and expenses incurred by Bank arising from Borrower's failure to satisfy its obligations under any FX Contract or the execution of any FX Contract; and (iii) Bank shall not be liable to Borrower for any gain in value of a FX Contract that Bank may obtain in covering Borrower's breach);
- (e) verify the amount of, demand payment of and performance under, and collect any Accounts and General Intangibles, settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that Bank considers advisable, and notify any Person owing Borrower money of Bank's security interest in such funds. Borrower shall collect all payments in trust for Bank and, if requested by Bank, immediately deliver the payments to Bank in the form received from the Account Debtor, with proper endorsements for deposit;
- (f) make any payments and do any acts it considers necessary or reasonable to protect the Collateral and/or its security interest in the Collateral. Borrower shall assemble the Collateral if Bank requests and make it available as Bank designates. Bank may enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. Borrower grants Bank a license to enter and occupy any of its premises, without charge, to exercise any of Bank's rights or remedies;
- (g) apply to the Obligations any (i) balances and deposits of Borrower it holds, or (ii) amount held by Bank owing to or for the credit or the account of Borrower;
- (h) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. Bank is hereby granted a non-exclusive, royalty-free license or other right to use,

without charge, Borrower's labels, Patents, Copyrights, mask works, rights of use of any name, trade secrets, trade names, Trademarks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with Bank's exercise of its rights under this Section 9.1, Borrower's rights under all licenses and all franchise agreements inure to Bank's benefit;

- (i) place a "hold" on any account maintained with Bank and/or deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral;
- (j) direct Borrower to deliver, in kind, all proceeds arising from the disposition of any Collateral to Bank in the original form in which received by Borrower not later than the following Business Day after receipt by Borrower, to be applied to the Obligations, pursuant to the terms of Section 9.4 hereof. Borrower agrees that it will not commingle proceeds of Collateral with any of Borrower's other funds or property, but will hold such proceeds separate and apart from such other funds and property and in an express trust for Bank;
  - (k) demand and receive possession of Borrower's Books; and
- (l) exercise all rights and remedies available to Bank under the Loan Documents or at law or equity, including all remedies provided under the Code or any Applicable Law (including disposal of the Collateral pursuant to the terms thereof).
- 9.2 Power of Attorney. Borrower hereby irrevocably appoints Bank as its true and lawful attorney-in-fact, exercisable following the occurrence and during the continuance of an Event of Default, to: (a) endorse Borrower's name on any checks, payment instruments, or other forms of payment or security; (b) sign Borrower's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) demand, collect, sue, and give releases to any Account Debtor for monies due, settle and adjust disputes and claims about the Accounts directly with Account Debtors, and compromise, prosecute, or defend any action, claim, case, or proceeding about any Collateral (including filing a claim or voting a claim in any bankruptcy case in Bank's or Borrower's name, as Bank chooses); (d) make, settle, and adjust all claims under Borrower's insurance policies; (e) pay, contest or settle any Lien, charge, encumbrance, security interest, or other claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of Bank or a third party as the Code permits. Borrower hereby appoints Bank as its lawful attorney-in-fact to sign Borrower's name on any documents necessary to perfect or continue the perfection of Bank's security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity obligations) have been satisfied in full and the Loan Documents have been terminated. Bank's foregoing appointment as Borrower's attorney in fact, and all of Bank's rights and powers, coupled with an interest, are irrevocable until all Obligations (other than inchoate indemnity obligations) have been satisfied in full, Bank is under no further obligation to make Credit Extensions and the Loan Documents have been terminated. Bank shall not incur any liability in connection with or arising from the exercise of such power of attorney and shall have no obligation to
- **9.3 Protective Payments**. If Borrower fails to obtain the insurance called for by Section 6.7 or fails to pay any premium thereon or fails to pay any other amount which Borrower is obligated to pay under this Agreement or any other Loan Document or which may be required to preserve the Collateral, Bank may obtain such insurance or make such payment, and all amounts so paid by Bank are Bank Expenses

and immediately due and payable, bearing interest at the then highest rate applicable to the Obligations, and secured by the Collateral. Bank will make reasonable efforts to provide Borrower with notice of Bank obtaining such insurance at the time it is obtained or within a reasonable time thereafter. No payments by Bank are deemed an agreement to make similar payments in the future or Bank's waiver of any Event of Default.

- **9.4 Application of Payments and Proceeds**. Bank shall have the right to apply in any order any funds in its possession, whether from Borrower account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations. Bank shall pay any surplus to Borrower by credit to any of Borrower's deposit accounts or to other Persons legally entitled thereto; Borrower shall remain liable to Bank for any deficiency. If Bank, directly or indirectly, enters into a deferred payment or other credit transaction with any purchaser at any sale of Collateral, Bank shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reduction of the Obligations until the actual receipt by Bank of cash therefor.
- 9.5 Bank's Liability for Collateral. So long as Bank complies with reasonable banking practices regarding the safekeeping of the Collateral in the possession or under the control of Bank, Bank shall not be liable or responsible for: (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warehouseman, bailee, or other Person. Borrower bears all risk of loss, damage or destruction of the Collateral.
- 9.6 No Waiver; Remedies Cumulative. Bank's failure, at any time or times, to require strict performance by Borrower of any provision of this Agreement or any other Loan Document shall not waive, affect, or diminish any right of Bank thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by the party granting the waiver and then is only effective for the specific instance and purpose for which it is given. Bank's rights and remedies under this Agreement and the other Loan Documents are cumulative. Bank has all rights and remedies provided under the Code, by law, or in equity. Bank's exercise of one right or remedy is not an election and shall not preclude Bank from exercising any other remedy under this Agreement or other remedy available at law or in equity, and Bank's waiver of any Event of Default is not a continuing waiver. Bank's delay in exercising any remedy is not a waiver, election, or acquiescence.
- **9.7 Demand Waiver**. Borrower waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Bank on which Borrower is liable.

# 10 NOTICES

All notices, consents, requests, approvals, demands, or other communication by any party to this Agreement or any other Loan Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by electronic mail or facsimile transmission; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address, facsimile number, or email address indicated below; provided that, for clause (b), if such notice, consent, request, approval, demand or other communication is not sent during the normal

business hours of the recipient, it shall be deemed to have been sent at the opening of business on the next Business Day of the recipient. Bank or Borrower may change its mailing or electronic mail address or facsimile number by giving the other party written notice thereof in accordance with the terms of this Section 10.

If to Borrower: COUCHBASE, INC.

[\*\*\*]
Attn: [\*\*\*]
Email: [\*\*\*]

If to Bank: MUFG BANK, LTD.

[\*\*\*]
Attn: [\*\*\*]
Fax: [\*\*\*]
Email: [\*\*\*]
with a copy to: Lowenstein Sandler LLP
[\*\*\*]
Attn: [\*\*\*]
Fax: [\*\*\*]
Email: [\*\*\*]
Email: [\*\*\*]

### 11 CHOICE OF LAW, VENUE, JURY TRIAL WAIVER AND JUDICIAL REFERENCE

Except as otherwise expressly provided in any of the Loan Documents, New York law governs the Loan Documents without regard to principles of conflicts of law that would require the application of the laws of another jurisdiction. Borrower and Bank each irrevocably and unconditionally submit to the exclusive jurisdiction of the State and Federal courts in New York, New York; provided, however, that nothing in this Agreement shall be deemed to operate to preclude Bank from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Bank. Borrower expressly, irrevocably and unconditionally, submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and Borrower hereby irrevocably and unconditionally waives, to the fullest extent permitted by Applicable Law, any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby irrevocably and unconditionally consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Borrower hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Borrower at the address set forth in, or subsequently provided by Borrower in accordance with, Section 10 of this Agreement and that service so made shall be deemed completed upon the earlier to occur of Borrower's actual receipt thereof or three (3) Business Days after deposit in the U.S. mails, proper postage prepaid.

TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND BANK EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

This Section 11 shall survive the termination of this Agreement.

### 12 GENERAL PROVISIONS

12.1 Termination Prior to Maturity Date; Survival. All covenants, representations and warranties made in this Agreement shall continue in full force until this Agreement has terminated pursuant to its terms and all Obligations have been satisfied. So long as Borrower has satisfied the Obligations (other than inchoate indemnity obligations, and any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement), this Agreement may be terminated prior to the Revolving Line Maturity Date by Borrower, effective three (3) Business Days after written notice of termination is given to Bank. Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination and the repayment of all Obligations.

12.2 Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. Borrower may not assign or transfer this Agreement or any rights or obligations under it without Bank's prior written consent (which may be granted or withheld in Bank's sole discretion) and any other attempted assignment or transfer by Borrower shall be null and void. Bank has the right, without the consent of or notice to Borrower, to sell, transfer, assign, negotiate, or grant participation in all or any part of, or any interest in, Bank's obligations, rights, and benefits under this Agreement and the other Loan Documents. Bank or any other lender hereunder may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Revolver Commitments and the Advances at the time owing to it). Notwithstanding the foregoing, if no Event of Default exists, Bank shall not be permitted to assign its interests in the Loan Documents to a direct competitor of Borrower, vulture fund or distressed debt fund, each as determined by Bank. The parties to each assignment shall execute and deliver to Bank an assignment and assumption agreement in a form reasonably satisfactory to Bank. The assignee shall deliver to Bank any such administrative information as Bank may request. Bank shall maintain at one of its offices a copy of each assignment and a register for the recordation of the names and addresses of the lenders, and the Advances of, and principal amounts (and stated interest) owing to, each lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error, and Borrower, Bank or any other lender shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by Borrower, at any reasonable time and from time to time upon reasonable prior notice. Further, each lender (including Bank) that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of Borrower, maintain a register on which it enters the name and address of each participant and the principal amounts (and stated interest) of each participant's interest in Bank's obligations, rights, and benefits under this Agreement and the other Loan Documents (the "Participant Register"); provided that no lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any participant or any information relating to a participant's interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the Treasury Regulations (or successor provisions of similar effect). The foregoing language is intended to cause the Advances, Loan or other obligations under any Loan Document, and any assignments and participation thereof, to be in "registered form" as defined in Sections 163(f), 871(h)(2) and 881(c)(2) of the U.S. Code and any related Treasury Regulations (or any other relevant or successor provisions of the U.S. Code or such Treasury Regulations). The entries in the Participant Register shall be conclusive absent manifest

error, and such lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

#### 12. 3 Indemnification.

- (a) General Indemnification. Borrower shall indemnify, defend and hold Bank and its Affiliates and the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors, attorneys, or any other Person affiliated with or representing Bank and its Affiliates (each, an "Indemnified Person") harmless against: all obligations, demands, losses, claims, damages, liabilities and related documented expenses (including Bank Expenses and the reasonable fees, charges and disbursements of any counsel for any Indemnified Person) (collectively, "Claims") arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereunder or the consummation of the transactions contemplated hereby or thereby, (ii) any Credit Extension or the use or proposed use of the proceeds therefrom, (iii) any actual or alleged presence or release of hazardous materials on or from any property owned or operated by Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by Borrower, and regardless of whether any Indemnified Person is a party thereto; provided that such indemnity shall not, as to any Indemnified Person, be available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnified Person. All amounts due under this Section 12.3 shall be payable promptly after demand therefor.
- (b) <u>Waiver of Consequential Damages</u>, <u>Etc.</u> To the fullest extent permitted by Applicable Law, Borrower shall not assert, and hereby waives, any claim against any Indemnified Person, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) or any loss of profits arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Credit Extension, or the use of the proceeds thereof. No Indemnified Person shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby except to the extent such damages are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnified Person.
- (c) This Section 12.3 shall survive the termination of this Agreement and the repayment of all Obligations until all statutes of limitation with respect to the Claims, losses, and expenses for which indemnity is given shall have run. This Section 12.3 shall not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, etc. arising from any non-Tax claim.
  - **12.4 Time of Essence**. Time is of the essence for the performance of all Obligations in this Agreement.
- **12.5 Severability of Provisions**. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.

- **12.6 Correction of Loan Documents**. Bank may correct patent errors and fill in any blanks in the Loan Documents consistent with the agreement of the parties so long as Bank provides Borrower with written notice of such correction and allows Borrower at least ten (10) Business Days to object to such correction. In the event of such objection, such correction shall not be made except by an amendment signed by both Bank and Borrower.
- 12.7 Amendments in Writing; Waiver; Integration. No purported amendment or modification of any Loan Document, or waiver, discharge or termination of any obligation under any Loan Document, shall be enforceable or admissible unless, and only to the extent, expressly set forth in a writing signed by the party against which enforcement or admission is sought. Without limiting the generality of the foregoing, no oral promise or statement, nor any action, inaction, delay, failure to require performance or course of conduct shall operate as, or evidence, an amendment, supplement or waiver or have any other effect on any Loan Document. Any waiver granted shall be limited to the specific circumstance expressly described in it, and shall not apply to any subsequent or other circumstance, whether similar or dissimilar, or give rise to, or evidence, any obligation or commitment to grant any further waiver. The Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of the Loan Documents merge into the Loan Documents.
- **12.8 Counterparts**. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement. Delivery of an executed signature page of this Agreement by electronic mail transmission shall be effective as delivery of a manually executed counterpart hereof.
- 12.9 Confidentiality. Bank agrees to maintain the confidentiality of Information (as defined below), except that Information may be disclosed (a) to Bank's Subsidiaries and Affiliates and their respective employees, directors, agents, attorneys, accountants and other professional advisors (collectively, "Representatives" and, together with Bank, collectively, "Bank Entities"); (b) to prospective transferees, assignees, credit providers or purchasers of Bank's interests under or in connection with this Agreement and their Representatives (provided, however, Bank shall use commercially reasonable efforts to obtain any such prospective transferee's, assignee's, credit provider's, purchaser's or their Representatives' agreement to the terms of this provision); (c) as required by law, regulation, subpoena, or other order; (d) to Bank's regulators or as otherwise required or requested in connection with Bank's examination or audit; (e) in connection with the exercise of remedies under the Loan Documents or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder; and (f) to third-party service providers of Bank so long as such service providers have executed a confidentiality agreement with Bank with terms no less restrictive than those contained herein. "Information" means all information received from Borrower regarding Borrower or its business, in each case other than information that is either: (i) in the public domain or in Bank's possession when disclosed to Bank, or becomes part of the public domain (other than as a result of its disclosure by Bank in violation of this Agreement) after disclosure to Bank; or (ii) disclosed to Bank by a third party, if Bank does not know that the third party is prohibited from disclosing the information.

Bank Entities may use anonymous forms of confidential information for aggregate datasets, for analyses or reporting, and for any other uses not expressly prohibited in writing by Borrower. The provisions of the immediately preceding sentence shall survive the termination of this Agreement.

**12.10 Attorneys' Fees, Costs and Expenses**. In any action or proceeding between Borrower and Bank arising out of or relating to the Loan Documents, the prevailing party shall be entitled to recover its

reasonable attorneys' fees and other costs and expenses incurred, in addition to any other relief to which it may be entitled.

- 12.11 Electronic Execution of Documents. The words "execution," "signed," "signature" and words of like import in any Loan Document shall be deemed to include electronic signatures, including any Electronic Signature as defined in the Electronic Transactions Law (2003 Revision) of the Cayman Islands (the "Cayman Islands Electronic Signature Law"), if applicable, or the keeping of records in electronic form, including any Electronic Record, as defined in Cayman Islands Electronic Signature Law, each of which shall be of the same legal effect, validity and enforceability as a manually executed signature or the use of a paper-based recordkeeping systems, as the case may be, to the extent and as provided for in any Applicable Law, including, without limitation, any state law based on the Uniform Electronic Transactions Act or the Cayman Islands Electronic Signature Law; provided, however that sections 8 and 19(3) of the Cayman Islands Electronic Signature Law shall not apply to this Agreement or the execution or delivery thereof.
- 12.12 Right of Setoff. Borrower hereby grants to Bank a Lien and a right of setoff as security for all Obligations to Bank, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Bank or any entity under the control of Bank (including a subsidiary of Bank) or in transit to any of them and other obligations owing to Bank. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Bank may setoff the same or any part thereof and apply the same to any liability or Obligation of Borrower even though unmatured and regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE BANK TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER, ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.
- **12.13 Captions and Section References**. The headings used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement. Unless indicated otherwise, section references herein are to sections of this Agreement.
- **12.14 Construction of Agreement**. The parties hereto mutually acknowledge that they and their attorneys have participated in the preparation and negotiation of this Agreement. In cases of uncertainty this Agreement shall be construed without regard to which of the parties caused the uncertainty to exist.
- **12.15 Relationship**. The relationship of the parties to this Agreement is determined solely by the provisions of this Agreement. The parties do not intend to create any agency, partnership, joint venture, trust, fiduciary or other relationship with duties or incidents different from those of parties to an arm's-length contract.
- **12.16 Third Parties**. Nothing in this Agreement, whether express or implied, is intended to: (a) confer any benefits, rights or remedies under or by reason of this Agreement on any persons other than the express parties to it and their respective permitted successors and assigns; (b) relieve or discharge the obligation or liability of any person not an express party to this Agreement; or (c) give any person not an express party to this Agreement any right of subrogation or action against any party to this Agreement.
- 12.17 Anti-Terrorism Law. Bank hereby notifies Borrower that, pursuant to the requirements of Anti-Terrorism Law, Bank may be required to obtain, verify and record information that identifies

Borrower, which information may include the name and address of Borrower and other information that will allow Bank to identify Borrower in accordance with Anti-Terrorism Law. Borrower hereby agrees to take any action necessary to enable Bank to comply with the requirements of Anti-Terrorism Law.

12.18 Marketing Consent. Borrower hereby authorizes Bank Entities, without any prior approval by Borrower, to include Borrower's name and logo in advertising, marketing, tombstones, case studies and training materials, and give such other publicity to this Agreement as Bank Entities may from time to time determine in their reasonable discretion; provided that notwithstanding the foregoing, no such advertising, marketing, tombstones, case studies, training materials, or other publicity may disclose any information that is restricted from being disclosed pursuant to the Loan Documents or that has not previously been disclosed by Borrower in an 8-K or similar public filing. The foregoing authorization shall remain in effect unless Borrower notifies Bank in writing that such authorization is revoked.

### 13 <u>DEFINITIONS</u>

- 13.1 Definitions. As used in the Loan Documents, (i) the words "shall" or "will" are mandatory, the word "may" is permissive, the word "or" is not exclusive, the words "includes" and "including" are not limiting, the singular includes the plural, and numbers denoting amounts that are set off in brackets are negative, (ii) the term "continuing" in the context of an Event of Default means that the Event of Default has not been remedied (if capable of being remedied) or waived; and (iii) whenever a representation or warranty is made to Borrower's knowledge or awareness, to the "best of" Borrower's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer. Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth in this Section 13.1. All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein. As used in this Agreement, the following capitalized terms have the following meanings:
- "Account" is, as to any Person, any "account" of such Person as "account" is defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to such Person.
  - "Account Debtor" is any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.
  - "Administrator" is an individual that is named as an Authorized Signer of Borrower in an approval by the Board.
  - "Advance" or "Advances" means a revolving credit loan (or revolving credit loans) under the Revolving Line.
- "Affiliate" is, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members. For purposes of the definition of Eligible Accounts, Affiliate shall include a Specified Affiliate.
  - "Agreement" is defined in the preamble hereof.
- "Anti-Terrorism Law" means any law relating to terrorism or money-laundering, including Executive Order No. 13224 and the USA Patriot Act.

- "Applicable Law" means, as to any person, all applicable Laws binding upon such person or to which such a person is subject.
- "ASU" is defined in Section 1.
- "Authorized Signer" is any individual listed in Borrower's Borrowing Resolution who is authorized to execute the Loan Documents, including making (and executing if applicable) any Credit Extension request, on behalf of Borrower.
- "Availability Amount" is (a) the Revolving Line minus (b) the outstanding principal balance of any Advances minus (c) the aggregate face amount of Letters of Credit issued under the Letter of Credit Sublimit.
- "Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 2.13(d).
  - "Bank" is defined in the preamble hereof.
  - "Bank Entities" is defined in Section 12.9.
- "Bank Expenses" are all audit fees and expenses, costs, and expenses (including reasonable and documented attorneys' fees and expenses) for preparing, amending, negotiating, administering, defending and enforcing the Loan Documents (including, without limitation, those incurred in connection with appeals or Insolvency Proceedings) or otherwise incurred with respect to Borrower or any Guarantor.
- "Bank Services" are any products, credit services, and/or financial accommodations previously, now, or hereafter provided to Borrower or any of its Subsidiaries by Bank or any Bank Affiliate, including, without limitation, any letters of credit, cash management services (including, without limitation, merchant services, direct deposit of payroll, business credit cards, and check cashing services), interest rate swap arrangements, and foreign exchange services as any such products or services may be identified in Bank's various agreements related thereto (each, a "Bank Services Agreement").
  - "Bank Services Agreement" is defined in the definition of Bank Services.
- "Benchmark" means, initially, the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 2.13(a).
- "Benchmark Replacement" means, with respect to any Benchmark Transition Event, the sum of: (a) the alternate benchmark rate that has been selected by Bank and Borrower giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (b) the related Benchmark Replacement Adjustment; provided that, if such Benchmark Replacement as so determined would be less than the

Floor, such Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by Bank and Borrower giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate, all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been or, if such Benchmark is a term rate, all Available Tenors of such Benchmark (or such component thereof) have been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; provided that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if such Benchmark (or such component thereof) or, if such Benchmark is a term rate, any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, if such Benchmark is a term rate, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate, all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate, any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the

Federal Reserve Board, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component thereof) or, if such Benchmark is a term rate, all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate, any Available Tenor of such Benchmark (or such component thereof); or

(c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such Benchmark (or such component thereof) or, if such Benchmark is a term rate, all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

For the avoidance of doubt, if such Benchmark is a term rate, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Start Date" means, in the case of a Benchmark Transition Event, the earlier of (a) the applicable Benchmark Replacement Date and (b) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication of information (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication).

"Benchmark Unavailability Period" means, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 2.13 and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 2.13.

"Board" is Borrower's board of directors or equivalent governing body.

"Borrower" is defined in the preamble hereof.

"Borrower's Books" are all Borrower's books and records including ledgers, federal and state tax returns, records regarding Borrower's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.

"Borrowing Request" means a request for an Advance in the form attached hereto as Exhibit C.

"Borrowing Resolutions" are, with respect to any Person, those resolutions adopted by such Person's board of directors (and, if required under the terms of such Person's Operating Documents, stockholders) and delivered by such Person to Bank approving the Loan Documents to which such Person is a party and the transactions contemplated thereby.

"Business Day" means any day that is not a Saturday, Sunday or other day that is a legal holiday under the laws of the State of New York or is a day on which banking institutions in such state are authorized or required by Law to close.

"Cash Equivalents" means (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and having the highest rating from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc.; (c) Bank's certificates of deposit issued maturing no more than one (1) year after issue; and (d) money market funds at least ninety-five percent (95.0%) of the assets of which constitute Cash Equivalents of the kinds described in clauses (a) through (c) of this definition.

"CFC" means a "controlled foreign corporation" within the meaning of Section 957 of the U.S. Code.

"Change in Control" means (a) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act), directly or indirectly, of forty-nine percent (49.0%) or more of the ordinary voting power for the election of directors of Borrower (determined on a fully diluted basis) other than by the sale of Borrower's equity securities in a public offering or to venture capital or private equity investors so long as Borrower identifies to Bank the venture capital or private equity investors at least seven (7) Business Days prior to the closing of the transaction and provides to Bank a description of the material terms of the transaction; (b) [reserved]; or (c) at any time, Borrower shall cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100.0%) of each class of outstanding capital stock, partnership, membership, or other ownership interest or other equity securities of each Subsidiary of Borrower (other than directors' qualifying shares or other similar shares as required by applicable law) free and clear of all Liens (except Liens created by this Agreement).

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application thereof by any Governmental Authority or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of law) by any Governmental Authority; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted or issued.

"Claims" is defined in Section 12.3.

"Code" is the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the State of New York; provided, that, to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, or priority of, or remedies with respect to. Bank's Lien on any Collateral is governed by the Uniform

Commercial Code in effect in a jurisdiction other than the State of New York, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

- "Collateral" is any and all properties, rights and assets of Borrower described on Exhibit A.
- "Collateral Account" is any Deposit Account, Securities Account or Commodity Account (other than any Excluded Account).
- "Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.
  - "Compliance Statement" is that certain statement in the form attached hereto as Exhibit B.

"Conforming Changes" means, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Prime Rate," the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of Section 2.9 and other technical, administrative or operational matters) that Bank decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by Bank in a manner substantially consistent with market practice (or, if Bank decides that adoption of any portion of such market practice is not administratively feasible or if Bank determines that no market practice for the administration of any such rate exists, in such other manner of administration as Bank decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

- "Consolidated EBITDA": with respect to the Borrower for any period,
- (a) Consolidated Net Income, plus
- (b) the sum, without duplication, of the amounts for such period but (other than with respect to subclauses (xv) and (xvi)) solely to the extent deducted in calculating Consolidated Net Income, for such period of:
  - (i) Consolidated Interest Expense, plus
  - (ii) provisions for taxes based on income, plus
  - (iii) total depreciation expense, plus
  - (iv) total amortization or related impairments, plus

- (v) other non-cash items reducing Consolidated Net Income (excluding any such non-cash item to the extent that it represents an accrual or reserve for potential cash items in any future period or amortization of a prepaid cash item that was paid in a prior period), plus
- (vi) costs and expenses relating to the Loan Documents, plus
- (vii) other extraordinary, unusual or nonrecurring charges or expenses; provided that, for any period, the aggregate amount added back pursuant to this clause (b)(vii) (x) shall not exceed 2,500,000 for such period and (y) when taken together with amounts added pursuant to clauses (b)(xiv), (b)(xv), and (b)(xvi), shall not exceed 7,500,000 for such period, plus
- (viii) non-cash charges for employee compensation plans (including stock option compensation), plus
- (ix) Public Company Costs paid in cash during such period, plus
- (x) any fees, costs, expenses or charges related to any actual, proposed or contemplated issuance of capital stock, Investment, acquisition, disposition outside of the ordinary course of business, recapitalization or the incurrence of Indebtedness permitted to be incurred hereunder (including a refinancing thereof), plus
- (xi) contingent obligations, purchase price adjustments, milestone payments, earn-out payments and indemnity obligations incurred in connection with any Permitted Acquisition, plus
- (xii) any losses, charges or expenses that are (or are expected to be within one year of the end of such period with a deduction in the subsequent period to the extent not so reimbursed or paid) reimbursed or actually paid by a third party or under indemnification or reimbursement provisions, plus
- (xiii) pro forma "run rate" cost savings, operating expense reductions and synergies relating to acquisitions, business combinations, Dispositions, and other cost savings initiatives that are factually supportable, reasonably identifiable and projected in good faith by the Borrower to be realized within 18 months from actions that have been taken or with respect to which substantial steps have been taken or initiated or are expected to be taken (in the good faith determination of the Borrower) within twelve months after the end of such measurement period, in each case, calculated on a pro forma basis as though such cost savings, operating expense reductions and synergies had been realized on the first day of such period and net of the amount of actual benefits realized during such period from such actions; provided that, for any period, the aggregate amount added back pursuant to this clause (b)(xiii), (x) shall not exceed \$2,500,000 for such period and (y) when taken together with amounts added pursuant to clauses (b)(xii), (b)(xiv), and (b)(xv), shall not exceed 7,500,000 for such period, plus
- (xiv) adjustments, exclusions and add-backs set forth in any due diligence adjustments section of a quality of earnings analysis prepared by independent registered public accounts of recognized national standing and delivered to Bank in connection with any Permitted Acquisition or similar permitted Investment; provided that, the aggregate amount added back pursuant to this clause (b)(xiv), (x) shall not exceed \$2,500,000 for such period and (y) when taken together with amounts added pursuant to clauses (b) (vii), (b)(xiii), and (b)(xv), shall not exceed 7,500,000 for such period, plus

- (xv) restructuring, impairments, transitions, business optimization and similar fees, charges and expenses; provided that, for any period, the aggregate amount added back pursuant to this clause (b)(xv), (x) shall not exceed \$2,500,000 and (y) when taken together with amounts added pursuant to clauses (b)(vii), (b)(xiii), and (b)(xiv), shall not exceed 7,500,000 for such period, plus
- (xvi) other adjustments, charges or expenses as approved by Bank, minus
- (c) the sum, without duplication of the amounts for such period of:
  - (i) non-cash items increasing Consolidated Net Income for such period (excluding any such non-cash item to the extent it represents the reversal of an accrual or reserve for potential cash item in any prior period), plus
  - (ii) interest income increasing Consolidated Net Income for such period;

provided that, without duplication of any adjustment set forth above (or of the type set forth above) or other pro forma adjustments pursuant to this Agreement, Consolidated EBITDA for any period shall be determined on a pro forma basis to give effect to any Permitted Acquisitions or any similar permitted Investment or any disposition of any business or assets consummated during such period, in each case, as if such transaction occurred on the first day of such period.

"Consolidated Interest Expense": for any period, total cash interest expense (including that attributable to capital lease obligations) of Borrower for such period with respect to all outstanding Indebtedness of such Persons (including all commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing and net costs under interest rate swap or other funding arrangement agreement to the extent such net costs are allocable to such period in accordance with GAAP).

"Consolidated Net Income": for any period, the consolidated net income (or loss) of Borrower, determined on a consolidated basis in accordance with GAAP; provided that there shall be excluded from the calculation of "Consolidated Net Income":

- (a) the income (or deficit) of any such Person accrued prior to the date it becomes a Subsidiary of Borrower or is merged into or consolidated with a Subsidiary of Borrower,
- (b) the income (or deficit) of any such Person in which Borrower has an ownership interest, except to the extent that any such income is actually received by Borrower in the form of dividends or similar distributions,
- (c) the undistributed earnings of any Subsidiary of the Borrower to the extent that the declaration or payment of dividends or similar distributions by such Subsidiary is not at the time permitted by the terms of any contractual obligation agreement, instrument or other undertaking (other than under any Loan Document) or Applicable Law,
- (d) the cumulative effect of a change in accounting principles and changes as a result of the adoption or modification of accounting policies during such period whether effected through a cumulative effect adjustment or a retroactive application, in each case in accordance with GAAP,

- (e) effects of adjustments related to the application of recapitalization accounting or purchase accounting (including in the inventory, property and equipment, software, goodwill, intangible assets, in process research and development, deferred revenue and debt line items),
- (f) any non-cash gain (loss) attributable to the mark to market movement in the valuation of hedging obligations and other derivative instruments (including swap agreements) and earnouts, and
- (g) any net realized or unrealized gain or loss (after any offset) resulting in such period from currency transaction or translation gains or losses, including those related to currency remeasurements of Indebtedness (including any net loss or gain resulting from (i) hedging obligations for currency exchange risk and (ii) intercompany indebtedness) and any other foreign currency transaction or translation gains and losses.

"Contingent Obligation" is, for any Person, any direct or indirect liability, contingent or not, of that Person for (a) any indebtedness, lease, dividend, letter of credit, credit card or other obligation of another such as an obligation, in each case, directly or indirectly guaranteed, endorsed, co made, discounted or sold with recourse by that Person, or for which that Person is directly or indirectly liable; (b) any obligations for undrawn letters of credit for the account of that Person; and (c) all obligations from any interest rate, currency or commodity swap agreement, interest rate cap or collar agreement, or other agreement or arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; but "Contingent Obligation" does not include endorsements in the ordinary course of business. The amount of a Contingent Obligation is the stated or determined amount of the primary obligation for which the Contingent Obligation is made or, if not determinable, the maximum reasonably anticipated liability for it determined by the Person in good faith; but the amount may not exceed the maximum of the obligations under any guarantee or other support arrangement.

"Control Agreement" is any control agreement entered into among the depository institution at which Borrower maintains a Deposit Account or the securities intermediary or commodity intermediary at which Borrower maintains a Securities Account or a Commodity Account, Borrower, and Bank pursuant to which Bank obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.

"Copyrights" are any and all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work thereof, whether published or unpublished and whether or not the same also constitutes a trade secret.

"Credit Extension" is any Advance, any Overadvance, Letter of Credit, FX Contract or any other extension of credit by Bank for Borrower's benefit.

"Current Liabilities" are (a) all obligations and liabilities of Borrower to Bank, plus, (b) without duplication of (a), the aggregate amount of Borrower's Total Liabilities that mature within one (1) year.

"Default" means any event which with notice or passage of time or both, would constitute an Event of Default.

"Default Rate" is defined in Section 2.6(b).

"Deferred Revenue" is all amounts received or invoiced in advance of performance under contracts and not yet recognized as revenue.

"Deposit Account" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

"**Division**" means, in reference to any Person which is an entity, the division of such Person into two (2) or more separate Persons, with the dividing Person either continuing or terminating its existence as part of such division, including, without limitation, as contemplated under Section 18-217 of the Delaware Limited Liability Company Act for limited liability companies formed under Delaware law, Section 17-220 of the Delaware Revised Uniform Limited Partnership Act for limited partnerships formed under Delaware law, or any analogous action taken pursuant to any other Applicable Law with respect to any corporation, limited liability company, partnership or other entity.

"Dollars," "dollars" or use of the sign "\$" means only lawful money of the United States and not any other currency, regardless of whether that currency uses the "\$" sign to denote its currency or may be readily converted into lawful money of the United States.

"Dollar Equivalent" is, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in a Foreign Currency, the equivalent amount therefor in Dollars as determined by Bank at such time on the basis of the then-prevailing rate of exchange in New York, New York, for sales of the Foreign Currency for transfer to the country issuing such Foreign Currency.

"Effective Date" is defined in the preamble hereof.

"Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.

"ERISA" is the Employee Retirement Income Security Act of 1974, as amended, and its regulations.

"Event of Default" is defined in Section 8.

"Exchange Act" is the Securities Exchange Act of 1934, as amended.

"Excluded Account" means (a) Deposit Accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of Borrower's employees, (b) deposit securities, commodity or similar accounts with financial institutions other than Bank inside of the United States, so long as no more than Seven Million Five Hundred Thousand Dollars (\$7,500,000) in the aggregate is maintained in such accounts for more than seven (7) consecutive days, (c) deposit, securities, commodity or similar accounts with financial institutions other than Bank outside of the United States so long as no more than Seven Million Five Hundred Thousand Dollars (\$7,500,000) in the aggregate is maintained in such accounts for more than seven (7) consecutive days, and (d) PayPal, Stripe or similar payment processing accounts.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to Bank or required to be withheld or deducted from a payment to Bank, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of Bank being organized under the laws of, or having its principal office or its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) U.S. federal withholding Taxes imposed on amounts payable to or for the

account of Bank with respect to an applicable interest in an Advance pursuant to a law in effect on the date on which (i) Bank acquires such interest in the Advance or (ii) Bank changes its lending office, (c) Taxes attributable to Bank's failure to comply with Section 2.11(f) and (d) any withholding Taxes imposed under FATCA.

"FATCA" means Sections 1471 through 1474 of the U.S. Code as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the U.S. Code.

"Floor" means a rate of interest equal to 0%.

"Foreign Currency" means lawful money of a country other than the United States.

"Foreign Subsidiary" means any Subsidiary which (i) is not organized under the laws of the United States or any state or territory thereof or the District of Columbia or (ii) an entity organized under the laws of the United States or any state or territory thereof or the District of Columbia substantially all of the assets of which consist of equity interests in, or other securities of or debt obligations owed by, one or more CFCs (or are treated as consisting of such assets for U.S. federal income tax purposes).

"Funding Date" is any date on which a Credit Extension is made to or for the account of Borrower which shall be a Business Day.

"FX Contract" is any foreign exchange contract by and between Borrower and Bank under which Borrower commits to purchase from or sell to Bank a specific amount of Foreign Currency on a specified date.

"GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" is all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all Intellectual Property, claims, income and other tax refunds, security and other deposits, payment intangibles, contract rights, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including without limitation key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.

"Governmental Approval" is any consent, authorization, approval, order, license, franchise, permit, certificate, accreditation, registration, filing or notice, of, issued by, from or to, or other act by or in respect of, any Governmental Authority.

"Governmental Authority" is any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity

exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.

"Guarantor" is any Person providing a Guaranty in favor of Bank.

"Guaranty" is any guarantee of all or any part of the Obligations, as the same may from time to time be amended, restated, modified or otherwise supplemented.

"Indebtedness" is (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations and (d) Contingent Obligations with respect to Indebtedness described in clauses (a) through (c) and (e) of this definition and (e) and other short- and long-term obligations under debt agreements and lines of credit.

"Indemnified Person" is defined in Section 12.3.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of Borrower under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.

"Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, receivership, arrangement, or other relief.

"Intellectual Property" means, with respect to any Person, all of such Person's right, title, and interest in and to the following:

- (a) its Copyrights, Trademarks and Patents;
- (b) any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions, know-how and operating manuals;
  - (c) any and all source or object code;
  - (d) any and all design rights which may be available to such Person;
- (e) any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
  - (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.

"Interest Election Request" means a request by Borrower to convert or continue an Advance in accordance with Section 2.5, which shall be in such form as Bank may approve.

"Interest Payment Date" means (a) as to any Prime Rate Advance, the last Business Day of each calendar month and the Revolving Line Maturity Date and (b) as to any SOFR Advance, the last day of each Interest Period therefor and, in the case of any Interest Period of more than three months' duration,

each day prior to the last day of such Interest Period that occurs at three month intervals after the first day of such Interest Period, and the Revolving Line Maturity Date.

"Interest Period" means, as to any SOFR Advance, the period commencing on the date such credit extension is borrowed and ending on the numerically corresponding day in the calendar month that is one, three, six or twelve months thereafter (in each case, subject to the availability thereof), as specified in the applicable Borrowing Request or Interest Election Request; provided that (i) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (ii) any Interest Period that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period) shall end on the last Business Day of the last calendar month of such Interest Period, (iii) no Interest Period shall extend beyond the Revolving Line Maturity Date and (iv) no tenor that has been removed from this definition pursuant to Section 2.13(d) shall be available for specification in such Borrowing Request or Interest Election Request. For purposes hereof, the date of an Advance initially shall be the date on which such Advance is made and thereafter shall be the effective date of the most recent conversion or continuation of such Advance.

"Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of Borrower's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.

"Investment" is any beneficial ownership interest in any Person (including stock, partnership, membership, or other ownership interest or other securities), and any loan, advance or capital contribution to any Person.

"JPMorgan" is defined in Section 7.1.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"Letter of Credit" is a standby or commercial letter of credit issued by Bank upon request of Borrower based upon an application, guarantee, indemnity, or similar agreement in accordance with Section 2.1(b).

"Letter of Credit Fee" is defined in Section 2.5(c).

"Letter of Credit Sublimit" means a sublimit for Letters of Credit under the Revolving Line not to exceed Five Million Dollars (\$5,000,000).

"Lien" is a claim, mortgage, deed of trust, levy, attachment, charge, hypothecation, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.

"Liquidity" is on any date, Borrower's unrestricted and unencumbered cash and Cash Equivalents maintained with Bank, Bank's Affiliates or any other bank or financial institution, provided that such bank or financial institution executed and delivered a Control Agreement or other appropriate instrument in favor of Bank in a form acceptable to Bank.

"Loan Documents" are, collectively, this Agreement and any schedules, exhibits, certificates, notices, the Perfection Certificate, any Control Agreement, any Bank Services Agreement, any subordination agreement, any note or notes, any guaranties executed by Borrower or any Guarantor, any other documents related to the foregoing, and any other present or future agreement by Borrower and/or any Guarantor with or for the benefit of Bank, all as amended, restated, or otherwise modified in accordance with the terms thereof.

"Material Adverse Change" is (a) a material impairment in the perfection or priority of Bank's Lien in the Collateral or in the value of such Collateral; (b) a material adverse change in the business, operations, or condition (financial or otherwise) of Borrower; or (c) a material impairment of the prospect of repayment of any portion of the Obligations.

"Obligations" are Borrower's obligations to pay when due any debts, principal, interest, fees, Revolving Line Commitment Fee, Unused Revolving Line Facility Fee, Bank Expenses and other amounts Borrower owes Bank now or later, whether under this Agreement, the other Loan Documents or otherwise, including, without limitation, all obligations relating to Bank Services and interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of Borrower assigned to Bank, and to perform Borrower's duties under the Loan Documents.

"OFAC" is the Office of Foreign Assets Control of the United States Department of the Treasury and any successor thereto.

"Operating Documents" are, for any Person, such Person's formation documents, as certified by the Secretary of State (or equivalent agency) of such Person's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Effective Date, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.

"Other Connection Taxes" means Taxes imposed as a result of a present or former connection between Bank and the jurisdiction imposing such Tax (other than connections arising from Bank having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment.

"Overadvance" is defined in Section 2.3.

- "Patents" means all patents, patent applications and like protections including without limitation improvements, divisions, continuations, renewals, reissues, extensions and continuations-in-part of the same.
  - "Payment Date" is the last calendar day of each month.
  - "Perfection Certificate" is defined in Section 5.1.
  - "Periodic Term SOFR Determination Day" is defined in the definition of "Term SOFR".
- "Permitted Acquisition" means a transaction whereby Borrower acquires all or substantially all of the capital stock or property of another Person, which satisfies each of the following conditions:
- (a) such transaction shall only involve an entity formed, and assets located, in the United States, and the party or parties being acquired is in the same or a substantially similar line of business as Borrower;
- (b) no Event of Default has occurred and is continuing or would exist after giving effect to the transaction and Bank has received satisfactory evidence that Borrower is in compliance with all terms and conditions of this Agreement (and that it will be in compliance after giving effect to the transaction);
  - (c) the acquisition is approved by the board of directors (or equivalent control group) of all parties to the transaction;
- (d) the total aggregate cash consideration to be paid by Borrower and its Subsidiaries in connection with any individual transaction or in the aggregate for all such transactions does not exceed Fifty Million Dollars (\$50,000,000) at any time; provided that immediately after giving effect to the transaction, Borrower shall (i) have Liquidity of at least Thirty Million Dollars (\$30,000,000), and (ii) be in compliance with the Financial Covenant immediately prior to such transaction and on a pro forma basis; provided further that if immediately after giving effect to the transaction, Borrower maintains Liquidity of at least Seventy-Five Million Dollars (\$75,000,000), the total aggregate cash consideration to be paid by Borrower and its Subsidiaries in connection with such transaction may exceed Fifty Million Dollars (\$50,000,000) so long as Borrower shall be in compliance with the Financial Covenant immediately prior to such transaction and on a pro forma basis;
- (e) Borrower provides Bank (i) written notice of the transaction at least three (3) Business Days prior to the closing of the transaction, and (ii) copies of the acquisition agreement and other material documents relative to the contemplated transaction and such other financial information, financial analysis, documentation or other information relating to such transaction as Bank shall reasonably request within five (5) Business Days after the closing of the transaction;
  - (f) Borrower is a surviving legal entity after completion of the contemplated transaction;
  - (g) the contemplated transaction is consensual and non-hostile;
- (h) no Indebtedness will be incurred, assumed, or would exist with respect to Borrower or its Subsidiaries as a result of the contemplated transaction, other than Permitted Indebtedness, and no Liens will be incurred, assumed, or would exist with respect to the assets of Borrower or its Subsidiaries as a result of the contemplated transaction, other than Permitted Liens;

- (i) any Subsidiary of Borrower acquired in the contemplated transaction shall provide to Bank a joinder to the Loan Agreement to cause such Subsidiary to become a co-borrower or guarantor hereunder, together with such appropriate financing statements and/or Control Agreements, all in form and substance satisfactory to Bank (including being sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the assets of such Subsidiary in accordance with Section 6.13;
  - (j) the acquisition and the company being acquired is accretive in all respects; and
- (k) Borrower shall have delivered to Bank, at least five (5) Business Days prior to the date on which any such acquisition is to be consummated (or such later date as is agreed by Bank in its sole discretion), a certificate of a Responsible Officer of Borrower, in form and substance reasonably satisfactory to Bank, certifying that all of the requirements set forth in this definition have been satisfied or will be satisfied on or prior to the consummation of such purchase or other acquisition.

#### "Permitted Indebtedness" is:

- (a) Borrower's Indebtedness to Bank under this Agreement and the other Loan Documents and any other Indebtedness in favor of Bank;
  - (b) Indebtedness existing on the Effective Date which is shown on the Perfection Certificate;
  - (c) Subordinated Debt;
  - (d) unsecured Indebtedness to trade creditors incurred in the ordinary course of business;
  - (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
  - (f) Indebtedness secured by Liens permitted under clauses (a) and (c) of the definition of "Permitted Liens" hereunder;
- (g) Indebtedness of Borrower to any Subsidiary and Contingent Obligations of any Subsidiary with respect to obligations of Borrower (provided that the primary obligations are not prohibited hereby), and Indebtedness of any Subsidiary to Borrower in an aggregate principal amount not to exceed Two Million Five Hundred Thousand Dollars (\$2,500,000) or any other Subsidiary and Contingent Obligations of any Subsidiary with respect to obligations of any other Subsidiary (provided that the primary obligations are not prohibited hereby):
- (h) Indebtedness in respect of performance bonds, bid bonds, appeal bonds, surety bonds and similar obligations, in each case provided in the ordinary course of business;
  - (i) Indebtedness that otherwise constitutes Permitted Investments:
  - (j) Permitted Publicly Traded Indebtedness;
- (k) Indebtedness in connection with corporate credit cards in an aggregate principal amount not to exceed Five Million Dollars (\$5,000,000);
  - (1) Indebtedness incurred in connection with the financing of insurance premiums in the ordinary course of business;

- (m) contingent liabilities in respect of any indemnification obligation, adjustment of purchase price (including working capital adjustments), non-compete, or similar obligations of the Borrower or its Subsidiaries incurred in connection with the consummation of one or more Permitted Investments;
- (n) other Indebtedness in an aggregate principal amount outstanding not to exceed Two Million Five Hundred Thousand Dollars (\$2,500,000); and
- (o) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (n) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be.

#### "Permitted Investments" are:

- (a) Investments (including, without limitation, Subsidiaries) existing on the Effective Date which are shown on the Perfection Certificate;
- (b) (i) Investments consisting of Cash Equivalents, and (ii) any Investments permitted by Borrower's investment policy, as amended from time to time, provided that such investment policy (and any such amendment thereto) has been approved in writing by Bank;
- (c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of Borrower's business;
- (d) Investments consisting of deposit accounts (but only to the extent that Borrower is permitted to maintain such accounts pursuant to Section 6.8 of this Agreement) in which Bank has a first priority perfected security interest (subject to Permitted Liens);
  - (e) Investments accepted in connection with Transfers permitted by Section 7.1;
- (f) Investments consisting of the ownership of equity interests in Subsidiaries and/or creation of a Subsidiary for the purpose of consummating a merger transaction permitted by Section 7.3 of this Agreement, which is otherwise a Permitted Investment;
- (g) Investments (i) by Borrower in Subsidiaries not to exceed Two Million Five Hundred Thousand Dollars (\$2,500,000) in the aggregate in any fiscal year, (ii) by Subsidiaries (which is not a Borrower or guarantor) in other Subsidiaries or in Borrower, and (iii) Investments by a Borrower or Guarantor in any other Borrower or Guarantor;
- (h) Investments consisting of (i) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business, and (ii) loans to employees, officers or directors relating to the purchase of equity securities of Borrower or its Subsidiaries pursuant to employee stock purchase plans or agreements approved by the Board;
- (i) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business;

- (j) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this paragraph (j) shall not apply to Investments of Borrower in any Subsidiary; and
- (k) Investments not otherwise permitted in an aggregate amount of not more than Two Million Five Hundred Thousand Dollars (\$2,500,000) in each fiscal year.

#### "Permitted Liens" are:

- (a) Liens existing on the Effective Date which are shown on the Perfection Certificate or arising under this Agreement or the other Loan Documents;
- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due and payable or (ii) being contested in good faith and for which Borrower maintains adequate reserves on Borrower's Books, provided that no notice of any such Lien has been filed or recorded under the U.S. Code and the Treasury Regulations adopted thereunder;
- (c) purchase money Liens (i) on Equipment acquired or held by Borrower incurred for financing the acquisition of the Equipment securing no more than Two Million Five Hundred Thousand Dollars (\$2,500,000)in the aggregate amount outstanding, or (ii) existing on Equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the Equipment;
- (d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as such Liens attach only to Inventory, securing liabilities in the aggregate amount not to exceed Two Million Five Hundred Thousand Dollars (\$2,500,000) and which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the Indebtedness secured by Liens described in (a) through (c), but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase;
- (g) leases or subleases of real property granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non- exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Bank a security interest therein;
- (h) non-exclusive licenses of Intellectual Property granted to third parties in the ordinary course of business, and licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discreet geographical areas outside of the United States;

- (i) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default under Sections 8.4 and 8.7;
- (j) deposits to secure the performance of bids, tenders, trade contracts, leases, government contracts, statutory obligations, surety, stay, customs and appeal bonds, performance and other similar obligations, in each case provided in the ordinary course of business;
  - (k) Liens securing Subordinated Debt;
  - (1) Liens on insurance policies and the proceeds thereof securing the financing of premiums with respect thereto; and
- (m) Liens in favor of other financial institutions arising in connection with Borrower's deposit and/or securities accounts held at such institutions, provided that (i) Bank has a first priority perfected security interest (subject to Permitted Liens) in the amounts held in such deposit and/or securities accounts (ii) such accounts are permitted to be maintained pursuant to Section 6.8 of this Agreement.
- "Permitted Publicly Traded Indebtedness" means any publicly traded Indebtedness incurred by the Borrower, including any high-yield bonds, convertible bonds or other Indebtedness, which satisfies each of the following conditions:
- (a) the aggregate principal amount of such Permitted Publicly Trade Indebtedness shall not to exceed One Hundred Million Dollars (\$100,000,000) at any time outstanding;
- (b) no Event of Default has occurred and is continuing or would exist after giving effect to the transaction and Bank has received satisfactory evidence that Borrower is in compliance with all terms and conditions of this Agreement (and that it will be in compliance after giving effect to the transaction);
  - (c) such Indebtedness shall be unsecured;
- (d) immediately after giving effect to the transaction, Borrower shall have Liquidity of at least Seventy-Five Million Dollars (\$75,000,000);
  - (e) immediately after giving effect to the transaction, Borrower shall be in compliance with the Financial Covenant; and
  - (f) Borrower has received the prior written consent of Bank.
- "Person" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.
- "Prepayment Notice" means a notice delivered by Borrower to Bank of any prepayment of an Advance in accordance with Section 2.8(b), which shall be in such form as Bank may approve.
- "Prime Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement; and provided further that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as

determined by Bank, the "Prime Rate" shall mean the rate of interest per annum announced by Bank as its prime rate in effect at its principal office in the State of New York (such Bank announced Prime Rate not being intended to be the lowest rate of interest charged by Bank in connection with extensions of credit to debtors); provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement. Changes to the interest rate of any Credit Extension based on changes to the Prime Rate shall be effective on the effective date of any change to the Prime Rate and to the extent of any such change.

"Prime Rate Advance" means an Advance that bears interest at a rate based on the Prime Rate.

"Public Company Costs": as to any Person, costs associated with, or in anticipation of, or preparation for, compliance with the requirements of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated in connection therewith and costs relating to compliance with the provisions of the Securities Act of 1933 (as amended, and the rules and regulations of the SEC promulgated thereunder, as amended) and the Securities Exchange Act of 1934 (as amended, and the rules and regulations of the SEC promulgated thereunder, as amended) or any other comparable body of laws, rules or regulations, as companies with listed equity, directors' compensation, fees and expense reimbursement, costs relating to enhanced accounting functions and investor relations, stockholder meetings and reports to stockholders, directors' and officers' insurance and other executive costs, legal and other professional fees, listing fees and other transaction costs, in each case to the extent arising solely by virtue of the listing of such Person's equity securities on a national securities exchange or issuance of public debt securities.

"Quarterly Financial Statements" is defined in Section 6.2(a).

"Quick Assets" is on any date, (a) Borrower's unrestricted and unencumbered cash and Cash Equivalents maintained with Bank, Bank's Affiliates or any other bank or financial institution, provided that such bank or financial institution executed and delivered a Control Agreement or other appropriate instrument in favor of Bank in a form acceptable to Bank, plus (b) net billed accounts receivable, determined according to GAAP.

"Registered Organization" is any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.

"Relevant Governmental Body" means the Federal Reserve Board or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Federal Reserve Board or the Federal Reserve Bank of New York, or any successor thereto.

"Requirement of Law" is as to any Person, the organizational or governing documents of such Person, and any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Responsible Officer" is any of the Chief Executive Officer, President, Chief Financial Officer and Controller of Borrower.

"Restricted License" is any material license agreement (excluding any "shrink wrap" or other licenses that are generally commercially available) with respect to which Borrower is the licensee (a) that validly prohibits or otherwise restricts Borrower from granting a security interest in Borrower's interest in such license agreement or any other property subject to such license agreement in favor of Bank, or

- (b) for which a default under or termination of could reasonably be expected to interfere in any material respect with Bank's right to sell any Collateral.
- "Revolver Commitments" means the obligations of Bank (or any other lender pursuant to Section 2.15) to make Advances under the Revolving Line, as hereafter may be increased pursuant to Section 2.15; provided that such Advances shall not exceed the Availability Amount.
- "Revolving Line" is an aggregate principal amount equal to Twenty-Five Million Dollars (\$25,000,000), as hereafter may be increased pursuant to Section 2.15.
  - "Revolving Line Maturity Date" is February 7, 2027.
- "Sanctioned Person" means a Person that: (a) is listed on any Sanctions list maintained by OFAC or any similar Sanctions list maintained by any other Governmental Authority having jurisdiction over Borrower; (b) is located, organized, or resident in any country, territory, or region that is the subject or target of Sanctions; or (c) is owned or controlled by one (1) or more Persons described in clauses (a) and (b) hereof.
- "Sanctions" means the economic sanctions laws, regulations, embargoes or restrictive measures administered, enacted or enforced by the United States government and any of its agencies, including, without limitation, OFAC and the U.S. State Department, or any other Governmental Authority having jurisdiction over Borrower.
  - "SEC" shall mean the Securities and Exchange Commission, any successor thereto, and any analogous Governmental Authority.
  - "Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.
  - "SOFR" means a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.
- "SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).
  - "SOFR Advance" means an Advance that bears interest at a rate based on Term SOFR.
  - "SOFR Borrowing" means, as to any Advance, the SOFR Advances comprising such Advance.
- "Specified Affiliate" is any Person (a) more than ten percent (10.0%) of whose aggregate issued and outstanding equity or ownership securities or interests, voting, non-voting or both, are owned or held directly or indirectly, beneficially or of record, by Borrower, and/or (b) whose equity or ownership securities or interests representing more than ten percent (10.0%) of such Person's total outstanding combined voting power are owned or held directly or indirectly, beneficially or of record, by Borrower.
- "Subordinated Debt" is indebtedness incurred by Borrower subordinated to all of Borrower's now or hereafter indebtedness to Bank (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Bank entered into between Bank and the other creditor), on terms acceptable to Bank.

"Subsidiary" is, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock, partnership, membership, or other ownership interest or other equity securities having ordinary voting power (other than stock, partnership, membership, or other ownership interest or other equity securities having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless the context otherwise requires, each reference to a Subsidiary herein shall be a reference to a Subsidiary of Borrower or Guarantor.

"Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Term SOFR" means, for any calculation with respect to a SOFR Advance, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day; provided further that if Term SOFR determined as provided above (including pursuant to the proviso above) shall ever be less than the Floor, then Term SOFR shall be deemed to be the Floor.

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by Bank in its reasonable discretion).

"Term SOFR Reference Rate" means the forward-looking term rate based on SOFR.

"Total Liabilities" is on any day, obligations that should, under GAAP, be classified as liabilities on Borrower's consolidated balance sheet, including all Indebtedness.

"Trademarks" means any trademark and servicemark rights, whether registered or not, applications to register and registrations of the same and like protections, and the entire goodwill of the business of Borrower connected with and symbolized by such trademarks.

"Transfer" is defined in Section 7.1.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"U.S. Code" means the Internal Revenue Code of 1986, as amended.

- "U.S. Person" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the U.S. Code.
- "U.S. Subsidiary" means any Subsidiary that is incorporated or formed under the laws of the United States of America, any state thereof or the District of Columbia (excluding any Subsidiary organized under the laws of any political subdivision of the United States (including any disregarded entity for U.S. federal income tax purposes), substantially all of the assets of which consist of, directly or indirectly, equity securities of one or more CFCs or indebtedness of such CFCs).
- "Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Unused Revolving Line Facility Fee" is defined in Section 2.5(b).

"UPS" is defined in Section 7.1.

"USA Patriot Act" means the "Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001" (Public Law 107-56, signed into law on October 26, 2001), as amended from time to time.

[Balance of Page Intentionally Left Blank]

**IN WITNESS WHEREOF**, the parties hereto have caused this Agreement to be executed as of the Effective Date.

# **BORROWER:**

# COUCHBASE, INC.

By: /s/ GREG HENRY

Name: Greg Henry

Title: Chief Financial Officer

[Signature Page to Loan and Security Agreement]

# **BANK:**

# MUFG BANK, LTD.

By: /s/ LAWRENCE CHAO

Name: Lawrence Chao
Title: Vice President

[Signature Page to Loan and Security Agreement]

#### **EXHIBIT A**

#### COLLATERAL DESCRIPTION

The Collateral consists of all of Borrower's right, title and interest in and to the following personal property:

All goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles (except as provided below), commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, certificates of deposit, fixtures, letters of credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and

all Borrower's Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing.

Notwithstanding the foregoing, the Collateral does not include (i) any accounts receivable and other claims which arise out of the sale of goods or services to United Parcel Service, Inc., a Delaware corporation, and/or its subsidiaries or affiliates, to JPMorgan and/or one (1) or more other investors, pursuant to the terms of a Master Receivables Purchase Acceptance Letter by and between Borrower and JPMorgan or to any other financial institution pursuant to any similar arrangement, (ii) with respect to stock in Foreign Subsidiaries, more than sixty-five percent (65%) of the presently existing and hereafter arising issued and outstanding shares of capital stock owned by Borrower of any Foreign Subsidiary which shares entitle the holder thereof to vote for directors or any other matter (as determined under U.S. federal tax principles), (iii) any property to the extent that such grant of security interest is prohibited by any Requirement of Law of a Governmental Authority or constitutes a breach or default under or results in the termination of or requires any consent not obtained under, any contract, license, agreement, instrument or other document evidencing or giving rise to such property, except to the extent that such Requirement of Law or the term in such contract, license, agreement, instrument or other document providing for such prohibition, breach, default or termination or requiring such consent is ineffective under Section 9-406, 9-407, 9-408 or 9-409 of the Code (or any successor provision or provisions) of any relevant jurisdiction or any other applicable law (including the Bankruptcy Code) or principles of equity; provided, however, that such security interest shall attach immediately at such time as such Requirement of Law is not effective or applicable, or such prohibition, breach, default or termination is no longer applicable or is waived, and to the extent severable, shall attach immediately to any portion of the Collateral that does not result in such consequences, (iv) any interest of Borrower as a lessee or sublessee under a real property lease or an Equipment lease if Borrower is prohibited by the terms of such lease from granting a security interest in such lease or under which such an assignment or Lien would cause a default to occur under such lease (but only to the extent that such prohibition is enforceable under all applicable laws including, without limitation, the Code); provided, however, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by Borrower or Bank, (v) the Excluded Accounts, or (vi) any Intellectual Property; provided, however, the Collateral shall include all Accounts and all proceeds of Intellectual Property. If a judicial authority (including a U.S. Bankruptcy Court) would hold that a security interest in the underlying Intellectual Property is necessary to have a security interest in such Accounts and such property that are proceeds of Intellectual Property, then the Collateral shall automatically, and effective as of the Effective Date, include the Intellectual

Property to the extent necessary to permit perfection of Bank's security interest in such Accounts and such other property of Borrower that are proceeds of the Intellectual Property.

Pursuant to the terms of a certain negative pledge arrangement with Bank, Borrower has agreed not to encumber any of its Intellectual Property without Bank's prior written consent.

Exhibit A - 62

#### **EXHIBIT B**

#### **COMPLIANCE STATEMENT**

TO: MUFG BANK, LTD.	Date:
FROM: COUCHBASE IN	C. (the "Borrower")

Under the terms and conditions of the Loan and Security Agreement between Borrower and Bank (as amended, modified, supplemented and/or restated from time to time, "Agreement"), Borrower is in complete compliance for the period ending \_\_\_\_\_ with all required covenants except as noted below. Attached are the required documents evidencing such compliance, setting forth calculations prepared in accordance with GAAP consistently applied from one period to the next except as explained in an accompanying letter or footnotes (other than, with respect to unaudited financial statements for the absence of footnotes and year-end audit adjustments). Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Agreement.

## Please indicate compliance status by circling Yes/No under "Complies" column.

Reporting Covenants	Required	Complies
Quarterly Financial Statements with Compliance Statement (first three fiscal quarters of each year)	Within the earlier of (i) 45 days of fiscal quarter end, or (ii) 5 days after filing with SEC (satisfied if periodic reports containing such information are available on the SEC website)	Yes No
	Within the earlier of (i) FYE within 120 days or (ii) 5 days after filing with SEC (satisfied if periodic reports containing such information are available on the SEC website)	Yes No
Board approved projections	Within the earlier of (i) 15 days after approval by the Board or (ii) 60 after FYE, and as amended/updated	Yes No

Financial Covenant	Required	Actual	<u>Complies</u>
Minimum LTM Consolidated EBITDA			Yes No

The following financial covenant analyses and information set forth in Schedule 1 attached hereto are true and accurate as of the fiscal period described therein.

The following are the exceptions with respect to the statements above: (If no exceptions exist, state "No exceptions to note.")

Exhibit B - 64

# **Schedule 1 to Compliance Statement**

# **Financial Covenant of Borrower**

Calculations showing compliance with the financial covenant.

# EXHIBIT C

# BORROWING REQUEST FORM

Date:		
Advance:		
Complete <i>Outgoing Wire Request</i> section below if all or	a portion of the funds from this loan advance are for an outgoing wire.	
From Account #(Loan Account #) (D	To Account #eposit Account #)	
Amount of Advance \$ Interest Period	(If no Interest Period is specified with respect to any requested SOFR Advance, Borrower shall nth's duration.)	
be deemed to have selected an Interest Period of one mo	nth's duration.)	
All Borrower's representations and warranties in the Loan and Security Agreement and the other Loan Documents are true, correct and complete in all material respects on the date of the request for an advance; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true and correct in all material respects as of such date:		
Authorized Signature:Phone Number: Print Name/Title:		
Outgoing Wire Request: Complete only if all or a portion of funds from the loa	an advance above is to be wired.	
Beneficiary Name:	Amount of Wire: \$	
Beneficiary Bank:	Account Number:	
City and State:		
Beneficiary Bank Transit (ABA) #:Beneficiary Bank Code (Swift, Sort, Chip, etc.):  (For International Wire Only)		
Intermediary Bank:Transit (ABA) #: For Further Credit to:		
Special Instruction:		
By signing below, I (we) acknowledge and agree that my (our) funds transfer request shall be processed in accordance with and subject to the terms and conditions set forth in the agreements(s) covering funds transfer service(s), which agreements(s) were previously received and executed by me (us).		
Authorized Signature:	2 <sup>nd</sup> Signature (if required):	
Print Name/Title:	Print Name/Title:	
Telephone #:Telephone #:		

# CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, Matthew M. Cain, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Couchbase, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 6, 2024	Ву:	/s/ MATTHEW M. CAIN
	Name:	Matthew M. Cain
	Title:	Chair, President and Chief Executive Officer
		(Principal Executive Officer)

# CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, Greg Henry, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Couchbase, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 6, 2024	Ву:	/s/ GREG HENRY
	Name:	Greg Henry
	Title:	Senior Vice President and Chief Financial Officer
		(Principal Financial Officer)

# CERTIFICATIONS OF PRINCIPAL EXECUTIVE OFFICER AND PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Matthew M. Cain, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report on Form 10-Q of Couchbase, Inc. for the period ended April 30, 2024 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and that the information contained in such Quarterly Report on Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Couchbase, Inc.

Date: June 6, 2024	By:	/s/ MATTHEW M. CAIN
	Name:	Matthew M. Cain
	Title:	Chair, President and Chief Executive Officer
		(Principal Executive Officer)
Quarterly Report on Form 10-Q of Couchbas	se, Inc. for the period ended April 30, 2024 funded, and that the information contained in s	o Section 906 of the Sarbanes-Oxley Act of 2002, that the ally complies with the requirements of Section 13(a) or 15(d) of uch Quarterly Report on Form 10-Q fairly presents, in all material
Date: June 6, 2024	By:	/s/ GREG HENRY
	Name:	Greg Henry
	Title:	Senior Vice President and Chief Financial Officer
		(Principal Financial Officer)